## **UNITED STATES SECURITIES AND EXCHANGE COMMISSION**

Washington, D.C. 20549

	FORM 10	)-Q
(Marl ⊠	k One) QUARTERLY REPORT PURSUANT TO SECTION 13 OR 1934	15(D) OF THE SECURITIES EXCHANGE ACT OF
	For the quarterly period ended September 30, 2005	
	OR	
	TRANSITION REPORT PURSUANT TO SECTION 13 OR 1934	15(D) OF THE SECURITIES EXCHANGE ACT OF
	For the transition period from to	
	Commission file numb	er 1-13300
	(Exact name of registrant as speci  Delaware  (State or other jurisdiction of incorporation or organization)	fied in its charter)  54-1719854  (I.R.S. Employer Identification No.)
	1680 Capital One Drive, McLean, Virginia (Address of principal executive offices)	22102 (Zip Code)
	(703) 720-100 (Registrant's telephone number, in	
	(Not Applicable) (Former name, former address and former fiscale)	
the pre	te by check mark whether the registrant (1) has filed all reports required to be file eceding 12 months (or for such shorter period that the registrant was required to fi st 90 days. Yes 🗵 No 🗆	
Indica	te by check mark whether the registrant is an accelerated filer (as defined in Rule	12b-2 of the Exchange Act). Yes $\boxtimes$ No $\square$

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act.): Yes  $\square$  No  $\boxtimes$ As of September 30, 2005 there were 266,611,478 shares of the registrant's Common Stock, par value \$.01 per share, outstanding.

## CAPITAL ONE FINANCIAL CORPORATION FORM 10-Q

## INDEX

September 30, 2005

	Page
PART I. FINANCIAL INFORMATION	<u>—</u>
Item 1. Financial Statements (unaudited):	
Condensed Consolidated Balance Sheets	3
Condensed Consolidated Statements of Income	4
Condensed Consolidated Statements of Changes in Stockholders' Equity	5
Condensed Consolidated Statements of Cash Flows	6
Notes to Condensed Consolidated Financial Statements	7
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	16
Item 3. Quantitative and Qualitative Disclosure of Market Risk	54
Item 4. <u>Controls and Procedures</u>	54
PART II. OTHER INFORMATION	
Item 1. <u>Legal Proceedings</u>	55
Item 2. Changes in Securities, Uses of Proceeds and Issuer Purchases of Equity Securities	55
Item 6. Exhibits and Reports on Form 8-K	55
<u>Signatures</u>	57

## **Part I. Financial Information**

Item 1. Financial Statements (unaudited)

## CAPITAL ONE FINANCIAL CORPORATION

**Condensed Consolidated Balance Sheets** 

(Dollars in thousands, except share and per share data) (unaudited)

	September 30 2005	December 31 2004
Assets: Cash and due from banks	\$ 812,330	\$ 327,517
Federal funds sold and resale agreements	2,409,392	773,695
Interest-bearing deposits at other banks	1,380,880	309,999
Cash and cash equivalents	4,602,602	1,411,211
Securities available for sale	9,436,667	9,300,454
Loans	38,851,763	38,215,591
Less: Allowance for loan losses	(1,447,000)	(1,505,000)
Net loans	37,404,763	36,710,591
Accounts receivable from securitizations	6,126,282	4,081,271
Premises and equipment, net	768,198	817,704
Interest receivable	367,757	252,857
Goodwill	736,058	352,157
Other	982,190	821,010
Total assets	\$60,424,517	\$53,747,255
Liabilities:		
Interest-bearing deposits	26,772,538	25,636,802
Senior and subordinated notes	6,651,891	6,874,790
Other borrowings	11,613,179	9,637,019
Interest payable	350,842	237,227
Other	3,998,840	2,973,228
Total liabilities	49,387,290	45,359,066
Stockholders' Equity:		
Preferred stock, par value \$.01 per share; authorized		
50,000,000 shares, none issued or outstanding	_	_
Common stock, par value \$.01 per share; authorized		
1,000,000,000 shares; 268,200,535 and 248,354,259 shares issued as of September 30, 2005 and December 31, 2004,		
respectively	2,682	2,484
Paid-in capital, net	3,979,525	2,711,327
Retained earnings	7,104,796	5,596,372
Cumulative other comprehensive income	20,104	144,759
Less: Treasury stock, at cost; 1,589,057 and 1,520,962 shares as of September 30, 2005 and December 31, 2004, respectively	(69,880)	(66,753)
Total stockholders' equity	11,037,227	8,388,189
Total liabilities and stockholders' equity	\$60,424,517	\$53,747,255

# CAPITAL ONE FINANCIAL CORPORATION Condensed Consolidated Statements of Income (Dollars in thousands, except per share data) (unaudited)

	Septe	onths Ended mber 30	Septer	nths Ended nber 30
	2005	2004	2005	2004
Interest Income:				
Loans, including past-due fees	\$1,228,160	\$1,083,286	\$3,602,294	\$ 3,137,379
Securities available for sale	87,978	84,492	269,387	224,289
Other	88,477	60,635	221,102	183,422
Total interest income	1,404,615	1,228,413	4,092,783	3,545,090
Interest Expense:				
Deposits	285,611	257,349	829,074	741,83
Senior and subordinated notes	98,309	121,166	317,382	370,393
Other borrowings	110,476	74,523	303,084	214,444
Total interest expense	494,396	453,038	1,449,540	1,326,676
Net interest income	910,219	775,375	2,643,243	2,218,414
Provision for loan losses	374,167	267,795	925,398	753,719
Net interest income after provision for loan losses	536,052	507,580	1,717,845	1,464,695
Non-Interest Income:				
Servicing and securitizations	993,788	940,246	2,923,768	2,724,605
Service charges and other customer-related fees	381,381	385,648	1,179,857	1,108,610
Interchange	125,454	117,043	380,962	339,967
Other	93,993	96,447	208,004	205,400
Total non-interest income	1,594,616	1,539,384	4,692,591	4,378,582
Non-Interest Expense:				
Salaries and associate benefits	414,348	415,988	1,289,950	1,260,075
Marketing	343,708	317,653	932,501	826,638
Communications and data processing	144,321	112,191	426,056	337,488
Supplies and equipment	86,866	94,190	256,973	257,093
Occupancy	39,426	41,407	97,536	150,620
Other	336,969	330,555	1,026,071	933,778
Total non-interest expense	1,365,638	1,311,984	4,029,087	3,765,692
Income before income taxes	765,030	734,980	2,381,349	2,077,585
Income taxes	273,881	244,819	852,520	<b>729,23</b> 1
Net income	\$ 491,149	\$ 490,161	\$1,528,829	\$ 1,348,354
Basic earnings per share	\$ 1.88	\$ 2.07	\$ 6.05	\$ 5.75
Diluted earnings per share	\$ 1.81	\$ 1.97	\$ 5.82	\$ 5.45
Dividends paid per share	\$ 0.03	\$ 0.03	\$ 0.08	\$ 0.08

## CAPITAL ONE FINANCIAL CORPORATION

Condensed Consolidated Statements of Changes in Stockholders' Equity (Dollars in thousands, except share and per share data) (unaudited)

	Common S	tock	Paid-in	Retained		umulative Other nprehensive	Treasury	Total Stockholder's
	Shares	Amount	Capital	Earnings		ome (Loss)	Stock	Equity
Balance, December 31, 2003	236,352,914	\$2,364	\$1,937,302	\$4,078,508	\$	83,158	\$(49,521)	\$ 6,051,811
Comprehensive income:								
Net income				1,348,354				1,348,354
Other comprehensive income, net of income tax:								
Unrealized losses on securities, net of income tax benefit of \$13,099						(21,404)		(21,404)
Foreign currency translation adjustments						12,387		12,387
Unrealized gains on cash flow hedging								
instruments, net of income taxes of \$18,772						31,670		31,670
Other comprehensive income						22,653		22,653
Comprehensive income								1,371,007
Cash dividends - \$.08 per share				(18,979)				(18,979)
Purchases of treasury stock				(10,575)			(182)	(182)
Issuances of common and restricted stock, net of							(102)	(10=)
forfeitures	(61,308)	(1)	18,135					18,134
Exercise of stock options and related tax benefits	7,737,076	77	411,782					411,859
Amortization of compensation expense for restricted stock awards	7,757,676	,,	66,476					66,476
Common stock issuable under incentive plan			29,934					29,934
Balance, September 30, 2004	244,028,682	\$2,440	\$2,463,629	\$5,407,883	\$	105,811	\$(49,703)	\$ 7,930,060
	240.054.250	<b>#2.101</b>	<b>#2 544 225</b>	A	Φ.	111 ==0	A (00 == D)	<b>*</b> 0.200.400
Balance, December 31, 2004	248,354,259	\$2,484	\$2,711,327	\$5,596,372	\$	144,759	\$(66,753)	\$ 8,388,189
Comprehensive income:				1 500 000				1 530 030
Net income				1,528,829				1,528,829
Other comprehensive loss, net of income tax benefit: Unrealized losses on securities, net of income								
tax benefit of \$37,196						(65,963)		(65,963)
Foreign currency translation adjustments						(73,533)		(73,533)
Unrealized gains on cash flow hedging						(73,333)		(73,333)
instruments, net of income taxes of \$13,710						14,841		14,841
Other comprehensive loss						(124,655)		(124,655)
Comprehensive income								1,404,174
Cash dividends - \$.08 per share				(20,405)				(20,405)
Purchases of treasury stock							(3,127)	(3,127)
Issuances of common and restricted stock, net of								
forfeitures	11,226,410	112	762,856					762,968
Exercise of stock options, and related tax benefits	8,619,866	86	392,321					392,407
Amortization of compensation expense for restricted stock								
awards			68,583					68,583
Common stock issuable under incentive plan			44,438					44,438
Balance, September 30, 2005	268,200,535	\$2,682	\$3,979,525	\$7,104,796	\$	20,104	\$(69,880)	\$11,037,227

## CAPITAL ONE FINANCIAL CORPORATION Condensed Consolidated Statements of Cash Flows (Dollars in thousands) (unaudited)

		ths Ended aber 30
	2005	2004
Operating Activities:		
Net income	\$ 1,528,829	\$ 1,348,354
Adjustments to reconcile net income to cash provided by operating activities:		
Provision for loan losses	925,398	753,719
Depreciation and amortization	306,537	290,379
(Reparation) impairment of long-lived assets	(14,938)	50,457
Losses on securities available for sale	6,397	23,123
Gains on sales of auto loans	(10,894)	(35,923
Losses on repurchases of senior notes	12,444	1,206
Stock plan compensation expense	113,021	96,410
Changes in assets and liabilities, net of effects from companies acquired:		
Increase in interest receivable	(114,900)	(18,513
Increase in accounts receivable from securitizations	(1,686,389)	(209,588
(Increase) decrease in other assets	(84,406)	181,699
Increase (decrease) in interest payable	111,071	(5,788
Increase in other liabilities	743,332	255,786
Net cash provided by operating activities	1,835,502	2,731,321
Investing Activities:		
Purchases of securities available for sale	(1,973,972)	(5,904,742
Proceeds from maturities of securities available for sale	1,050,509	1,014,943
Proceeds from sales of securities available for sale	651,032	1,149,862
Proceeds from sale of automobile loans	257,230	862,330
Proceeds from securitization of consumer loans	7,535,437	8,487,788
Net increase in consumer loans	(9,250,240)	(12,959,961
Principal recoveries of loans previously charged off	339,082	344,154
Additions of premises and equipment, net	(69,055)	(133,568
Net payments for companies acquired	(470,694)	_
Net cash used for investing activities	(1,930,671)	(7,139,194
Financing Activities:		
Net increase In Interest-bearing Deposits	1,135,736	2,937,991
Net increase in other borrowings	1,418,783	693,825
Issuances of senior notes	1,262,035	998,190
Maturities of senior notes	(876,567)	(1,031,892
Repurchases of senior notes	(648,840)	(28,159
Purchases of treasury stock	(3,127)	(182
Dividends paid	(20,405)	(18,979
Net proceeds from issuances of common stock	762,968	18,134
Proceeds from exercise of stock options	255,977	301,530
Net cash provided by financing activities	3,286,560	3,870,458
Increase (decrease) in cash and cash equivalents	3,191,391	(537,415
Cash and cash equivalents at beginning of period	1,411,211	1,980,282
Cash and cash equivalents at end of period	\$ 4,602,602	\$ 1,442,867

## CAPITAL ONE FINANCIAL CORPORATION Notes to Condensed Consolidated Financial Statements

(in thousands, except per share data) (unaudited)

## **Note 1: Significant Accounting Policies**

#### **Business**

The condensed consolidated financial statements include the accounts of Capital One Financial Corporation (the "Corporation") and its subsidiaries. The Corporation is a diversified financial services company whose subsidiaries market a variety of financial products and services to consumers. The principal subsidiaries are Capital One Bank (the "Bank"), which offers credit card products and deposit products, Capital One, F.S.B. (the "Savings Bank"), which offers consumer and commercial lending and consumer deposit products and Capital One Auto Finance, Inc. ("COAF") which offers automobile and other motor vehicle financing products. Capital One Services, Inc. ("COSI"), another subsidiary of the Corporation, provides various operating, administrative and other services to the Corporation and its subsidiaries. The Corporation and its subsidiaries are collectively referred to as the "Company."

#### **Basis of Presentation**

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States ("GAAP") for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by GAAP for complete consolidated financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Actual results could differ from these estimates.

Operating results for the three and nine months ended September 30, 2005 are not necessarily indicative of the results for the year ending December 31, 2005.

The notes to the consolidated financial statements contained in the Annual Report on Form 10-K for the year ended December 31, 2004 should be read in conjunction with these condensed consolidated financial statements.

All significant intercompany balances and transactions have been eliminated.

#### Reclassifications

Certain prior years' amounts have been reclassified to conform to the 2005 presentation.

#### Stock-Based Compensation

Prior to 2003, the Company applied Accounting Principles Board Opinion No. 25, *Accounting for Stock Issued to Employees* ("APB 25") and related Interpretations in accounting for its stock-based compensation plans. No compensation cost has been recognized for the Company's fixed stock options for years prior to 2003, as the exercise price of all such options equals or exceeds the market value of the underlying common stock on the date of grant. Effective January 1, 2003, the Company adopted the expense recognition provisions of SFAS No. 123, prospectively to all awards granted, modified, or settled after January 1, 2003. Typically, awards under the Company's plans vest over a three year period. Therefore, cost related to stock-based compensation included in net income for 2005, 2004 and 2003 in the interim periods is less than that which would have been recognized if the fair value method had been applied to all awards since the original effective date of SFAS 123. The effect on net income and earnings per share if the fair value based method had been applied to all outstanding and unvested awards in each period is presented in the table below.

	For the Three Months Ended September 30		For the Nine Mo Ended Septemb					
Pro Forma Information		2005		2004	- 2	2005	2	2004
Net income, as reported	\$49	91,149	\$49	90,161	\$1,5	28,829	\$1,3	48,354
Stock-based employee compensation expense included in reported net income	:	23,409		20,542		70,097		55,803
Stock-based employee compensation expense determined under fair value based method <sup>(1)</sup>	(	27,841)	(!	54,089)	(	(83,763)	(1	54,904)
Pro forma net income	\$4	86,717	\$45	56,614	\$1,5	515,163	\$1,2	49,253
Earnings per share:								
Basic – as reported	\$	1.88	\$	2.07	\$	6.05	\$	5.75
Basic – pro forma	\$	1.87	\$	1.93	\$	6.00	\$	5.33
Diluted – as reported	\$	1.81	\$	1.97	\$	5.82	\$	5.45
Diluted – pro forma	\$	1.78	\$	1.82	\$	5.72	\$	5.02

<sup>(1)</sup> Includes amortization of compensation expense for current year stock option grants and prior year stock option grants over the stock options' vesting period.

The fair value of the options granted during the three and nine months ended September 30, 2005 and 2004 was estimated at the date of grant using a Black-Scholes option-pricing model with the weighted average assumptions described below.

		For the Three Months Ended September 30		
Assumptions	2005 <sup>(1)</sup>	2004 <sup>(1)</sup>	2005	2004 <sup>(1)</sup>
Dividend yield	.13%	.15%	.14%	.15%
Volatility factors of expected market price of stock	21%	42%	52%	53%
Risk-free interest rate	3.91%	2.54%	4.18%	2.32%
Expected option lives (in years)	1.9	2.1	4.8	2.6

Stock options granted during the three months ended September 30, 2005 and 2004 and during the nine month period ended September 30, 2004 consisted primarily of grants issued upon exercises of vested stock options through stock swaps in accordance with stock option agreements.

#### **Note 2: Segments**

The Company maintains three distinct operating segments: U.S. Card, Auto Finance, and Global Financial Services. The U.S. Card segment consists of domestic credit card lending activities. The Auto Finance segment consists of automobile and other motor vehicle financing activities. The Global Financial Services segment consists of international lending activities, small business lending, installment loans, home loans, healthcare financing and other diversified activities. The U.S. Card, Auto Finance and Global Financial Services segments are considered reportable segments based on quantitative thresholds applied to the managed loan portfolio for reportable segments provided by SFAS No. 131, *Disclosures about Segments of an Enterprise and Related Information*, and are disclosed separately. The Other category includes the Company's liquidity portfolio, emerging businesses not included in the reportable segments, investments in external companies, and various non-lending activities. The Other category also includes the net impact of transfer pricing, certain unallocated expenses and gains/losses related to the securitization of assets.

As management makes decisions on a managed portfolio basis within each segment, information about reportable segments is provided on a managed basis. An adjustment to reconcile the managed financial information to the reported financial information in the consolidated financial statements is provided. This adjustment reclassifies a portion of net interest income, non-interest income and provision for loan losses into non-interest income from servicing and securitizations.

The Company maintains its books and records on a legal entity basis for the preparation of financial statements in conformity with GAAP. The following tables present information prepared from the Company's internal management information system, which is maintained on a line of business level through allocations from the consolidated financial results.

#### For the Three Months Ended September 30, 2005 **Financial** Total Securitization Total U.S. Card Reported **Auto Finance** Other Managed Services Adjustments Net interest income \$ 1,207,832 \$ 423,629 \$ (368)\$ 1,931,195 \$ (1,020,976) 910,219 300,102 (27,301)Non-interest income 851,036 3,005 273,067 1,099,807 494,809 1,594,616 Provision for loan losses 483,759 185,219 217,032 14,324 900,334 374,167 (526, 167)833,925 129,719 356,254 45,740 Non-interest expenses 1,365,638 1,365,638 Income tax provision (benefit) 259,414 41,521 (22,913)273,881 273,881 (4,141)481,770 (7,690)81,889 \$(64,820) 491,149 491,149 Net income (loss) Loans receivable \$46,291,468 \$15,730,713 \$22,770,803 \$(25,301) \$84,767,683 \$(45,915,920) \$38,851,763

## For the Three Months Ended September 30, 2004

			Global Financial		Total	Securitization	Total
	U.S. Card	Auto Finance	Services	Other	Managed	Adjustments	Reported
Net interest income	\$ 1,172,447	\$ 205,385	\$ 361,165	\$(68,630)	\$ 1,670,367	\$ (894,992)	\$ 775,375
Non-interest income	811,465	20,926	240,597	26,785	1,099,773	439,611	1,539,384
Provision for loan losses	503,179	56,483	150,921	12,593	723,176	(455,381)	267,795
Non-interest expenses	833,183	83,401	322,552	72,848	1,311,984		1,311,984
Income tax provision (benefit)	233,118	31,114	41,445	(60,858)	244,819	_	244,819
-							
Net income (loss)	\$ 414,432	\$ 55,313	\$ 86,844	\$(66,428)	\$ 490,161	\$ —	\$ 490,161
Loans receivable	\$46,081,967	\$9,734,254	\$19,614,693	\$ 25,917	\$75,456,831	\$(40,296,196)	\$35,160,635

## For the Nine Months Ended September 30, 2005

			Global Financial		Total	Securitization	Total
	U.S. Card	Auto Finance	Services	Other	Managed	Adjustments	Reported
Net interest income	\$ 3,610,162	\$ 835,353	\$ 1,248,187	\$(113,444)	\$ 5,580,258	\$ (2,937,015)	\$ 2,643,243
Non-interest income	2,477,171	21,309	772,407	45,081	3,315,968	1,376,623	4,692,591
Provision for loan losses	1,512,006	297,862	662,113	13,809	2,485,790	(1,560,392)	925,398
Non-interest expenses	2,464,079	368,068	1,086,008	110,932	4,029,087	_	4,029,087
Income tax provision (benefit)	738,937	66,756	93,497	(46,670)	852,520	_	852,520
Net income (loss)	\$ 1,372,311	\$ 123,976	\$ 178,976	\$(146,434)	\$ 1,528,829	\$ —	\$ 1,528,829
	· /- /-			(( ', ', ',	, , , , , , ,	•	, ,= =,= =
Loans receivable	\$46,291,468	\$15,730,713	\$22,770,803	\$ (25,301)	\$84,767,683	\$(45,915,920)	\$38,851,763

## For the Nine Months Ended September 30, 2004

	U.S. Card	Auto Finance	Global Financial Services	Other	Total Managed	Securitization Adjustments	Total Reported
Net interest income	\$ 3,497,124	\$ 590,557	\$ 1,031,246	\$(186,012)	\$ 4,932,915	\$ (2,714,501)	\$ 2,218,414
Non-interest income	2,396,555	67,022	603,410	58,620	3,125,607	1,252,975	4,378,582
Provision for loan losses	1,558,027	191,573	463,359	2,286	2,215,245	(1,461,526)	753,719
Non-interest expenses	2,483,533	249,278	897,529	135,352	3,765,692		3,765,692
Income tax provision (benefit)	666,763	78,023	89,898	(105,453)	729,231	_	729,231
Net income (loss)	\$ 1,185,356	\$ 138,705	\$ 183,870	\$(159,577)	\$ 1,348,354	\$ —	\$ 1,348,354
Loans receivable	\$46,081,967	\$9,734,254	\$19,614,693	\$ 25,917	\$75,456,831	\$(40,296,196)	\$35,160,635

## **Significant Segment Adjustments**

## The Gulf Coast Hurricanes' Impacts

During the three months ended September 30, 2005, the Company recorded a \$28.5 million additional build in allowance for loan losses and recognized a \$15.6 million write-down on retained interests as a result of the Gulf Coast Hurricanes. Of the additional allowance build, \$10.0 million was allocated to the U.S. Card

segment, \$2.5 million was allocated to the Global Financial Services segment, and \$16.0 million was allocated to the Auto Finance segment. The \$15.6 million write-down of retained interests was held in the Other category.

#### **Bankruptcy Legislation Impacts**

The Bankruptcy Abuse Prevention and Consumer Protection Act of 2005 ("new bankruptcy legislation") became effective on October 17, 2005. As a result, the Company expected a significant increase in bankruptcy related charge-offs much of which was allowed for at the end of the second quarter. However, the magnitude of the unprecedented spike in bankruptcies experienced immediately before the new legislation became effective was larger than previously anticipated. As a result, the Company built additional allowance for loan losses of \$27.0 million and took a \$48.0 million write-down of retained interests related to the Company's loan securitization programs. These amounts were held in the Other category.

#### Non-recurring Operating Expense Items

During the three months ended September 30, 2005 and 2004, the Company recognized non-interest expense of \$16.2 million and \$26.7 million, respectively for employee termination and facility consolidation charges related to continued cost reduction initiatives. In addition, the Company recognized \$20.6 million related to a change in asset capitalization thresholds and \$15.8 million related to impairment of internally developed software during the three months ended September 30, 2004. Of these amounts, \$12.7 million and \$27.4 million was allocated to the U.S. Card segment, \$2.7 million and \$32.6 million was allocated to the Global Financial Services segment, \$0.3 and \$2.6 million was allocated to the Auto Finance Segment and the remainder was held in the Other category for the three months ended September 30, 2005 and 2004, respectively.

During the nine months ended September 30, 2005 and 2004, the Company recognized non-interest expense of \$65.9 million and \$82.7 million, respectively for employee termination and facility consolidation charges related to continued cost reduction initiatives. In addition, the Company recognized \$20.6 million related to a change in asset capitalization thresholds and \$15.8 million related to impairment of internally developed software during the nine months ended September 30, 2004. Of these amounts, \$40.4 million and \$77.3 million was allocated to the U.S. Card segment, \$15.8 million and \$37.7 million was allocated to the Global Financial Services segment, \$8.5 million and \$3.1 million was allocated to the Auto Finance Segment and the remainder was held in the Other category for the nine months ended September 30, 2005 and 2004, respectively.

In the first quarter 2005, the Company closed on the sale of its Tampa, Florida facilities. The ultimate sales price was greater than the impaired value of the held-for-sale property, and as such, the Company reversed \$18.8 million of its previously recorded impairment in Occupancy expense. Of this amount, \$17.4 million was allocated to the U.S. Card segment, \$1.3 million was allocated to the Global Financial Services segment, and the balance was held in the Other category.

In September 2004, the Company sold its interest in a South African joint venture with a book value of \$3.9 million to its joint venture partner. The Company received \$26.2 million in cash, was forgiven \$9.2 million in liabilities and recognized a pre-tax gain of \$31.5 million. The gain was recorded in non-interest income and reported in the Global Financial Services segment.

#### Sale of Auto Loans

During the three months ended September 30, 2005, the Company did not sell any auto loans. During the three months ended September 30, 2004, the Company sold auto loans of \$252.8 million. These transactions resulted in pre-tax gains allocated to the Auto Finance segment, inclusive of allocations related to funds transfer pricing of \$11.5 million for the three months ended September 30, 2004. In addition the company recognized an additional \$4.3 million in gains related to the settlement of contingencies from prior period sales of auto receivables for the three month period ended September 30, 2005.

During the nine months ended September 30, 2005 and 2004, the Company sold auto loans of \$257.7 and \$835.2 million, respectively. These transactions resulted in pre-tax gains allocated to the Auto Finance segment, inclusive of allocations related to funds transfer pricing of \$4.5 million and \$37.2 million for the nine months ended September 30, 2005 and 2004, respectively. In addition the company recognized an additional \$9.3 million in gains related to the settlement of contingencies from prior period sales of auto receivables for the nine month period ended September 30, 2005.

#### **Note 3: Capitalization**

In October 2005, the Company filed an updated shelf registration statement providing for the sale of senior or subordinated debt securities, preferred stock, common stock, common equity units, and stock purchase contracts, in an aggregate amount not to exceed \$2.5 billion.

In May 2005, the Company issued 10.4 million shares of common stock resulting in proceeds of \$747.5 million in accordance with the settlement provisions of the forward purchase contracts of its mandatory convertible debt securities (the "Upper DECs") issued in April of 2002. The number of shares was based on the average closing price of \$71.77 for the Company's stock calculated over the twenty trading days prior to the settlement.

In May 2005, the Company issued \$500.0 million of ten year 5.50% fixed rate senior notes through its shelf registration statement.

In March 2005, COAF entered into a revolving warehouse credit facility collateralized by a security interest in certain auto loan assets (the "Capital One Auto Loan Facility II"). The Capital One Auto Loan Facility II has the capacity to issue up to \$750.0 million in secured notes. The Capital One Auto Loan Facility II has a renewal date of March 27, 2006. The facility does not have a final maturity date. Instead, the participant may elect to renew the commitment for another set period of time. Interest on the facility is based on commercial paper rates.

In February of 2005, pursuant to the original terms of the Upper Decs mandatory convertible securities issued in April of 2002, the Company completed a remarketing of approximately \$704.5 million aggregate principal amount of its 6.25% senior notes due May 17, 2007. As a result of the remarketing, the annual interest rate on the senior notes was reset to 4.738%. Following the remarketing, the Company extinguished \$585.0 million of the remarketed senior notes using the proceeds from the issuance of \$300.0 million of seven year 4.80% fixed rate senior notes and \$300.0 million of twelve year 5.25% fixed rate senior notes. The Company recognized a \$12.4 million loss on the extinguishment of the remarketed senior notes.

#### **Note 4: Comprehensive Income**

Comprehensive income for the three months ended September 30, 2005 and 2004, respectively was as follows:

	Three Mon Septem	
	2005	2004
Comprehensive Income:		
Net income	\$491,149	\$490,161
Other comprehensive (loss) income, net of tax	(54,920)	110,178
Total comprehensive income	\$436,229	\$600,339

### **Note 5: Earnings Per Share**

The following table sets forth the computation of basic and diluted earnings per share:

	Three M Sep	Nine Months Ended September 30					
	2005		2004	2005			2004
Numerator:							
Net income	\$491,14	9 \$	490,161	\$1,52	28,829	\$ 1	,348,354
Denominator:							
Denominator for basic earnings per share - Weighted-average shares	260,91	3	236,405	25	52,556		234,393
Effect of dilutive securities:							
Stock options	6,98	L	9,925		7,498		10,517
Restricted stock	2,79	1	2,677		2,521		2,434
Dilutive potential common shares	9,77	5	12,602	1	10,019		12,951
Denominator for diluted earnings per share - Adjusted weighted-average shares	270,69	3	249,007	26	52,575		247,344
Basic earnings per share	\$ 1.8	3 \$	2.07	\$	6.05	\$	5.75
Diluted earnings per share	\$ 1.8	1 \$	5 1.97	\$	5.82	\$	5.45

## **Note 6: Goodwill and Other Intangible Assets**

The following table provides a summary of goodwill.

	Auto Finance	Global Financial Services	Total
Balance at December 31, 2004	\$218,957	\$133,200	\$352,157
Additions	109,235	281,470	390,705
Foreign currency translation	_	(6,804)	(6,804)
Balance at September 30, 2005	\$328,192	\$407,866	\$736,058

During the first quarter of 2005, the Company closed the acquisitions of Onyx Acceptance Corporation, a specialty auto loan originator; Hfs Group, a United Kingdom based home equity broker; InsLogic, an insurance brokerage firm, and eSmartloan, a U.S. based online originator of home equity loans and mortgages, which created approximately \$390.7 million of goodwill, in the aggregate.

The results of operations for the businesses acquired during the first quarter of 2005 were included in the Company's Consolidated Statement of Income beginning at the dates of acquisition.

#### **Note 7: Commitments and Contingencies**

#### **Industry Litigation**

Over the past several years, MasterCard International and Visa U.S.A., Inc., as well as several of their member banks, have been involved in several different lawsuits challenging various practices of MasterCard and Visa.

In 1998, the United States Department of Justice filed an antitrust lawsuit against the MasterCard and Visa membership associations composed of financial institutions that issue MasterCard or Visa credit or debit cards ("associations"), alleging, among other things, that the associations had violated antitrust law and engaged in unfair practices by not allowing member banks to issue cards from competing brands, such as American Express and Discover Financial Services. In 2001, a New York district court entered judgment in favor of the Department of Justice and ordered the associations, among other things, to repeal these policies. The United States Court of Appeals for the Second Circuit affirmed the district court and, on October 4, 2004, the United States Supreme Court denied certiorari in the case.

After the Supreme Court denied certiorari, American Express Travel Related Services Company, Inc., on November 15, 2004, filed a lawsuit against the associations and several member banks under United States federal antitrust law. The lawsuit alleges, among other things, that the associations and member banks implemented and enforced illegal exclusionary agreements that prevented member banks from issuing American Express and Discover cards. The complaints, among other things, request civil monetary damages, which could be trebled. Capital One Bank, Capital One, F.S.B. and Capital One Financial Corporation are named defendants in this lawsuit.

Between June 22, 2005 to the present date, a number of entities, each purporting to represent a class of retail merchants, filed five separate lawsuits against the associations and several member banks under United States federal antitrust law. The lawsuits allege, among other things, that the associations and member banks conspired to fix the level of interchange fees. The complaints, among other things, request civil monetary damages, which could be trebled. Capital One Bank, Capital One, F.S.B., and Capital One Financial Corporation, among others, are named defendants.

Capital One Bank, Capital One, F.S.B. and Capital One Financial Corporation believe that they have meritorious defenses with respect to these cases and intend to defend these cases vigorously. At the present time, management is not in a position to determine whether the resolution of these cases will have a material adverse effect on either the consolidated financial position of the Company or the Company's results of operations in any future reporting period.

In addition, several United States merchants have filed class action lawsuits, which have been consolidated, against the associations under United States federal antitrust law relating to certain debit card products. In April 2003, the associations agreed to settle the lawsuit in exchange for payments to plaintiffs by MasterCard of \$1 billion and Visa of \$2 billion, both over a ten-year period, and for changes in policies and interchange rates for debit cards. Certain merchant plaintiffs have opted out of the settlements and have commenced separate lawsuits. Additionally, consumer class action lawsuits with claims mirroring the merchants' allegations have been filed in several courts. Finally, the associations, as well as certain member banks, continue to face additional lawsuits regarding policies, practices, products and fees.

With the exception of the antitrust lawsuits brought by certain merchants between June 22, 2005 to the present date and the American Express antitrust lawsuit, the banks and their affiliates are not parties to the lawsuits against the associations described above and therefore will not be directly liable for any amount related to any possible or known settlements, the lawsuits filed by merchants who have opted out of the settlements of those lawsuits or the class action lawsuits pending in state and federal courts. However, the banks are member banks of MasterCard and Visa and thus may be affected by settlements or lawsuits relating to these issues. In addition, it is possible that the scope of these lawsuits may expand and that other member banks, including the banks, may be brought into the lawsuits or future lawsuits.

Given the complexity of the issues raised by the lawsuits described above and the uncertainty regarding: (i) the outcome of these suits, (ii) the likelihood and amount of any possible judgment against the associations or the member banks, (iii) the likelihood, amount and validity of any claim against the associations' member banks, including the banks and the Corporation, and (iv) the effects of these suits, in turn, on competition in the industry, member banks, and interchange and association fees, we cannot determine at this time the long-term effects of these suits on us.

#### Other Pending and Threatened Litigation

In addition, the Company also commonly is subject to various pending and threatened legal actions relating to the conduct of its normal business activities. In the opinion of management, the ultimate aggregate liability, if any, arising out of any such pending or threatened legal actions will not be material to the consolidated financial position or results of operations of the Company.

#### Note 8: Impairment or Disposal of Long-Lived of Assets

During the first quarter of 2005, the Company closed on the sale of its Tampa, Florida facilities. The Company had previously classified the property as held-for sale and recognized an impairment charge of \$44.9 million in accordance with FASB Statement No. 144, *Accounting for the Impairment or Disposal of Long-Lived Assets*. The ultimate sales price was greater than the recorded impaired value, and as such, the Company reversed \$18.8 million of its previously recorded impairment in Occupancy expense during the quarter.

#### **Note 9: Consolidation of Variable Interest Entities**

In July 2005, the Company established and consolidated Capital One Appalachian II LLC ("COAL II") which qualifies as a variable interest entity under the requirements of FASB Interpretation No. 46(R), Consolidation of Variable Interest Entities, an Interpretation of ARB No. 51 ("FIN 46"). COAL II purchased a limited interest in a limited liability company from a third party that operates a facility which produces a coal-based synthetic fuel that qualifies for tax credits pursuant to Section 29 of the Internal Revenue Code. COAL II paid \$450.0 thousand in cash and agreed to make fixed note payments, variable payments, and fund its share of operating costs in the amount of \$44.1 million. These payments will be based upon the amount of tax credits generated by the limited liability company from July 2005 through the end of 2007. COAL II has an ongoing commitment to fund the losses of the limited liability company to maintain its 8% minority ownership interest, and make fixed and variable payments to the third party. The Corporation has guaranteed COAL II's commitments to both the limited liability company and the third party. COAL II's equity investment in the limited liability company was included in Other Assets at September 30, 2005.

#### Note 10: Acquisition of Hibernia Corporation

The Company and Hibernia Corporation mutually agreed to delay the merger transaction originally scheduled to close in September of 2005 due to impacts of Hurricane Katrina and have negotiated a new acquisition price. As previously disclosed in the Post-effective Amendment No. 1 to Form S-4, the estimated \$5.2 billion merger with Hibernia Corporation is now expected to close on November 16, 2005, subject to Hibernia shareholders' approval of the amended merger agreement.

## Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations (Dollars in thousands) (yields and rates presented on an annualized basis)

#### I. Introduction

Capital One Financial Corporation (the "Corporation") is a diversified financial services company whose banking and non-banking subsidiaries market a variety of financial products and services. The Corporation's principal subsidiaries are Capital One Bank (the "Bank") which currently offers credit card products and takes retail deposits, Capital One, F.S.B. (the "Savings Bank"), which offers consumer and commercial lending and consumer deposit products, and Capital One Auto Finance, Inc. ("COAF") which offers automobile and other motor vehicle financing products. Capital One Services, Inc. ("COSI"), another subsidiary of the Corporation, provides various operating, administrative and other services to the Corporation and its subsidiaries. The Corporation and its subsidiaries are hereafter collectively referred to as the "Company". As of September 30, 2005, the Company had 49.2 million accounts and \$84.8 billion in managed loans outstanding and was one of the largest providers of MasterCard and Visa credit cards in the United States.

The Company's profitability is affected by the net interest income and non-interest income generated on earning assets, consumer usage and payment patterns, credit quality, levels of marketing expense and operating efficiency. The Company's revenues consist primarily of interest income on loans (including past-due fees) and securities and non-interest income consisting of servicing income on securitized loans, fees (such as annual membership, cash advance, overlimit and other fee income, collectively "fees"), cross-sell, interchange and gains on the securitizations of loans. Loan securitization transactions qualifying as sales under accounting principles generally accepted in the United States ("GAAP") remove the loan receivables from the consolidated balance sheet; however, the Company continues to both own and service the related accounts. The Company generates earnings from its "managed" loan portfolio that includes both on-balance sheet and off-balance sheet loans. Interest income, fees, and recoveries in excess of the interest paid to investors and charge-offs generated from off-balance sheet loans are recognized as servicing and securitizations income.

The Company's primary expenses are the costs of funding assets, provision for loan losses, operating expenses (including associate salaries and benefits), marketing expenses and income taxes. Marketing expenses (e.g., advertising, printing, credit bureau costs and postage) to implement the Company's product strategies are expensed as incurred while the revenues resulting from acquired accounts are recognized over their life.

#### **II. Significant Accounting Policies**

See the Corporation's Annual Report on Form 10-K for the year ended December 31, 2004, Part I, Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations" for a summary of the Company's significant accounting policies.

## III. Reconciliation to GAAP Financial Measures

The Company's consolidated financial statements prepared in accordance with GAAP are referred to as its "reported" financial statements. Loans included in securitization transactions which qualified as sales under GAAP have been removed from the Company's "reported" balance sheet. However, servicing fees, finance charges, and other fees, net of charge-offs, and interest paid to investors of securitizations are recognized as servicing and securitizations income on the "reported" income statement.

The Company's "managed" consolidated financial statements reflect adjustments made related to the effects of securitization transactions qualifying as sales under GAAP. The Company generates earnings from its "managed" loan portfolio which includes both the on-balance sheet loans and off-balance sheet loans. The Company's "managed" income statement takes the components of the servicing and securitizations income generated from the securitized portfolio and distributes the revenue and expense to appropriate income statement line items from which it originated. For this reason, the Company believes the "managed" consolidated financial statements and related managed metrics to be useful to stakeholders.

	As of and for the Three Months Ended September 30, 2005					
		Securitization	7(2)			
(Dollars in thousands) Income Statement Measures	Total Reported	Adjustments <sup>(1)</sup>	Total Managed <sup>(2)</sup>			
	<b>.</b>	A 4 000 000	<b>*</b> 4.004.40 <b>=</b>			
Net interest income	\$ 910,219	\$ 1,020,976	\$ 1,931,195			
Non-interest income	\$ 1,594,616	\$ (494,809)	\$ 1,099,807			
Total revenue	\$ 2,504,835	\$ 526,167	\$ 3,031,002			
Provision for loan losses	\$ 374,167	\$ 526,167	\$ 900,334			
Net charge-offs	\$ 341,821	\$ 526,167	\$ 867,988			
Balance Sheet Measures						
Consumer loans	\$ 38,851,763	\$ 45,915,920	\$ 84,767,683			
Total assets	\$ 60,424,517	\$ 45,318,131	\$ 105,742,648			
Average consumer loans	\$ 38,555,575	\$ 45,271,890	\$ 83,827,465			
Average earning assets	\$ 53,452,923	\$ 43,243,421	\$ 96,696,344			
Average total assets	\$ 59,203,532	\$ 44,709,269	\$ 103,912,801			
30+ day delinquencies	\$ 1,496,713	\$ 1,667,064	\$ 3,163,777			

- Includes adjustments made related to the effects of securitization transactions qualifying as sales under GAAP and adjustments made to reclassify to "managed" loans outstanding the collectible portion of billed finance charge and fee income on the investors' interest in securitized loans excluded from loans outstanding on the "reported" balance sheet in compliance with relevant disclosure guidance.
- The managed loan portfolio does not include auto loans which have been sold in whole loan sale transactions where the Company has retained servicing rights.

## **IV. Management Summary**

#### Summary of Quarter ending September 30, 2005

The third quarter of 2005 was marked by continued strength in profitability and diversified loan growth.

The Gulf Coast Hurricanes resulted in an estimated \$44.1 million in probable loan losses. Of the total estimated impact, \$28.5 million was reflected in provision for loan losses through an increase in the allowance for loan losses and \$15.6 million was reflected as a reduction of servicing and securitizations income through a write-down of retained interests related to the Company's loan securitization programs.

The Bankruptcy Abuse Prevention and Consumer Protection Act of 2005 ("new bankruptcy legislation") became effective on October 17, 2005. As a result, the Company expected a significant increase in bankruptcy related charge-offs much of which was allowed for at the end of the second quarter. However, the magnitude of the unprecedented spike in bankruptcies experienced immediately before the new legislation became effective was larger than previously anticipated. As a result, the Company built additional allowance for loan losses of \$27.0 million and took a \$48.0 million write-down of retained interests related to the Company's loan securitization programs.

For the three months ended September 30, 2005, net income of \$491.1 million was slightly higher than the same period in the prior year, while diluted earnings per share decreased 8% to \$1.81 per share compared to the same period in the prior year. The slight increase in net income was driven by growth in total revenues offset by increases in provision for loan losses, marketing spend, and operating expenses. Earnings per share growth was constrained by the increase in share count resulting from the second quarter issuance of shares in connection with the execution of the forward purchase contracts related to the mandatory convertible debt securities. Growth in total revenues reflects year over year growth of the Company's managed loan portfolio and increasing loan yields, offset by the reduction related to the Gulf Coast Hurricanes and increase in bankruptcies reflected through a write-down of retained interests related to loan securitization programs as discussed above. The increase in the provision for loan losses resulted from growth in the reported loan portfolio, the build in the allowance for loan losses related to the Gulf Coast Hurricanes and increase in bankruptcies discussed above, as well as higher net charge-offs. Marketing expense increased as the Company took advantage of identified opportunities in the third quarter. Although operating expenses increased for the quarter, operating expense as a percentage of average managed loans continued to decline, reflecting the improved operating efficiencies of the Company's return on managed average assets of 1.89 % reflects the strength of the Company's earnings.

The Company continues to achieve strong loan growth in its Auto Finance and Global Financial Services segments, which accounted for the majority of third quarter loan growth and represent 45% of managed loans at September 30, 2005 compared with 39% at September 30, 2004.

The Company continues to grow profitably while maintaining a strong balance sheet. Total assets continue to grow, capital ratios remain well above the regulatory "well capitalized" thresholds, and the Company continues to maintain significant levels of liquidity.

## Q3 2005 Significant Events

The Company and Hibernia Corporation mutually agreed to delay the merger transaction originally scheduled to close in September of 2005 due to impacts of Hurricane Katrina and have negotiated a new acquisition price. As previously disclosed in the Post-effective Amendment No. 1 to Form S-4, the estimated \$5.2 billion merger with Hibernia Corporation is now expected to close on November 16, 2005, subject to Hibernia shareholders' approval of the amended merger agreement.

#### V. Financial Summary

Table 1 provides a summary view of the consolidated income statement and selected metrics at and for the three and nine month periods ended September 30, 2005 and 2004.

#### TABLE 1 - FINANCIAL SUMMARY

		Three Mor Septen						Nine Months Ended September 30				
(Dollars in thousands)		2005		2004		Change		2005		2004		Change
Earnings (Reported):												
Net interest income	\$	910,219	\$	775,375	\$	134,844	\$ 2	,643,243	\$	2,218,414	\$	424,829
Non-interest income		1,594,616		1,539,384		55,232	4	,692,591		4,378,582		314,009
Total Revenue (1)		2,504,835		2,314,759		190,076	7	,335,834		6,596,996		738,838
Provision for loan losses		374,167		267,795		106,372		925,398		753,719		171,679
Marketing		343,708		317,653		26,055		932,501		826,638		105,863
Operating expenses		1,021,930		994,331		27,599	3	,096,586		2,939,054		157,532
Income before taxes		765,030		734,980		30,050	2	,381,349		2,077,585		303,764
Income taxes		273,881		244,819		29,062		852,520		729,231		123,289
Net income	\$	491,149	\$	490,161	\$	988	\$ 1	,528,829	\$	1,348,354	\$	180,475
Common Share Statistics:												
Basic EPS	\$	1.88	\$	2.07	\$	(0.19)	\$	6.05	\$	5.75	\$	0.30
Diluted EPS		1.81		1.97		(0.16)		5.82		5.45		0.37
Selected Balance Sheet Data:												
Reported loans (period end)	\$3	38,851,763	\$3	5,160,635	\$3	3,691,128	\$38	,851,763	\$3	5,160,635	\$	3,691,128
Managed loans (period end)	8	34,767,683	7	5,456,831	9	,310,852	84	,767,683	7	5,456,831		9,310,852
Reported loans (average)	3	38,555,575	3	4,772,489	3	3,783,086	38	,332,568	3	3,650,942		4,681,626
Managed loans (average)	8	33,827,465	7	4,398,301	9	,429,164	82	,654,676	7	2,631,076	1	0,023,600
Allowance for loan losses		1,447,000		1,395,000		52,000	1	,447,000		1,395,000		52,000
Selected Company Metrics (Reported):												
Return on average assets (ROA)		3.32%		3.81%		(0.49)		3.55%		3.61%		(0.06)
Return on average equity (ROE)		18.19		25.93		(7.74)		20.65		25.74		(5.09)
Net charge-off rate		3.55		3.43		0.12		3.46		3.77		(0.31)
30+ day delinquency rate		3.85		4.00		(0.15)		3.85		4.00		(0.15)
Net interest margin		6.81		6.56		0.25		6.78		6.47		0.31
Revenue margin		18.74		19.59		(0.85)		18.80		19.24		(0.44)
Selected Company Metrics (Managed):		1.000/		0.470/		(0.20)		2.020/		2.040/		(0.02)
Return on average assets (ROA)		1.89%		2.17%		(0.28)		2.02%		2.04%		(0.02)
Net charge-off rate		4.14		4.05		0.09		4.12		4.43		(0.31)
30+ day delinquency rate		3.73		3.90		(0.17)		3.73		3.90		(0.17)
Net interest margin		7.99		7.86		0.13		7.88		7.94		(0.06)

In accordance with the Company's finance charge and fee revenue recognition policy, the amounts billed to customers but not recognized as revenue were \$255.6 million, \$269.7 million, \$759.3 million and \$818.7 million for the three and nine months ended September 30, 2005 and 2004, respectively.

13.03

(0.49)

12.56

12.97

(0.41)

## Summary of Reported Statement of Income

Revenue margin

The following is a detailed description of the financial results reflected in Table 1 – Financial Summary. Additional information is provided in section XIV, Tabular Summary as detailed in the sections below.

12.54

The results for the three and nine month periods ended September 30, 2005 may not be indicative of the full year results. Seasonal fourth quarter build in the loan balances are generally accompanied by a build in the allowance for loan losses, and increased operating expenses. See Section X. Business Outlook for further

All quarterly comparisons are made between the three month period ended September 30, 2005 and the three month period ended September 30, 2004, unless otherwise indicated.

All year to date comparisons are made between the nine month period ended September 30, 2005 and the nine month period ended September 30, 2004, unless otherwise indicated.

#### Net Interest Income

Net interest income is comprised of interest income and past-due fees earned and deemed collectible from the Company's loans and income earned on investment securities, less interest expense on interest-bearing deposits, senior and subordinated notes and other borrowings.

For the three and nine month periods ended September 30, 2005, reported net interest income increased 17% and 19%, respectively. These increases were primarily the result of growth in reported average earning assets and increases in earning asset yields. The increases in the earning asset yields for the three and nine month periods ended September 30, 2005, respectively, were primarily the result of 28 and 10 basis point increases in the reported loan yields for the same periods. The increases in loan yields for those periods resulted from the overall rising interest rate environment combined with improved collectibility of finance charges billed to customers.

For additional information, see section XIV, Tabular Summary, Table A (Statements of Average Balances, Income and Expense, Yields and Rates) and Table B (Interest Variance Analysis).

#### Non-Interest Income

Non-interest income is comprised of servicing and securitizations income, service charges and other customer-related fees, interchange income, and other non-interest income.

For the three and nine month periods ended September 30, 2005, reported non-interest income increased 4% and 7%, respectively. The quarterly increase was primarily related to increases in servicing and securitizations income and interchange income, offset by slight decreases in service charges and other customer-related fees and other non-interest income. The year-to-date increase was primarily the result of an increase in servicing and securitizations income, service charges and other customer-related fees and interchange income.

Servicing and securitizations income represents servicing fees, excess spread and other fees derived from the off-balance sheet loan portfolio, adjustments to the fair value of retained interests derived through securitization transactions, as well as gains and losses resulting from securitizations and other sale transactions. Servicing and securitizations income increased 6% and 7% for the three and nine month periods ended September 30, 2005, respectively. These increases were primarily the result of 14% increases in both the third quarter and year-to-date average off-balance sheet loan portfolio and increases in the excess spread generated from the off-balance sheet portfolio, partially offset by the \$63.6 million decrease in retained interests as a result of the Gulf Coast Hurricanes and new bankruptcy legislation discussed above. In addition, businesses acquired during 2005 contributed \$9.2 million and \$36.7 million to servicing and securitizations income for the three and nine month periods ended September 30, 2005, respectively.

Service charges and other customer-related fees decreased 1% in the third quarter of 2005 and increased 6% for the nine months ended September 30, 2005 while quarter-to-date and year-to-date average reported loan portfolios increased 11% and 14%, respectively. The lower growth in service charges and other customer-related fee income when compared to reported loan growth is reflective of the reported loan growth being concentrated in its Auto Finance and Global Financial Services Segments that generate lower fee income.

Interchange income, net of costs related to the Company's rewards programs, increased 7% and 12% for the three and nine month periods ended September 30, 2005, respectively, primarily due to increased purchase volumes. Purchase volumes increased 21% and 18% for the three and nine month periods ended September 30, 2005. Costs related to the Company's rewards programs increased \$6.9 million and \$33.2 million for the three and nine month periods ended September 30, 2005, respectively. The increases in rewards expense were due to increases in purchase volumes and the continued expansion of the rewards programs.

Other non-interest income includes, among other items, gains and losses on sales of securities, gains and losses associated with hedging transactions, service provider revenue generated by the Company's healthcare finance business, gains on the sale of auto loans and mortgage loans and income earned related to purchased charged-off loan portfolios. Other non-interest income decreased 3% for the third quarter of 2005. This decrease was primarily related to the \$31.5 million pre-tax gain on the sale of the Company's joint venture investment in South Africa that occurred during the three months ended September 30, 2004 and a \$7.2 million reduction in gains on the sale of auto loans recognized in the third quarter of 2005. These decreases were partially offset by a \$19.8 million increase in revenue generated from the Company's mortgage business.

Other non-interest income increased 1% for the nine month period ended September 30, 2005. Growth in other non-interest income was offset by the \$31.5 million pre-tax gain on the sale of the Company's joint venture investment in South Africa that occurred during the nine month period ended September 30, 2004 and a \$25.0 million reduction in revenue due to fewer gains on the sale of auto loans recognized during the nine month period ended September 30, 2005.

#### Provision for loan losses

The provision for loan losses increased 40% and 23% for the three and nine month periods ended September 30, 2005, respectively. These increases were a result of a 10% growth in the reported loan portfolio, an increase in net charge-offs of \$43.5 million and \$45.6 million for the three and nine month periods ended September 30, 2005, respectively, and the \$55.5 million additional build in the allowance for loan losses as a result of the Gulf Coast Hurricanes and new bankruptcy legislation discussed above.

#### Non-interest expense

Non-interest expense consists of marketing and operating expenses. The 8% and 13% increases in marketing expense for the three and nine month periods ended September 30, 2005, respectively, reflect origination opportunities in new and existing products.

Operating expenses for the three and nine month periods ended September 30, 2005, included \$52.9 million and \$139.6 million, respectively, in expenses related to businesses acquired during the first quarter of 2005. These additional expenses were partially offset by a reduction of \$10.5 million and \$16.8 million in the employee termination and facility consolidation expenses related to continued corporate-wide cost reduction initiatives for the three and nine month periods ended September 30, 2005, compared to the same periods in the prior year. In addition, the Company recognized a \$15.8 million charge related to the impairment of internally developed software and a \$20.6 million charge related to a change in asset capitalization thresholds during the three and nine month periods ended September 30, 2004. Exclusive of the aforementioned items, operating expenses increased 2% and 3% for the three and nine month periods ended September 30, 2005, respectively. These increases were the result of 12% managed loan growth, offset by improved operating efficiencies. These improvements were evidenced by 47 and 40 basis point declines in operating expenses a a percentage of average managed loans for the three and nine months ended September 30, 2005, respectively.

#### Income taxes

The Company's effective tax rate was 35.8% for both the three and nine month periods ended September 30, 2005, compared to 33.3% and 35.1% for the same periods in the prior year. The slight increase in the effective tax rate was due to changes in the Company's international and state tax positions.

#### Loan Portfolio Summary

The Company analyzes its financial performance on a managed loan portfolio basis. The managed loan portfolio is comprised of on-balance sheet and off-balance sheet loans. The Company has retained servicing rights for its securitized loans and receives servicing fees in addition to the excess spread generated from the off-balance sheet loan portfolio.

The 13% and 14% growth in the average managed loan portfolio for the three and nine month periods ended September 30, 2005, respectively, was concentrated in the Auto Finance and Global Financial Services segments, which tend to create lower yields and lower losses, when compared to the same periods in the prior year. Auto Finance and Global Financial Services together contributed 93% and 86% of the average managed loan growth for the respective periods.

For additional information, see section XIV, Tabular Summary, Table C (Managed Consumer Loan Portfolio).

## Asset Quality

The Company's credit risk profile is managed to maintain strong risk adjusted returns and diversification across the full credit spectrum and in each of its consumer lending products. Certain consumer lending products have, in some cases, higher expected delinquency and charge-off rates. The costs associated with higher delinquency and charge-off rates are considered in the pricing of individual products.

#### **Delinquencies**

The Company believes delinquencies to be an indicator of loan portfolio credit quality at a point in time. The entire balance of an account is contractually delinquent if the minimum payment is not received by the payment due date. Delinquencies not only have the potential to impact earnings if the account charges off, but they also result in additional costs in terms of the personnel and other resources dedicated to resolving the delinquencies.

The reduction in reported and managed 30+ day delinquency rates was due to the Company's continued asset diversification away from U.S. credit cards to collateralized and other assets with higher credit quality and improved collections experience.

For additional information, see section XIV, Tabular Summary, Table D (Delinquencies).

#### Net Charge-Offs

Net charge-offs include the principal amount of losses (excluding accrued and unpaid finance charges, fees and fraud losses) less current period principal recoveries. The Company generally charges off credit card loans at 180 days past the due date, and charges off other consumer loans at the earlier of 120 days past the due date or upon repossession of collateral. Costs to recover previously charged-off accounts are recorded as collection expenses in non-interest expense. Non-collateralized bankruptcies are typically charged off within 2-7 days upon notification and in any event within 30 days.

The increases in reported and managed net charge-off rates for the third quarter were due to an increase in bankruptcy related charge-offs driven primarily by the new bankruptcy legislation that became effective in October of 2005. Managed charge-offs related to bankruptcies increased \$91.7 million for the three month period ended September 30, 2005.

The reported and managed net charge-off rates for the nine month period decreased as a result of the Company's continued asset diversification beyond U.S. credit cards with a continued focus on originating higher credit quality loans and improved collections experience. This decrease was partially offset by an increase in the nine month bankruptcy related managed charge-offs of \$134.0 million driven by the new bankruptcy legislation that became effective in October of 2005.

For additional information, see section XIV, Tabular Summary, Table E (Net Charge-offs).

#### Allowance For Loan Losses

The allowance for loan losses is maintained at an amount estimated to be sufficient to absorb probable losses, net of principal recoveries (including recovery of collateral), inherent in the existing reported loan portfolio. The provision for loan losses is the periodic cost of maintaining an adequate allowance. Management believes that, for all relevant periods, the allowance for loan losses was adequate to cover anticipated losses in the total reported consumer loan portfolio under then current conditions, met applicable legal and regulatory guidance and was consistent with GAAP. There can be no assurance as to future credit losses that may be incurred in connection with the Company's consumer loan portfolio, nor can there be any assurance that the loan loss allowance that has been established by the Company will be sufficient to absorb such future credit losses. The allowance is a general allowance applicable to the reported consumer loan portfolio. The amount of allowance necessary is determined primarily based on a migration analysis of delinquent and current accounts and forward loss curves. In evaluating the sufficiency of the allowance for loan losses, management also takes into consideration the following factors: recent trends in delinquencies and charge-offs including bankrupt, deceased and recovered amounts; forecasting uncertainties and size of credit risks; the degree of risk inherent in the composition of the loan portfolio; economic conditions; legal and regulatory guidance; credit evaluations and underwriting policies; seasonality; and the value of collateral supporting the loans.

The allowance for loan losses increased \$42.0 million since June 30, 2005 driven primarily by the \$55.5 million additional increase in potential losses resulting from the Gulf Coast Hurricanes and the new bankruptcy legislation. Absent the impact from the Gulf Coast Hurricanes and the new bankruptcy legislation, the allowance would have decreased due to a change in the mix of the reported loan portfolio to a higher concentration of auto loans, which tend to generate losses considerably below the average loss rate for the Company's overall portfolio.

For additional information, see section XIV, Tabular Summary, Table F (Summary of Allowance for Loan Losses).

#### Finance Charge and Fee Revenue Recognition

The Company recognizes earned finance charges and fee income on loans according to the contractual provisions of the credit arrangements. When the Company does not expect full payment of finance charges and fees, it does not accrue the estimated uncollectible portion as income (hereafter the "suppression amount"). To calculate the suppression amount, the Company first estimates the uncollectible portion of finance charge and fee receivables using a formula based on historical account migration patterns and current delinquency status. This formula is consistent with that used to estimate the allowance related to expected principal losses on reported loans. The suppression amount is calculated by adding any current period change in the estimate of the uncollectible portion of finance charge and fee receivables to the amount of finance charges and fees charged-off (net of recoveries) during the period. The Company subtracts the suppression amount from the total finance charges and fees billed during the period to arrive at total reported revenue.

The amount of finance charges and fees suppressed were \$255.6 million and \$759.3 million for the three and nine months ended September 30, 2005, respectively, compared to \$269.7 million and \$818.7 million for the three and nine months ended September 30, 2004, respectively. The reduction in the suppression amount was driven by higher expectations of collectibility resulting from improved credit performance. Actual payment experience could differ significantly from management's assumption, resulting in higher or lower future finance charge and fee income.

#### VI. Reportable Segment Summary

The Company manages its business as three distinct operating segments: U.S. Card, Auto Finance and Global Financial Services. The U.S. Card, Auto Finance and Global Financial Services segments are considered reportable segments based on quantitative thresholds applied to the managed loan portfolio for reportable segments provided by SFAS No. 131, *Disclosures about Segments of an Enterprise and Related Information*.

As management makes decisions on a managed portfolio basis within each segment, information about reportable segments is provided on a managed basis.

The Company maintains its books and records on a legal entity basis for the preparation of financial statements in conformity with GAAP. The following table presents information prepared from the Company's internal management information system, which is maintained on a line of business level through allocations from legal entities.

All quarterly comparisons are made between the three month period ended September 30, 2005 and the three month period ended September 30, 2004, unless otherwise indicated. All year to date comparisons are made between the nine month period ended September 30, 2005 and the nine month period ended September 30, 2004, unless otherwise indicated.

## **US Card Segment**

#### TABLE 2 - U.S. CARD

	As of and for the Ended Sept		As of and for the Ended Sept	
(Dollars in thousands)	2005	2004	2005	2004
Earnings (Managed Basis)				
Net interest income	\$ 1,207,832	\$ 1,172,447	\$ 3,610,162	\$ 3,497,124
Non-interest income	851,036	811,465	2,477,171	2,396,555
Total revenue	2,058,868	1,983,912	6,087,333	5,893,679
Provision for loan losses	483,759	503,179	1,512,006	1,558,027
Non-interest expense	833,925	833,183	2,464,079	2,483,533
Income before taxes	741,184	647,550	2,111,248	1,852,119
Income taxes	259,414	233,118	738,937	666,763
Net income	\$ 481,770	\$ 414,432	\$ 1,372,311	\$ 1,185,356
Selected Metrics (Managed Basis)				
Period end loans	\$46,291,468	\$46,081,967	\$46,291,468	\$46,081,967
Average loans	46,405,569	45,729,569	46,817,749	45,487,504
Net charge-off rate	4.69%	4.68%	4.77%	5.09%
30+ day delinquency rate	3.86	4.14	3.86	4.14

The U.S. Card segment consists of domestic consumer credit card lending activities. The U.S. Card segment continued to provide earnings growth primarily as a result of year over year loan growth, higher loan yields and purchase volumes, improved credit quality and improved operating efficiencies.

Third quarter U.S. Card segment net income grew as a result of higher revenue and lower provision for loan losses. Total revenues grew 4% in the third quarter, driven primarily by increases in loan yields and higher purchase volumes.

The provision for loan losses decreased 4% for the three month period ended September 30, 2005, reflecting continued strong credit quality metrics, despite an increase in bankruptcy related charge-offs as a result of the rush to file before the new bankruptcy legislation took effect on October 17, 2005 and a \$10.0 million build in the allowance for loan losses as a result of the Gulf Coast Hurricanes allocated to U.S. Card.

The increase in net income for the nine months ended September 30, 2005 was the result of higher revenue, lower provision for loan losses and lower non-interest expense. Total revenue grew 3%, in the nine months ended September 30, 2005, as a result of 3% growth in the average loan portfolio coupled with improved loan yields and higher purchase volumes.

The provision for loan losses decreased 3% for the nine month period ended September 30, 2005 as a result of improvements in the credit quality metrics, despite an increase in bankruptcy related charge-offs as a result of the rush to file before the new bankruptcy legislation took effect on October 17, 2005 and the \$10.0 million build in the allowance as a result of the Gulf Coast Hurricanes allocated to U.S. Card.

Non-interest expense decreased 1% in the nine months ended September 30, 2005 primarily as a result of a \$36.9 million reduction in the allocation of severance and facility consolidations charges related to the Company's corporate-wide cost reduction initiatives, as well as, the charge taken in 2004 related to the change in fixed asset capitalization thresholds.

#### Auto Finance Segment

#### **TABLE 3 – AUTO FINANCE**

	As of and for the Three Months Ended September 30,					As of and for the Ended Septe			
(Dollars in thousands)		2005		2004		2005		2004	
Earnings (Managed Basis)									
Net interest income	\$	300,102	\$	205,385	\$	835,353	\$	590,557	
Non-interest income		3,005		20,926		21,309		67,022	
Total revenue		303,107		226,311		856,662		657,579	
Provision for loan losses		185,219		56,483		297,862		191,573	
Non-interest expense		129,719		83,401		368,068		249,278	
(Loss) income before taxes		(11,831)		86,427		190,732		216,728	
Income taxes		(4,141)		31,114		66,756		78,023	
Net (loss) income	\$	(7,690)	\$	55,313	\$	123,976	\$	138,705	
Selected Metrics (Managed Basis)									
Period end loans	\$15	5,730,713	\$9,	734,254	\$1	5,730,713	\$9	,734,254	
Average loans	15	5,104,464	9,	623,676	1	3,946,063	9	,112,398	
Net charge-off rate		2.54%		2.63%		2.38%		3.07%	
30+ day delinquency rate		4.65		5.54		4.65		5.54	

The Auto Finance segment consists of automobile and other motor vehicle financing activities. The Auto Finance segment's loan portfolio had significant year over year loan growth as a result of its 2005 acquisitions of Onyx Acceptance Corporation and the Key Bank non-prime auto loan portfolio, as well as, strong originations growth due to auto manufacturers' employee-pricing initiatives in the third quarter 2005.

Auto Finance net income decreased for the three and nine month periods ended September 30, 2005, as a result of increases in the provision for loan losses and non-interest expense, offset by increases in revenue. The 34% and 30% increases in revenue for the three and nine month periods ended September 30, 2005, respectively, were due primarily to 57% and 53% increases in the quarter-to-date and year-to-date average loan portfolios, offset by decreases in gains on the sale of auto loans compared to the same periods in the prior year.

The provision for loan losses increased for both the three and nine month periods ended September 30, 2005, as a result of the significant loan growth in the portfolio, a \$16.0 million build in the allowance for loan losses as a result of the Gulf Coast Hurricanes allocated to Auto Finance, and the return to normal charge-off

levels following an unusually low charge-off rate experienced in the second quarter of 2005. Despite the higher provision expense, credit quality metrics continue to improve as a result of improved credit conditions compared to the prior year.

Non-interest expense increased 56% and 48% for the three and nine month periods ended September 30, 2005, respectively, which is consistent with the growth in the portfolio.

#### Global Financial Services Segment

#### Table 4 - GLOBAL FINANCIAL SERVICES

	As of and for the Three Months Ended September 30,						the Nine Months ptember 30,		
(Dollars in thousands)		2005		2004		2005		2004	
Earnings (Managed Basis)									
Net interest income	\$	423,629	\$	361,165	\$	1,248,187	\$	1,031,246	
Non-interest income		273,067		240,597		772,407		603,410	
Total revenue		696,696		601,762		2,020,594		1,634,656	
Provision for loan losses		217,032		150,921		662,113		463,359	
Non-interest expense		356,254		322,552		1,086,008		897,529	
Income before taxes		123,410		128,289		272,473		273,768	
Income taxes		41,521		41,445		93,497		89,898	
Net income	\$	81,889	\$	86,844	\$	178,976	\$	183,870	
Selected Metrics (Managed Basis)									
Period end loans	\$2	2,770,803	\$1	9,614,693	\$2	2,770,803	\$1	9,614,693	
Average loans	2	2,373,995	1	9,088,779	2	1,903,815	1	8,082,029	
Net charge-off rate		4.09%		3.26%		3.85%		3.43%	
30+ day delinquency rate		2.93		2.65		2.93		2.65	

The Global Financial Services segment consists of international lending activities, small business lending, installment loans, home loans, healthcare financing and other consumer financial service activities.

Global Financial Services net income decreased slightly for the three and nine month periods ended September 30, 2005 as a result of increases in provision for loan losses and non-interest expense and a one-time gain on the sale of the Company's joint venture investment in South Africa that occurred during the third quarter 2004. Total revenue however increased 16% and 24% for the three and nine month period ended September 30, 2005 as a result of a 17% and 21% growth in average loans for the same periods, respectively, and contributions from the GFS businesses acquired in the first quarter of 2005.

The provision for loan losses increased 44% and 43% for the three and nine month periods ended September 30, 2005, respectively, as a result of growth in the loan portfolio combined with deteriorating credit quality metrics in the U.K, and a build in the allowance for loan losses of \$2.5 million as a result of the Gulf Coast Hurricanes allocated to Global Financial Services.

Non-interest expense, exclusive of a \$29.9 and \$21.9 million reduction in employee termination and facility consolidation charges associated with the Company's continued cost reduction initiatives for the three and nine month periods ended September 30, 2005, respectively, as well as, reductions in charges associated with the change in fixed asset capitalization thresholds and impairment of internally developed software that occurred in the third quarter 2004, increased 22% and 24% for the three and nine month periods ended September 30, 2005, primarily as a result of loan growth, offset in part by increased operating efficiencies .

## VII. Funding

The Company has established access to a variety of funding sources. Table 5 illustrates the Company's unsecured funding sources and its two auto securitization warehouses.

#### TABLE 5 - FUNDING AVAILABILITY AS OF SEPTEMBER 30, 2005

(Dollars or dollar equivalents in millions)	Effective/ Issue Date	Availa	ability <sup>(1) (5)</sup>	0	tstanding	Final Maturity <sup>(4)</sup>
Senior and Subordinated Global Bank Note Program <sup>(2)</sup>	1/03	\$	1,800	\$	4,163	—
Senior Domestic Bank Note Program <sup>(3)</sup>	4/97		_	\$	213	_
Credit Facility	6/04	\$	750		_	6/07
Capital One Auto Loan Facility I	_	\$	3,357	\$	993	_
Capital One Auto Loan Facility II	3/05	\$	750		_	_
Corporation Shelf Registration	7/05	\$	843		N/A	_

- (1) All funding sources are non-revolving except for the Credit Facility and the Capital One Auto Loan Facilities. Funding availability under the credit facilities is subject to compliance with certain representations, warranties and covenants. Funding availability under all other sources is subject to market conditions.
- (2) The notes issued under the Senior and Subordinated Global Bank Note Program may have original terms of thirty days to thirty years from their date of issuance. This program was updated in June 2005.
- (3) The notes issued under the Senior Domestic Bank Note Program have original terms of one to ten years. The Senior Domestic Bank Note Program is no longer available for issuances.
- (4) Maturity date refers to the date the facility terminates, where applicable.
- (5) Availability does not include unused conduit capacity related to securitization structures of \$4.7 billion at September 30, 2005.

The Senior and Subordinated Global Bank Note Program gives the Bank the ability to issue securities to both U.S. and non-U.S. lenders and to raise funds in U.S. and foreign currencies, subject to conditions customary in transactions of this nature.

Prior to the establishment of the Senior and Subordinated Global Bank Note Program, the Bank issued senior unsecured debt through an \$8.0 billion Senior Domestic Bank Note Program. The Bank did not renew the Senior Domestic Bank Note Program for future issuances following the establishment of the Senior and Subordinated Global Bank Note Program.

In June 2004, the Company terminated its Domestic Revolving and Multicurrency Credit Facilities and replaced them with a new revolving credit facility ("Credit Facility") providing for an aggregate of \$750.0 million in unsecured borrowings from various lending institutions to be used for general corporate purposes. The Credit Facility is available to the Corporation, the Bank, the Savings Bank, and Capital One Bank (Europe), plc, subject to covenants and conditions customary in transactions of this type. The Corporation's availability was increased to \$500.0 million under the Credit Facility. All borrowings under the Credit Facility are based upon varying terms of London Interbank Offering Rate ("LIBOR").

In April 2002, COAF entered into a revolving warehouse credit facility collateralized by a security interest in certain auto loan assets (the "Capital One Auto Loan Facility I"). As of September 30, 2005, the Capital One Auto Loan Facility I had the capacity to issue up to \$4.4 billion in secured notes. The Capital One Auto Loan Facility I has multiple participants each with separate renewal dates. The facility does not have a final maturity date. Instead, each participant may elect to renew the commitment for another set period of time. Interest on the facility is based on commercial paper rates.

In March 2005, COAF entered into a revolving warehouse credit facility collateralized by a security interest in certain auto loan assets (the "Capital One Auto Loan Facility II"). As of September 30, 2005, the Capital One Auto Loan Facility II had the capacity to issue up to \$750.0 million in secured notes. The Capital One Auto Loan Facility II has a renewal date of March 27, 2006. The facility does not have a final maturity date. Instead, the participant may elect to renew the commitment for another set period of time. Interest on the facility is based on commercial paper rates.

In October 2005, the Company filed an updated shelf registration statement providing for the sale of senior or subordinated debt securities, preferred stock, common stock, common equity units, and stock purchase contracts, in an aggregate amount not to exceed \$2.5 billion.

As of September 30, 2005, the Corporation had one effective shelf registration statement under which the Corporation from time to time may offer and sell senior or subordinated debt securities, preferred stock, common stock, common equity units and stock purchase contracts.

On May 17, 2005, the Company issued 10.4 million shares of common stock in accordance with the settlement provisions of the forward purchase contracts related to the mandatory convertible debt securities (the "Upper DECs") issued in April of 2002. The issuance provided \$747.5 million in cash proceeds.

In May 2005, the Company issued \$500.0 million of ten year 5.50% fixed rate senior notes through its shelf registration.

In February 2005, the Company completed a remarketing of approximately \$704.5 million aggregate principal amount of its 6.25% senior notes due May 17, 2007. As a result of the remarketing, the annual interest rate on the senior notes was reset to 4.738%. The remarketing was conducted pursuant to the original terms of the Uppers Decs mandatory convertible securities issued in April of 2002. Subsequently in February 2005, the Company extinguished \$585.0 million principal amount of the remarketed senior notes and issued in its place \$300.0 million of seven year 4.80% fixed rate senior notes and \$300.0 million of twelve year 5.25% fixed rate senior notes.

The Company continues to expand its retail deposit gathering efforts through both direct and broker marketing channels. The Company uses its data analysis capabilities to test and market a variety of retail deposit origination strategies, including via the Internet, as well as to develop customized account management programs. As of September 30, 2005, the Company had \$26.8 billion in interest-bearing deposits, of which \$2.5 billion were held in foreign banking offices and \$9.8 billion represented large denomination certificates of \$100 thousand or more with original maturities up to ten years.

Table 6 shows the maturities of domestic time certificates of deposit in denominations of \$100 thousand or greater (large denomination CDs) as of September 30, 2005.

#### TABLE 6- MATURITIES OF LARGE DENOMINATION CERTIFICATES-\$100,000 OR MORE

Sentember 30, 2005

	осресност в	70, 2005
(Dollars in thousands)	Balance	Percent
Three months or less	\$1,279,003	13.08%
Over 3 through 6 months	908,851	9.29
Over 6 through 12 months	1,522,008	15.56
Over 12 months through 10 years	6,070,535	62.07
Total	\$9,780,397	100.00%

#### **VIII. Off-Balance Sheet Arrangements**

## Off-Balance Sheet Securitizations

The Company actively engages in off-balance sheet securitization transactions of loans for funding purposes. The Company receives proceeds from third party investors for securities issued from the Company's securitization vehicles, which are collateralized by transferred receivables from the Company's portfolio. Securities outstanding totaling \$45.5 billion as of September 30, 2005, represent undivided interests in the pools of consumer loan receivables that are sold in underwritten offerings or in private placement transactions.

The securitization of consumer loans is a significant source of liquidity for the Company. Maturity terms of the existing securitizations vary from 2005 to 2025 and, for revolving securitizations, have accumulation periods during which principal payments are aggregated to make payments to investors. As payments on the loans are accumulated and are no longer reinvested in new loans, the Company's funding requirements for such new loans increase accordingly. The Company believes that it has the ability to continue to utilize off-balance sheet securitization arrangements as a source of liquidity; however, a significant reduction or termination of the Company's off-balance sheet securitizations could require the Company to draw down existing liquidity and/or to obtain additional funding through the issuance of secured borrowings or unsecured debt, the raising of additional deposits or the slowing of asset growth to offset or to satisfy liquidity needs.

#### Recourse Exposure

The credit quality of the receivables transferred is supported by credit enhancements, which may be in various forms including interest-only strips, subordinated interests in the pool of receivables, cash collateral accounts, cash reserve accounts and accrued interest and fees on the investor's share of the pool of receivables. Some of these credit enhancements are retained by the seller and are referred to as retained residual interests. The Company's retained residual interests are generally restricted or subordinated to investors' interests and their value is subject to substantial credit, repayment and interest rate risks on transferred assets if the off-balance sheet loans are not paid when due. Securitization investors and the trusts only have recourse to the retained residual interests, not the Company's assets. See the Corporation's Annual Report on Form 10-K for the year ended December 31, 2004, Part I, Item 8 "Financial Statements and Supplementary Data—Notes to the Consolidated Financial Statements—Note 18" for quantitative information regarding retained interests.

#### Collections and Amortization

Collections of interest and fees received on securitized receivables are used to pay interest to investors, servicing and other fees, and are available to absorb the investors' share of credit losses. For revolving securitizations, amounts collected in excess of that needed to pay the above amounts are remitted, in general, to the Company. Under certain conditions, some of the cash collected may be retained to ensure future payments to investors. For amortizing securitizations, amounts collected in excess of the amount that is used to pay the above amounts are generally remitted to the Company, but may be paid to investors in further reduction of their outstanding principal. See the Corporation's Annual Report on Form 10-K for the year ended December 31, 2004, Part I, Item 8 "Financial Statements and Supplementary Data—Notes to the Consolidated Financial Statements—Note 18" for quantitative information regarding revenues, expenses and cash flows that arise from securitization transactions.

Securitization transactions may amortize earlier than scheduled due to certain early amortization triggers, which would accelerate the need for funding. Additionally, early amortization would have a significant impact on the ability of the Bank and Savings Bank to meet regulatory capital adequacy requirements as all off-balance sheet loans experiencing such early amortization would be recorded on the balance sheet and accordingly would require incremental regulatory capital. As of September 30, 2005, no early amortization events related to its off-balance sheet securitizations have occurred.

#### Funding Commitments Related to Synthetic Fuel Tax Credit Transaction

In June of 2004 and July of 2005, the Company, through two separate transactions and two consolidated special purpose entities (SPVs), purchased minority ownership interests in two entities established to operate facilities which produce a coal-based synthetic fuel that qualifies for tax credits pursuant to Section 29 of the Internal Revenue Code. The SPVs purchased their minority interests from third parties paying \$2.6 million in cash and agreeing to pay an estimated \$159.1 million comprised of fixed note payments, variable payments and the funding of their share of operating losses sufficient to maintain their minority ownership percentages. Actual total payments will be based on the amount of tax credits generated through the end of 2007. In exchange, the SPVs will receive an estimated \$192.0 million in tax benefits resulting from a combination of deductions, allocated operating losses, and tax credits. The Corporation has guaranteed the SPVs commitments under the purchase agreements. As of September 30, 2005, the Company has recorded \$52.3 million in tax benefits and had an estimated remaining commitment for fixed note payments, variable payments and the funding of their proportionate share of the operating losses totaling \$116.7 million.

#### IX. Capital

#### Capital Adequacy

The Company and the Bank are subject to capital adequacy guidelines adopted by the Federal Reserve Board (the "Federal Reserve") while the Savings Bank is subject to capital adequacy guidelines adopted by the Office of Thrift Supervision (the "OTS") (collectively, the "regulators"). The capital adequacy guidelines require the Company, the Bank and the Savings Bank to maintain specific capital levels based upon quantitative measures of their assets, liabilities and off-balance sheet items. In addition, the Bank and Savings Bank must also adhere to the regulatory framework for prompt corrective action.

The most recent notifications received from the regulators categorized the Bank and the Savings Bank as "well-capitalized." As of September 30, 2005, the Company's, the Bank's and the Savings Bank's capital exceeded all minimum regulatory requirements to which they were subject, and there were no conditions or events since the notifications discussed above that management believes would have changed either the Company, the Bank or the Savings Bank's capital category.

#### **TABLE 7 - REGULATORY CAPITAL RATIOS**

	Regulatory Filing Basis Ratios	Applying Subprime Guidance Ratios	Minimum for Capital Adequacy Purposes	To Be "Well-Capitalized" Under Prompt Corrective Action Provisions
September 30, 2005				
Capital One Financial Corp (1)				
Tier 1 Capital	19.93%	N/A	4.00%	N/A
Total Capital	22.40	N/A	8.00	N/A
Tier 1 Leverage	17.89	N/A	4.00	N/A
Capital One Bank				
Tier 1 Capital	13.46%	11.01%	4.00%	6.00%
Total Capital	17.48	14.49	8.00	10.00
Tier 1 Leverage	10.77	10.77	4.00	5.00
Capital One, F.S.B.				
Tier 1 Capital	13.68%	11.58%	4.00%	6.00%
Total Capital	14.96	12.86	8.00	10.00
Tier 1 Leverage	13.33	13.33	4.00	5.00
September 30, 2004				
Capital One Bank				
Tier 1 Capital	14.91%	12.05%	4.00%	6.00%
Total Capital	18.99	15.55	8.00	10.00
Tier 1 Leverage	12.18	12.18	4.00	5.00
Capital One, F.S.B.				
Tier 1 Capital	15.71%	13.03%	4.00%	6.00%
Total Capital	17.01	14.31	8.00	10.00
Tier 1 Leverage	14.25	14.25	4.00	5.00

<sup>(1)</sup> The regulatory framework for prompt corrective action is not applicable for bank holding companies.

The Bank and Savings Bank treat a portion of their loans as "subprime" under the "Expanded Guidance for Subprime Lending Programs" (the "Subprime Guidelines") issued by the four federal banking agencies that comprise the Federal Financial Institutions Examination Council ("FFIEC"), and have assessed their capital and allowance for loan losses accordingly. Under the Subprime Guidelines, the Bank and Savings Bank each exceed the minimum capital adequacy guidelines as of September 30, 2005. Failure to meet minimum capital requirements can result in mandatory and possible additional discretionary actions by the regulators that, if undertaken, could have a material effect on the Company's consolidated financial statements.

For purposes of the Subprime Guidelines, the Company has treated as subprime all loans in the Bank's and the Savings Bank's targeted "subprime" programs to customers either with a FICO score of 660 or below or with no FICO score. The Bank and the Savings Bank hold on average 200% of the total risk-based capital charge that would otherwise apply to such assets. This results in higher levels of regulatory capital at the Bank and the Savings Bank.

Additionally, regulatory restrictions exist that limit the ability of the Bank and Savings Bank to transfer funds to the Corporation. As of September 30, 2005, retained earnings of the Bank and the Savings Bank of \$266.4 million and \$437.5 million, respectively, were available for payment of dividends to the Corporation without prior approval by the regulators.

#### **Dividend Policy**

Although the Company expects to reinvest a substantial portion of its earnings in its business, the Company also intends to continue to pay regular quarterly cash dividends on its common stock. The declaration and payment of dividends, as well as the amount thereof, are subject to the discretion of the Board of Directors of the Company and will depend upon the Company's results of operations, financial condition, cash requirements, future prospects and other factors deemed relevant by the Board of Directors. Accordingly, there can be no assurance that the Corporation will declare and pay any dividends. As a holding company, the ability of the Corporation to pay dividends is dependent upon the receipt of dividends or other payments from its subsidiaries. Applicable banking regulations and provisions that may be contained in borrowing agreements of the Corporation or its subsidiaries may restrict the ability of the Corporation's subsidiaries to pay dividends to the Corporation or the ability of the Corporation to pay dividends to its stockholders.

#### X. Business Outlook

This business outlook section summarizes the Company's expectations for earnings for 2005, and its primary goals and strategies for continued growth. The statements contained in this section are based on management's current expectations and do not take into account any acquisitions, including the pending acquisition of Hibernia (except for diluted earnings per share guidance), that might occur during the year. Certain statements are forward looking, and therefore actual results could differ materially from those in the Company's forward looking statements. Factors that could materially influence results are set forth throughout this section and below in the Risk Factors section.

#### **Expected Earnings**

The Company expects diluted earnings per share results likely to be on the lower end of the \$6.60 to \$7.00 range, inclusive of the acquisition of Hibernia Corporation, which represents an increase of between 6% and 13% over its diluted earnings per share of \$6.21 in 2004. The expected diluted earnings per share results for 2005 include the impact of issuing 10.4 million shares of common stock under the forward purchase contracts in May 2005 related to the Upper DECs mandatory convertible securities issued in April of 2002, as well as pro rata Institutional Brokers' Estimate System earnings for Hibernia and the corresponding 37.7 million shares consideration for the acquisition, which is expected to close on November 16, 2005. The Company's earnings are a function of its revenues (net interest income and non-interest income), consumer usage, payment and attrition patterns, the credit quality and growth rate of its earning assets (which affect fees, charge-offs and provision expense) and the Company's marketing and operating expenses. Specific factors likely to affect the Company's 2005 earnings are the portion of its loan portfolio it holds in higher credit quality assets, the level of off-balance sheet securitizations, changes in consumer payment behavior, the realized amount of bankruptcy losses, the competitive, legal, regulatory and reputational environment and the level of investments and growth in its businesses.

The Company expects to achieve these results based on the continued success of its business strategies and its current assessment of the competitive, regulatory and funding market environments that it faces (each of which is discussed elsewhere in this document), as well as the expectation that the geographies in which the Company competes will not experience significant consumer credit quality erosion, as might be the case in an economic downturn or recession.

#### Managed Revenue Margin

The Company expects its managed revenue margin (defined as managed net interest income plus managed non-interest income divided by average managed earning assets) to be modestly lower over time as a result of the Company's continuing diversification efforts and its bias towards higher credit quality assets. As a result of their product features, these assets may generate lower fee and interest revenues as a percentage of average loan balance than the Company's current portfolio. However, expenses as a percentage of average

assets are also expected to decline, thereby supporting overall returns. Auto Finance and Global Financial Services segments' assets are growing at a faster rate than the Company's managed loans. Such assets typically generate lower losses and higher average balances than those of the Company as a whole, thereby generating lower provision, operating and marketing expenses as a percentage of average managed loans. Efforts are also underway to improve operating expenses throughout the Company.

#### **Marketing Investment**

The Company expects its marketing investment, including brand advertising expenditures, to be approximately \$1.4 billion in 2005, exclusive of Hibernia, subject to market opportunities. The Company believes the branded franchise that it is building strengthens and enables its current and future direct marketing strategies across product lines. The Company cautions, however, that an increase or decrease in marketing expense does not necessarily correlate to a comparable increase or decrease in loan balances or accounts as a result of, among other factors, customer attrition and utilization patterns, variations in customer response rates, shifts over time in targeting consumers and/or products that have varying marketing acquisition costs, and the long-term nature of brand building.

The Company expects to deploy its marketing across its various products depending on the competitive dynamics of the various markets in which it participates. The Company expects to adjust its marketing allocations from time to time to target specific product lines that it believes offer attractive response rates and opportunities.

Due to the nature of competitive market dynamics and therefore the limited periods of opportunity identified by the Company's testing processes, marketing expenditures may fluctuate significantly from quarter to quarter. Marketing is often concentrated in the latter months of the year, aligned with the seasonality of card spending and borrowing. Additionally, a significant minority of the Company's marketing is related to brand advertising, which is supporting product roll outs and direct marketing, and building a long-term strategic asset for Capital One.

## **Operating Cost Trends**

The Company believes that a successful focus on managing operating costs is a critical component of its financial outlook. The Company measures operating efficiency using a variety of metrics which vary by specific department or business unit. Nevertheless, the Company believes that overall annual operating costs as a percentage of managed loans (defined as all non-interest expense less marketing, divided by average managed loans) is an appropriate gauge of the operating efficiency of the Company as a whole. As the Company continues its rigorous cost management program and to grow its Auto Finance and Global Financial Services segments' assets more quickly than the U.S. Card average, the Company expects operating costs as a percentage of its average managed loans to decline over time, excluding Hibernia, as a result of efficiency gains related to, among other things, servicing higher balance, higher credit quality assets.

## Managed Loan Growth

The Company expects managed loan growth rate to be approximately 12% in 2005, exclusive of Hibernia, with a higher growth rate in its diversification segments than in its U.S. Card segment.

## Managed Delinquencies and Net Charge-offs

The Company's managed net charge-off rate improved during 2004 and the first nine months of 2005 as a result of its continued asset diversification beyond U.S. Card and into higher credit quality categories, improved collections experience, and improving economic conditions. The Company does not expect to continue to realize improvement at the same rate. Due to the newly enacted bankruptcy legislation, the company estimates that there will be an incremental rise in the fourth quarter managed net charge-off rate of 50 to 100 basis points as a result of increased filings. Due to this one-time event, the company expects its quarterly managed charge-off rate to be approximately 5 percent for the fourth quarter of 2005, excluding Hibernia.

The Company's managed delinquency rate remained stable. Generally, fluctuations in delinquency levels have several effects, including changes in the amounts of past-due and overlimit fees assessed (lower delinquencies typically cause lower assessments), changes to the non-accrued amounts for finance charges and fees (lower delinquencies typically decrease non-accrued amounts), increased or decreased collections expenses, and/or changes in the reported allowance for loan losses and the associated provision expenses. The Company's allowance for loan losses in a given period is a function of reported charge-offs in the period, the delinquency status of reported loans and other factors, such as the Company's assessment of general economic conditions and the amount of outstanding loans added to the reported balance sheet during the period.

The Company expects a net allowance build in 2005, inclusive of the \$58 million release through the first three quarters of 2005, related to the seasonal increase in loans and delinquencies and charge-offs, and exclusive of Hibernia. The outlook is based on current and expected reported charge-off and delinquency rates, as well as expected reported loan growth, a higher growth rate in its Auto Finance and GFS businesses than its U.S. Card business, and a continuation of current economic conditions. This outlook is sensitive to general economic conditions, employment trends, and bankruptcy trends, in addition to growth of the Company's reported loans.

#### Return on Managed Assets

The Company expects that its return on managed assets will be between 1.7% and 1.8% for the full year of 2005, exclusive of Hibernia, with some quarterly variability, similar to 2004, as modest declines in revenue margin are more than offset by declines in provision and non-interest expenses as a percentage of managed loans.

The Company's objective is to continue diversifying its consumer finance activities, which may include expansion into additional geographic markets, other consumer loan products and/or additional branch banking businesses. In each business line, the Company expects to apply its proprietary marketing capabilities. The Company continues to seek to identify new product and new market opportunities, and to make investment decisions based on the Company's intensive testing and analysis.

The Company's lending products and other products are subject to intense competitive pressures that management anticipates will continue to increase as the lending markets mature, and it could affect the economics of decisions that the Company has made or will make in the future in ways that it did not anticipate, test or analyze.

#### **U.S.** Card Segment

The Company's U.S. Card segment consisted of \$46.3 billion of U.S. consumer credit card loans as of September 30, 2005, marketed to consumers across the full credit spectrum. The Company's strategy for its U.S. Card segment is to offer compelling, value-added products to its customers.

The competitive environment is currently intense for credit card products. Industry mail volume has increased substantially in recent years, resulting in declines in response rates to the Company's new customer solicitations over time. Additionally, the increase in other consumer loan products, such as home equity loans, puts pressure on growth throughout the credit card industry. These competitive pressures are continuing to increase as a result of, among other things, increasing consolidation within the industry. The industry's response to this competitive pressure has been to increase mail volumes to record levels, and in some parts of the market, most notably the prime revolver segment, offer extremely low up front pricing that appears to make profitability heavily dependent on penalty repricing well beyond "go to" rates for a substantial percentage of customers. The

Company is choosing to limit its marketing in those selected segments because it believes the prevailing pricing practices will compromise both economic returns and customer loyalty over the long term. Instead, the Company is focusing its efforts where it sees better opportunities to deliver profitable growth and create long term customer loyalty, such as in reward cards. Despite intense competitor pressure, the Company continues to believe that its marketing capabilities will enable it to originate new credit card accounts that exceed the Company's return on investment requirements and to generate a loan growth rate in the low single digits in 2005.

### **Auto Finance Segment**

The Company's Auto Finance segment consisted of \$15.7 billion of U.S. auto loans as of September 30, 2005, marketed across the full credit spectrum, via direct and dealer marketing channels.

The Company believes that its strong risk management skills, increasing operating scale, full credit spectrum product offerings and multi-channel marketing approach will enable it to continue to increase market share in the Auto Finance industry.

The Company expects that in 2005 the Auto Finance segment will continue to grow loans at a faster pace than the U.S. Card segment.

#### Global Financial Services Segment

The Global Financial Services segment consisted of \$22.8 billion of loans as of September 30, 2005, including international lending activities, small business lending, installment loans, home loans, healthcare financing and other consumer financial service activities.

The Company expects continued loan, credit and profit pressure from a deteriorating credit environment in its U.K. business. Despite this pressure, the Company continues to expect profitable long term growth from its U.K. business and is experiencing strong results from its North American business.

The Company expects the Global Financial Services segment will grow its loan portfolio at a faster pace than the U.S. Card segment.

## XI. Supervision and Regulation

The Corporation is a bank holding company ("BHC") under Section 3 of the Bank Holding Company Act of 1956, as amended (the "BHC Act") (12 U.S.C. § 1842). The Corporation is subject to the requirements of the BHC Act, including limiting its nonbanking activities to those that are permissible for a BHC. Such activities include those that are so closely related to banking as to be incident thereto such as consumer lending and other activities that have been approved by the Federal Reserve Board (the "Federal Reserve") by regulation or order. Certain servicing activities are also permissible for a BHC if conducted for or on behalf of the BHC or any of its affiliates. Impermissible activities for BHCs include activities that are related to commerce such as retail sales of nonfinancial products. Under Federal Reserve policy, the Corporation is expected to act as a source of financial and managerial strength to any banks that it controls, including the Bank and Savings Bank, and to commit resources to support them.

On May 27, 2005, the Corporation became a "financial holding company" under the Gramm-Leach-Bliley Act amendments to the BHC Act (the "GLBA"). The GLBA removed many of the restrictions on the activities of BHCs that become financial holding companies. A financial holding company, and the non-bank companies under its control, are permitted to engage in activities considered financial in nature (including, for example, insurance underwriting, agency sales and brokerage, securities underwriting, dealing and brokerage and merchant banking activities); incidental to financial activities; or complementary to financial activities if the Federal Reserve determines that they pose no risk to the safety or soundness of depository institutions or the financial system in general.

The Corporation's election to become a financial holding company under the GLBA certifies that the Bank and the Savings Bank meet certain criteria, including capital, management and Community Reinvestment Act requirements. If, after it becomes a financial holding company, the Corporation were to fail to continue to meet the criteria for financial holding company status, it could, depending on which requirements it failed to meet, face restrictions on new financial activities or acquisitions and/or be required to discontinue existing activities that are not generally permissible for bank holding companies.

The Bank is a banking corporation chartered under Virginia law and a member of the Federal Reserve System, the deposits of which are insured by the Bank Insurance Fund of the Federal Deposit Insurance Corporation (the "FDIC"). In addition to regulatory requirements imposed as a result of the Bank's international operations (discussed below), the Bank is subject to comprehensive regulation and periodic examination by the Bureau of Financial Institutions of the Virginia State Corporation Commission (the "Bureau of Financial Institutions"), the Federal Reserve, the FRB-R and the FDIC.

The Savings Bank is a federal savings bank chartered by the Office of Thrift Supervision (the "OTS") and is a member of the Federal Home Loan Bank System. Its deposits are insured by the Savings Association Insurance Fund of the FDIC. The Savings Bank is subject to comprehensive regulation and periodic examination by the OTS and the FDIC.

The Corporation is also registered as a financial institution holding company under Virginia law and as such is subject to periodic examination by Virginia's Bureau of Financial Institutions. The Corporation's automobile financing activities, conducted by COAF and its subsidiaries, fall under the scrutiny of the Federal Reserve and the state agencies having supervisory authority under applicable sales finance laws or consumer finance laws in most states. The Corporation also faces regulation in the international jurisdictions in which it conducts business.

For additional information on the Company's regulatory issues and activities, see the Corporation's Annual Report on Form 10-K for the year ended December 31, 2004, Part I, Item 1, "Supervision and Regulation."

### XII. Enterprise Risk Management

Risk is an inherent part of the Company's business and activities. The Company has an ongoing Enterprise Risk Management ("ERM") program designed to ensure appropriate and comprehensive oversight and management of risk. The ERM program operates at all levels in the Company: first, at the most senior levels with the Board of Directors and senior management committees that oversee risk and risk management practices; second, in the centralized departments headed by the Chief Enterprise Risk Officer and the Chief Credit Officer that establish risk management methodologies, processes and standards; and third, in the individual business areas throughout the Company which own the management of risk and perform ongoing identification, assessment and response to risks. The Company's Corporate Audit Services department also assesses risk and the related quality of internal controls and quality of risk management through its audit activities. To facilitate the effective management of risk, the Company utilizes a risk and control framework that includes eight categories of risk: credit, liquidity, market, operational, legal, strategic, reputation and compliance. For additional information on the Company's ERM program, see the Corporation's Annual Report on Form 10-K for the year ended December 31, 2004, Part I, Item 1, "Enterprise Risk Management".

#### XIII. Risk Factors

This Quarterly Report on Form 10-Q contains forward-looking statements. We also may make written or oral forward-looking statements in our periodic reports to the Securities and Exchange Commission on Forms 10-K and 8-K, in our annual report to shareholders, in our proxy statements, in our offering circulars and prospectuses, in press releases and other written materials and in statements made by our officers, directors or employees to third parties. Statements that are not about historical facts, including statements about our beliefs and expectations, are forward-looking statements. Forward-looking statements include information relating to our future earnings per share, growth in managed loans outstanding, product mix, segment growth, managed revenue margin, funding costs, operations costs, employment growth, marketing expense, delinquencies and charge-offs. Forward-looking statements also include statements using words such as "expect," "anticipate," "hope," "intend," "plan," "believe," "estimate" or similar expressions. We have based these forward-looking statements on our current plans, estimates and projections, and you should not unduly rely on them.

Forward-looking statements are not guarantees of future performance. They involve risks, uncertainties and assumptions, including the risks discussed below. Our future performance and actual results may differ materially from those expressed in these forward-looking statements. Many of the factors that will determine these results and values are beyond our ability to control or predict. We undertake no obligation to publicly update or revise any forward-looking statements, whether as a result of new information, future events or otherwise. You should carefully consider the factors discussed below in evaluating these forward-looking statements.

This section highlights specific risks that could affect our business and us. Although we have tried to discuss key factors, please be aware that other risks may prove to be important in the future. New risks may emerge at any time and we cannot predict such risks or estimate the extent to which they may affect our financial performance. In addition to the factors discussed elsewhere in this report, among the other factors that could cause actual results to differ materially are the following:

# We Face Intense Competition in All of Our Markets

We face intense competition from many other providers of credit cards and other consumer financial products and services. In particular, in our credit card activities, we compete with international, ragional and local bank card issuers, with other general purpose credit or charge card issuers, and to a certain extent, issuers of smart cards and debit cards. We also compete with providers of other types of financial services and consumer loans such as home equity lines and other mortgage related products that offer consumer debt consolidation. Thus, the cost to acquire new accounts will continue to vary among product lines and may rise. Other companies may compete with us for customers by offering lower initial interest rates and fees, higher credit limits and/or customer services or product features that are or may appear to be more attractive than those we offer. Because customers often choose credit card issuers (or other sources of financing) based on price (primarily interest rates and fees), credit limit and other product features, customer loyalty is limited. In addition, intense competition may lead to product and pricing practices that may adversely impact long-term customer loyalty; we may choose to not engage in such practices, which may adversely impact our ability to compete, particularly in the short term. Increased competition has resulted in, and may continue to cause, a decrease in credit card response rates and reduced productivity of marketing dollars invested in certain lines of business. Competition may also have an impact on customer attrition as our customers accept offers from other credit card lenders and/or providers of other consumer lending products, such as home equity financing.

Our other consumer lending businesses, including auto lending, small business lending, home loan lending, installment loans, and our businesses in international markets also compete on a similar variety of factors, including price, product features and customer service. These businesses may also experience a decline in marketing efficiency and/or an increase in customer attrition. In addition, some of our competitors may be substantially larger than we are, which may give those competitors advantages, including a more diversified product and customer base, operational efficiencies, broad-based local distribution capabilities, lower-cost

funding and more versatile technology platforms. These competitors may also consolidate with other financial institutions in ways that enhance these advantages and intensify our competitive environment. In addition, the Gramm-Leach-Bliley Financial Services Modernization Act of 1999 (the "GLB Act"), which permits greater affiliations between banks, securities firms and insurance companies, may increase competition in the financial services industry.

In such a competitive environment, we may lose entire accounts, or may lose account balances, to competing financial institutions, or find more costly to maintain our existing customer base. Customer attrition from any or all of our products, together with any lowering of interest rates or fees that we might implement to retain customers, could reduce our revenues and therefore our earnings. We expect that competition will continue to grow more intense with respect to most of our products, including our diversified products and the products we offer internationally.

# We May Experience Increased Delinquencies and Credit Losses

Like other credit card lenders and providers of consumer lending products, we face the risk that our customers will not repay their loans. A customer's failure to repay is generally preceded by missed payments. In some instances, a customer may declare bankruptcy prior to missing payments, although this is not generally the case. Customers who declare bankruptcy frequently do not repay credit card or other loans. When customers default on their loans and we have collateral, we attempt to seize it. However, the value of the collateral may not equal the amount of the unpaid loan and we may be unsuccessful in recovering the remaining balance from our customers. Rising delinquencies and rising rates of bankruptcy are often precursors of future charge-offs and may require us to increase our allowance for loan losses. Higher charge-off rates and an increase in our allowance for loan losses may hurt our overall financial performance if we are unable to raise revenue to compensate for these losses, may adversely impact the performance of our securitizations, and may increase our cost of funds.

Our ability to assess the credit worthiness of our customers may diminish. We market our products to a wide range of customers including those with less experience with credit products and those with a history of missed payments. We select our customers, manage their accounts and establish prices and credit limits using proprietary models and other techniques designed to accurately predict future charge-offs. Our goal is to set prices and credit limits such that we are appropriately compensated for the credit risk we accept for both high and low risk customers. We face a risk that the models and approaches we use to select, manage, and underwrite our customers may become less predictive of future charge-offs due to changes in the competitive environment or in the economy. Intense competition, a weak economy, or even falling interest rates can adversely affect our actual charge-offs and our ability to accurately predict future charge-offs. These factors may cause both a decline in the ability and willingness of our customers to repay their loans and an increase in the frequency with which our lower risk customers defect to more attractive, competitor products. In our auto finance business, declining used-car prices reduce the value of our collateral and can adversely affect charge-offs. We attempt to mitigate these risks by continually improving our approach to predicting future charge-offs and by evaluating potential adverse scenarios. Nonetheless, there can be no assurance that we will be able to accurately predict charge-offs, and our failure to do so may adversely affect our profitability and ability to grow.

Other factors may also impact our charge-off rates, and the trends that caused the reduction of charge-offs over the course of 2004 and the first part of 2005 may not continue. During that time, we increased the proportion of lower risk asset classes, like auto loans, relative to credit cards, continued our bias toward lower risk assets, and experienced a favorable economic environment. In 2004 and the first part of 2005, our managed loan portfolio continued to grow. The rate of growth in assets can impact charge-off rates as higher growth rates, especially in the credit card business, cause lower charge-off rates in the near term. This is primarily driven by lower charge-offs in the first six to eight months of the life of a pool of new accounts.

There are risks to the health of the U.S. economy. Specifically the effects of higher energy costs and hurricane damages may cause additional negative impacts on the economy. Additionally, we face a challenging credit environment in the United Kingdom, due in part to a downturn in the consumer credit cycle there and it is possible that the conditions may continue to worsen and continue to negatively affect charge-offs and delinquency rates.

Bankruptcy filing rates can also impact our charge-offs, and we anticipate an increase in our charge-off rate in the short-term due to increased bankruptcy filings made in anticipation of the effectiveness of new legislation in October 2005. Because there continues to be uncertainty regarding the on-going financial impact of this bankruptcy spike, our estimates regarding the impact may change.

We hold an allowance for expected losses inherent in our existing reported loan portfolio as provided for by the applicable accounting rules. There can be no assurance, however, that such allowances will be sufficient to account for actual losses. We record charge-offs according to accounting practices consistent with accounting and regulatory guidelines and rules. These guidelines and rules, including among other things, the FFIEC Account Management Guidance, could change and cause our charge-offs to increase for reasons unrelated to the underlying performance of our portfolio. Our charge-off and delinquency rates may also be impacted by industry developments, such as actions by our competitors to change minimum payment practices in response to advice from their regulators regarding the application of FFIEC Account Management Guidance, which may adversely impact industry charge-off and delinquency rates and, in turn, our rates. Unless offset by other changes, these impacts could reduce our profits.

#### We Face Risk From Economic Downturns

Delinquencies and credit losses in the consumer finance industry generally increase during economic downturns or recessions. Likewise, consumer demand may decline during an economic downturn or recession. Accordingly, an economic downturn (either local or national), can hurt our financial performance as accountholders default on their loans or, in the case of credit card accounts, carry lower balances and reduce credit card purchase activity. Furthermore, because our business model is to lend across the credit spectrum, we make loans to lower credit quality customers. These customers generally have higher rates of charge-offs and delinquencies than do higher credit quality customers. Additionally, as we continue to market our cards internationally, an economic downturn or recession outside the United States such as the one we are currently experiencing in the United Kingdom, could also hurt our financial performance.

#### We Face Strategic Risks in Sustaining Our Growth and Pursuing Diversification

Our growth strategy is threefold. First, we seek to continue to grow our domestic credit card business. Second, we desire to continue to build and grow our automobile finance business. Third, we hope to continue to diversify our business, both geographically and in product mix. We seek to do this by growing our lending businesses, including credit cards, internationally, principally in the United Kingdom and Canada, and by identifying, pursuing and expanding new business opportunities, such as branch banking and other consumer loan products. Our ability to continue to grow is driven by the success of our fundamental business plan, the level of our investments in new businesses or regions and our ability to apply our risk management skills to new businesses. In addition, our revenue may be adversely affected by our continuing diversification and bias toward lower loss assets (because of the potentially lower margins on such accounts). This risk has many components, including:

- Customer and Account Growth. Our growth is highly dependent on our ability to retain existing customers and attract new ones, grow existing and new
  account balances, develop new market segments and have sufficient funding available for marketing activities to generate these customers and account
  balances. Our ability to grow and retain customers is also dependent on customer satisfaction, which may be adversely affected by factors outside of
  our control, such as postal service and other marketing and customer service channel disruptions and costs.
- Product and Marketing Development. Difficulties or delays in the development, production, testing and marketing of new products or services, which
  may be caused by a number of factors including, among other things, operational constraints, technology functionality, regulatory and other capital
  requirements and legal difficulties, will affect the success of such products or services and can cause losses arising from the costs to develop
  unsuccessful products and services, as well as decreased capital availability. In addition, customers may not accept the new products and services
  offered.

• Diversification Risk. An important element of our strategy is our effort to continue diversifying beyond our U.S. Credit Card portfolio. Our ability to successfully diversify is impacted by a number of factors, including: identifying appropriate acquisition targets, executing on acquisition transactions, developing and executing strategies to grow our existing consumer financial services businesses, and our financial ability to undertake these diversification activities. In addition, part of our diversification strategy has been to grow internationally. Our growth internationally faces additional challenges, including limited access to information, differences in cultural attitudes toward borrowing, changing regulatory and legislative environments, political developments, possible economic downturns in other countries exchange rates and differences from the historical experience of portfolio performance in the United States and other countries.

#### Reputational Risk and Social Factors May Impact our Results

Our ability to originate and maintain accounts is highly dependent upon consumer and other external perceptions of our business practices and/or our financial health. Adverse perceptions regarding our business practices and/or our financial health could damage our reputation in both the customer and funding markets, leading to difficulties in generating and maintaining accounts as well as in financing them. Adverse developments with respect to the consumer or other external perceptions regarding the practices of our competitors, or our industry as a whole, may also adversely impact our reputation. In addition, adverse reputational impacts on third parties with whom we have important relationships, such as our independent auditors, may also adversely impact our reputation. Adverse impacts on our reputation, or the reputation of our industry, may also result in greater regulatory and/or legislative scrutiny, which may lead to laws or regulations that may change or constrain the manner in which we engage with our customers and the products we offer them. Adverse reputational impacts or events may also increase our litigation risk. See "We Face the Risk of a Complex and Changing Regulatory and Legal Environment" below. To this end, we carefully monitor internal and external developments for areas of potential reputational risk and have established governance structures to assist in evaluating such risks in our business practices and decisions.

In addition, a variety of social factors may cause changes in credit card and other consumer finance use, payment patterns and the rate of defaults by accountholders and borrowers. These social factors include changes in consumer confidence levels, the public's perception of the use of credit cards and other consumer debt, and changing attitudes about incurring debt and the stigma of personal bankruptcy.

# We Face Risk Related to the Strength of our Operational, Technological and Organizational Infrastructure

Our ability to grow and compete is dependent on our ability to build or acquire the necessary operational and technological infrastructure and to manage the cost of that infrastructure while we expand. Similar to other large corporations, in our case, operational risk can manifest itself in many ways, such as errors related to failed or inadequate processes, faulty or disabled computer systems, fraud by employees or persons outside of Capital One and exposure to external events. We are dependent on our operational infrastructure to help manage these risks. In addition, we are heavily dependent on the strength and capability of our technology systems which we use both to interface with our customers and to manage our internal financial and other systems. Our ability to develop and deliver new products that meet the needs of our existing customers and attract new ones depends on the functionality of our technology systems. Our ability to develop and implement effective marketing campaigns also depends on our technology.

We continuously monitor our operational and technological capabilities and make modifications and improvements when we believe it will be cost effective to do so. We are currently undertaking a project to

enhance the capabilities of our technological platform by working with Total System Services, Inc. ("Tsys") to outsource aspects of the system that supports most of our North American lending businesses. Our ability to successfully transition to this new platform as Tsys's ongoing ability to provide services to us, could impact our performance in the future. In some instances, we may build and maintain these capabilities ourselves. We also outsource some of these functions to third parties. These third parties may experience errors or disruptions that could adversely impact us and over which we may have limited control. As we increase the amount of our infrastructure that we outsource to third parties, we increase our exposure to this risk. We also face risk from the integration of new infrastructure platforms and/or new third party providers of such platforms into our existing businesses. Additionally, our ability to run our business in compliance with applicable laws and regulations is dependent on these infrastructures.

In addition to creating a solid infrastructure platform, we are also dependent on recruiting management and operations personnel with the experience to run an increasingly large and complex business. Although we take steps to retain our existing management talent and recruit new talent as needed, we face a competitive market for such talent and there can be no assurance that we will continue to be able to maintain and build a management team capable of running our increasingly large and complex business.

## We May Face Limited Availability of Financing, Variation in Our Funding Costs and Uncertainty in Our Securitization Financing

In general, the amount, type and cost of our funding, including financing from other financial institutions, the capital markets and deposits, directly impacts our expense in operating our business and growing our assets and therefore, can positively or negatively affect our financial results.

A number of factors could make such financing more difficult, more expensive or unavailable on any terms both domestically and internationally (where funding transactions may be on terms more or less favorable than in the United States), including, but not limited to, financial results and losses, changes within our organization, specific events that adversely impact our reputation, changes in the activities of our business partners, disruptions in the capital markets, specific events that adversely impact the financial services industry, counter-party availability, changes affecting our assets, our corporate and regulatory structure, interest rate fluctuations, ratings agencies actions, and the legal, regulatory, accounting and tax environments governing our funding transactions. In addition, our ability to raise funds is strongly affected by the general state of the U.S. and world economies, and may become increasingly difficult due to economic and other factors. Also, we compete for funding with other banks, savings banks and similar companies, some of which are publicly traded. Many of these institutions are substantially larger, may have more capital and other resources and may have better debt ratings than we do. In addition, as some of these competitors consolidate with other financial institutions, these advantages may increase. Competition from these institutions may increase our cost of funds.

Furthermore, we are dependent on the securitization of consumer loans, which involves the legal sale of beneficial interests in consumer loan balances and is a unique funding market. Despite the size and relative stability of these markets and our position as a leading issuer, if these markets experience difficulties we may be unable to securitize our loan receivables or to do so at favorable pricing levels. If we were unable to continue securitizing our loan receivables at current levels, we would have to use alternative funding sources to fund increases in loan receivables and meet our other liquidity needs. If we were unable to find cost-effective and stable alternatives, it could negatively impact our liquidity and potentially subject us to certain additional risks. These risks would include an increase in our cost of funds, an increase in the allowance for loan losses and the provision for possible credit losses as more loans would remain in our consolidated balance sheet, and lower loan growth.

In addition, the occurrence of certain events may cause the securitization transactions to amortize earlier than scheduled, which would accelerate the need for additional funding. This early amortization could, among other things, have a significant effect on the ability of the Bank and the Savings Bank to meet the capital adequacy requirements as all off-balance sheet loans experiencing such early amortization would have to be

recorded on the balance sheet. See pages 49-51 in Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity Risk Management" contained in the Corporation's Annual Report on Form 10-K for the year ended December 31, 2004.

# We May Experience Changes in Our Debt Ratings

In general, ratings agencies play an important role in determining, by means of the ratings they assign to issuers and their debt, the availability and cost of wholesale funding. We currently receive ratings from several ratings entities for our secured and unsecured borrowings. As private entities, ratings agencies have broad discretion in the assignment of ratings. A rating below investment grade typically reduces availability and increases the cost of market-based funding, both secured and unsecured. A debt rating of Baa3 or higher by Moody's Investors Service, or BBB- or higher by Standard & Poor's and Fitch Ratings, is considered investment grade. Currently, all three ratings agencies rate the unsecured senior debt of the Bank and the Corporation as investment grade. The following chart shows ratings for Capital One Financial Corporation and Capital One Bank as of September 30, 2005. As of that date, the ratings outlooks were as follows:

	&		
	Poor's	Moody's	Fitch
Capital One Financial Corporation	BBB-	Baa3	BBB
Capital One Financial Corporation—Outlook	Positive	Positive	Positive
Capital One Bank	BBB	Baa2	BBB
Capital One Bank—Outlook	Positive	Positive	Positive

Standard

Because we depend on the capital markets for funding and capital, we could experience reduced availability and increased cost of funding if our debt ratings were lowered. This result could make it difficult for us to grow at or to a level we currently anticipate. The immediate impact of a ratings downgrade on other sources of funding, however, would be limited, as our deposit funding and pricing, as well as some of our unsecured corporate borrowing, is not generally determined by corporate debt ratings.

# We Face Exposure from Our Unused Customer Credit Lines

Because we offer our customers credit lines, the full amount of which is most often not used, we have exposure to these unfunded lines of credit. These credit lines could be used to a greater extent than our historical experience would predict. If actual use of these lines were to materially exceed predicted line usage, we would need to raise more funding than anticipated in our current funding plans. It could be difficult to raise such funds, either at all, or at favorable rates.

# We Face Market Risk of Interest Rate and Exchange Rate Fluctuations

Like other financial institutions, we borrow money from institutions and depositors, which we then lend to customers. We earn interest on the consumer loans we make, and pay interest on the deposits and borrowings we use to fund those loans. Changes in these two interest rates affect the value of our assets and liabilities. If the rate of interest we pay on our borrowings increases more than the rate of interest we earn on our loans, our net interest income, and therefore our earnings, would fall. Our earnings could also be hurt if the rates on our consumer loans fall more quickly than those on our borrowings.

However, our goal is to maintain an interest rate position that limits the impact of instantaneous movements in interest rates of up to 300 basis points to plus or minus 3% of net interest income over a twelve month period. We also seek to minimize foreign exchange fluctuations' impact to a level that is immaterial to our net income. The financial instruments and techniques we use to manage the risk of interest rate and exchange rate fluctuations, such as asset/liability matching and interest rate and exchange rate swaps and hedges and some forward exchange contracts, may not always work successfully or may not be available at a reasonable cost. Furthermore, if these techniques become unavailable or impractical, our earnings could be subject to volatility and decreases as interest rates and exchange rates change.

Changes in interest rates also affect the balances our customers carry on their credit cards and affect the rate of pre-payment for installment loan products. When interest rates fall, there may be more low-rate product alternatives available to our customers. Consequently, their credit card balances may fall and pre-payment rates for installment loan products may rise. We can mitigate this risk by reducing the interest rates we charge or by refinancing installment loan products. However, these changes can reduce the overall yield on our portfolio if we do not adequately provide for them in our interest rate hedging strategies. When interest rates rise, there are fewer low-rate alternatives available to customers. Consequently, credit card balances may rise (or fall more slowly) and pre-payment rates on installment lending products may fall. In this circumstance, we may have to raise additional funds at higher interest rates. In our credit card business, we

could, subject to legal and competitive constraints, mitigate this risk by increasing the interest rates we charge, although such changes may increase opportunities for our competitors to offer attractive products to our customers and consequently increase customer attrition from our portfolio. We could also mitigate this risk through hedging strategies, if available, if we are unable to do so, we could suffer adverse impacts on overall portfolio yield. Rising interest rates across the industry may also lead to higher delinquencies as customers face increasing interest payments both on our products and on other loans they may hold. See pages 51-52 in Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations—Interest Rate Risk Management" contained in the Annual Report on Form 10-K for the year ended December 31, 2004.

#### We Face the Risk of a Complex and Changing Regulatory and Legal Environment

We operate in a heavily regulated industry and are therefore subject to a wide array of banking, consumer lending and deposit laws and regulations that apply to almost every element of our business. Failure to comply with these laws and regulations could result in financial, structural and operational penalties, including receivership. In addition, efforts to comply with these laws and regulations may increase our costs and/or limit our ability to pursue certain business opportunities. See "Supervision and Regulation" above. Federal and state laws and regulations, as well as laws and regulations to which we are subject in foreign jurisdictions in which we conduct business, significantly limit the types of activities in which we may engage. For example, federal and state consumer protection laws and regulations, and laws and regulations of foreign jurisdictions where we conduct business, limit the manner in which we may offer and extend credit. In addition, we are subject to a wide array of other laws and regulations that govern other aspects of how we conduct our business, such as in the areas of employment and intellectual property. From time to time, the U.S. Congress, the states and foreign governments consider changing these laws and may enact new laws or amend existing laws to regulate further the consumer lending industry or companies in general. Such new laws or regulations could limit the amount of interest or fees we can charge, restrict our ability to collect on account balances, or materially affect us or the banking or credit card industries in some other manner. Additional federal, state and foreign consumer protection legislation also could seek to expand the privacy protections afforded to customers of financial institutions and restrict our ability to share or receive customer information.

In addition, banking regulators possess broad discretion to issue or revise regulations, or to issue guidance, which may significantly impact us. For example, the Federal Trade Commission has issued, and will continue to issue, a variety of regulations under the FACT Act of 2003, the Federal Reserve has announced proposed rule-making, and has issued some final rules, and in the UK the Office of Fair Trading is conducting an industry investigation on the calculation of default charges, all of which may impact us. We cannot, however, predict whether and how any new guidelines issued or other regulatory actions taken by the banking or other regulators will be applied to the Bank or the Savings Bank, in what manner such regulations might be applied, or the resulting effect on us, the Bank or the Savings Bank. There can be no assurance that this kind of regulatory action will not have a negative impact on us and/or our financial results.

Finally, we face possible risks from the outcomes of certain industry litigation. Over the past several years, MasterCard International and Visa U.S.A., Inc., as well as several of their member banks, have been involved in several different lawsuits challenging various practices of MasterCard and Visa.

In 1998, the United States Department of Justice filed an antitrust lawsuit against the MasterCard and Visa membership associations composed of financial institutions that issue MasterCard or Visa credit or debit cards ("associations"), alleging, among other things, that the associations had violated antitrust law and engaged in unfair practices by not allowing member banks to issue cards from competing brands, such as American Express and Discover Financial Services. In 2001, a New York district court entered judgment in favor of the Department of Justice and ordered the associations, among other things, to repeal these policies. The United States Court of Appeals for the Second Circuit affirmed the district court and, on October 4, 2004, the United States Supreme Court denied certiorari in the case.

After the Supreme Court denied certiorari, American Express Travel Related Services Company, Inc., on November 15, 2004, filed a lawsuit against the associations and several member banks under United States

federal antitrust law. The lawsuit alleges, among other things, that the associations and member banks implemented and enforced illegal exclusionary agreements that prevented member banks from issuing American Express and Discover cards. The complaints, among other things, request civil monetary damages, which could be trebled. We, Capital One Bank, Capital One, F.S.B. and Capital One Financial Corporation are named defendants in this lawsuit.

Between June 22, 2005 to the present date, a number of entities, each purporting to represent a class of retail merchants, filed five separate lawsuits against the associations and several member banks under United States federal antitrust law. The lawsuits allege, among other things, that the associations and member banks conspired to fix the level of interchange fees. The complaints, among other things, request civil monetary damages, which could be trebled. We, Capital One Bank, Capital One, F.S.B., and Capital One Financial Corporation, among others, are named defendants.

We believe that we have meritorious defenses with respect to these cases and intend to defend these cases vigorously.

In addition, several United States merchants have filed class action lawsuits, which have been consolidated, against the associations under United States federal antitrust law relating to certain debit card products. In April 2003, the associations agreed to settle the lawsuit in exchange for payments to plaintiffs by MasterCard of \$1 billion and Visa of \$2 billion, both over a ten-year period, and for changes in policies and interchange rates for debit cards. Certain merchant plaintiffs have opted out of the settlements and have commenced separate lawsuits. Additionally, consumer class action lawsuits with claims mirroring the merchants' allegations have been filed in several courts. Finally, the associations, as well as certain member banks, continue to face additional lawsuits regarding policies, practices, products and fees.

With the exception of the antitrust lawsuits brought by certain merchants between June 22, 2005 to the present date, and the American Express antitrust lawsuit, the banks and their affiliates are not parties to the lawsuits against the associations described above and therefore will not be directly liable for any amount related to any possible or known settlements, the lawsuits filed by merchants who have opted out of the settlements of those lawsuits or the class action lawsuits pending in state and federal courts. However, the banks are member banks of MasterCard and Visa and thus may be affected by settlements or lawsuits relating to these issues. In addition, it is possible that the scope of these lawsuits may expand and that other member banks, including the banks, may be brought into the lawsuits or future lawsuits.

Given the complexity of the issues raised by the lawsuits described above and the uncertainty regarding: (i) the outcome of these suits, (ii) the likelihood and amount of any possible judgment against the associations or the member banks, (iii) the likelihood, amount and validity of any claim against the associations' member banks, including the banks and the Corporation, and (iv) the effects of these suits, in turn, on competition in the industry, member banks, and interchange and association fees, we cannot determine at this time the long-term effects of these suits on us.

# We Face the Risk of Fluctuations in Our Expenses and Other Costs that May Hurt Our Financial Results

Our expenses and other costs, such as operating and marketing expenses, directly affect our earnings results. In light of the extremely competitive environment in which we operate, and because the size and scale of many of our competitors provides them with increased operational efficiencies, it is important that we are able to successfully manage such expenses. Many factors can influence the amount of our expenses, as well as how quickly they may increase. For example, further increases in postal rates or termination of our negotiated service arrangement with the United States Postal Service could raise our costs for postal service. As our business develops, changes or expands, additional expenses can arise from management of outsourced services, asset purchases, structural reorganization, a reevaluation of business strategies and/or expenses to comply with new or changing laws or regulations. Other factors that can affect the amount of our expenses include legal and administrative cases and proceedings, which can be expensive to pursue or defend. In addition, changes in accounting fluctuations can significantly affect how we calculate expenses and earnings.

#### We Face Risks Related to our Proposed Merger with Hibernia Corporation

Completion of the proposed merger is subject to the satisfaction of various conditions, including the receipt of approval from the Hibernia stockholders. There is no assurance that the merger will be completed on the proposed terms and schedule. Additionally, when and if the merger is completed, we face the risks that the businesses may not be integrated successfully and that the cost savings and other synergies from the transaction may not be fully realized, or may take longer to realize than expected. Finally, uncertainties or disruptions related to the transaction may make it more difficult to maintain relationships with customers, employees or suppliers.

# Upon Consummation of the Merger with Hibernia, We Face Risks Related to the Impact of Hurricanes Katrina and Rita, That May Be Substantial and Cannot Be Predicted

Hibernia is headquartered in New Orleans, Louisiana, and maintains branches in the areas of Louisiana and Texas that sustained significant damage from hurricanes Katrina and Rita. Hibernia's operations in other parts of Louisiana and Texas have not been impacted, either significantly or at all, by the hurricanes.

As a result of the hurricanes, Hibernia is experiencing increased costs, including the costs of rebuilding or repairing branches and other properties as well as repairing or replacing equipment, some of which are not covered by insurance. Hibernia has also announced substantial employee and recovery related costs.

Hurricanes Katrina and Rita have also affected Hibernia's consumer, mortgage, auto, commercial and small business loan portfolios by damaging properties pledged as collateral and by impairing certain borrowers' ability to repay their loans. In addition, Hibernia may experience losses from certain customer assistance policies, such as fee waivers, adopted in the wake of the hurricanes. The hurricanes may continue to affect Hibernia's loan originations and loan portfolio quality into the future and could also adversely impact Hibernia's deposit base. More generally, the combined company's ability to compete effectively in the branch banking business in the future, especially with financial institutions whose operations were not concentrated in the affected area or which may have greater resources than the combined company, will depend primarily on Hibernia's ability to resume normal business operations. The severity and duration of these effects will depend on a variety of factors that are beyond Hibernia's control, including the amount and timing of government, private and philanthropic investment (including deposits) in the region, the pace of rebuilding and economic recovery in the region generally, the extent to which the hurricanes' property damage is covered by insurance, and the pace at which Hibernia restores its business operations in the various markets in which it operates.

None of the effects described above can be accurately predicted or quantified. As a result, significant uncertainty remains regarding the impact the hurricanes will have on the business, financial condition and results of operations of the combined company and the ability of the combined company to realize the anticipated benefits from the merger. Further, the area in which Hibernia operates may experience hurricanes and other storms in the future, and some of those hurricanes and storms may have effects similar to those caused by Hurricanes Katrina and Rita.

# **XIV. Tabular Summary**

# TABLE A – STATEMENTS OF AVERAGE BALANCES, INCOME AND EXPENSE, YIELDS AND RATES

Table A provides average balance sheet data and an analysis of net interest income, net interest spread (the difference between the yield on earning assets and the cost of interest-bearing liabilities) and net interest margin for the three and nine months ended September 30, 2005 and 2004.

Three Months Ended September 30

	Three Months Ended September 30					
		2005			2004	
(Dollars in thousands)	Average Balance	Income/ Expense	Yield/ Rate	Average Balance	Income/ Expense	Yield/ Rate
Assets:						
Earning assets						
Consumer loans	****	*				
Domestic	\$34,532,826	\$1,121,670	12.99%	\$31,144,773	\$ 983,697	12.63%
International	4,022,749	106,490	10.59%	3,627,716	99,589	10.98%
Total	38,555,575	1,228,160	12.74%	34,772,489	1,083,286	12.46%
Securities available for sale	9,535,858	87,978	3.69%	9,372,713	84,492	3.61%
Other						
Domestic	3,786,333	66,668	7.04%	2,138,874	46,533	8.70%
International	1,575,157	21,809	5.54%	983,334	14,102	5.74%
Total	5,361,490	88,477	6.60%	3,122,208	60,635	7.77%
Total earning assets	53,452,923	\$1,404,615	10.51%	47,267,410	\$1,228,413	10.40%
Cash and due from banks	1,009,573			450,726		
Allowance for loan losses	(1,405,106)			(1,424,672)		
Premises and equipment, net	771,857			859,328		
Other	5,374,285			4,342,788		
Total assets	\$59,203,532			\$51,495,580		
Liabilities and Equity:						
Interest-bearing liabilities						
Deposits						
Domestic	\$24,082,814	\$ 252,515	4.19%	\$22,917,496	\$ 234,162	4.09%
International	2,535,658	33,096	5.22%	1,796,428	23,187	5.16%
Total	26,618,472	285,611	4.29%	24,713,924	257,349	4.17%
Senior and subordinated notes	6,683,533	98,309	5.88%	7,218,916	121,166	6.71%
Other borrowings						
Domestic	10,683,104	110,436	4.13%	8,673,529	74,518	3.44%
International	15,112	40	1.06%	769	5	2.60%
Total	10,698,216	110,476	4.13%	8,674,298	74,523	3.44%
Total interest-bearing liabilities	44,000,221	\$ 494,396	4.49%	40,607,138	\$ 453,038	4.46%
Other	4,401,616	10 1,000		3,327,323	4 100,000	
Total liabilities	48,401,837			43,934,461		
Equity	10,801,695			7,561,119		
Total liabilities and equity	\$59,203,532			\$51,495,580		
Net interest spread			6.02%			5.94%
ivet interest spread						
			10 E10/			10.400
Interest income to average earning assets Interest expense to average earning assets			10.51% 3.70%			10.40%

Interest income includes past-due fees of approximately \$192,170 and \$197,109 for the three months ended September 30, 2005 and 2004, respectively.

		Nine	Months End	ed September 30		
		2005			2004	_
(Dollars in thousands)	Average Balance	Income/ Expense	Yield/ Rate	Average Balance	Income/ Expense	Yield/ Rate
Assets:						
Earning assets						
Consumer loans						
Domestic	\$34,103,733	\$3,256,624	12.73%	\$30,088,461	\$2,852,223	12.64%
International	4,228,835	345,670	10.90%	3,562,481	285,156	10.67%
Total	38,332,568	3,602,294	12.53%	33,650,942	3,137,379	12.43%
Securities available for sale	9,593,879	269,387	3.74%	8,590,499	224,289	3.48%
Other						
Domestic	2,715,891	163,043	8.00%	2,549,683	145,250	7.60%
International	1,374,753	58,059	5.63%	909,398	38,172	5.60%
Total	4,090,644	221,102	7.21%	3,459,081	183,422	7.07%
Total earning assets	52,017,091	\$4,092,783	10.49%	45,700,522	\$3,545,090	10.34%
Cash and due from banks	979,255			533,626		
Allowance for loan losses	(1,451,143)			(1,503,979)		
Premises and equipment, net	802,602			894,371		
Other	5,007,512			4,120,109		
Total assets	\$57,355,317			\$49,744,649		
Liabilities and Equity: Interest-bearing liabilities						
Deposits						
Domestic	\$23,677,496	\$ 730,677	4.11%	\$22,171,478	\$ 676,614	4.07%
International	2,547,516	98,397	5.15%	1,716,478	65,225	5.07%
Total Deposits	26,225,012	829,074	4.22%	23,887,956	741,839	4.14%
Senior and subordinated notes	6,859,069	317,382	6.17%	7,289,821	370,393	6.77%
Other borrowings						
Domestic	10,527,543	302,758	3.83%	8,332,505	214,420	3.43%
International	14,386	326	3.02%	867	24	3.69%
Total	10,541,929	303,084	3.83%	8,333,372	214,444	3.43%
Total interest-bearing liabilities	43,626,010	\$1,449,540	4.43%	39,511,149	\$1,326,676	4.48%
Other	3,860,057			3,248,837		
Total liabilities	47,486,067			42,759,986		
Equity	9,869,250			6,984,663		
Total liabilities and equity	\$57,355,317			\$49,744,649		
Net interest spread			6.06%			5.86%
Interest income to average earning assets			10.49%			10.34%
Interest expense to average earning assets			3.71%			3.87%

Net interest margin

6.78%

6.47%

<sup>(1)</sup> Interest income includes past-due fees of approximately \$595,729 and \$593,455 for the nine months ended September 30, 2005 and 2004, respectively.

### TABLE B - INTEREST VARIANCE ANALYSIS

Three Months Ended September 30, 2005 vs. 2004 Nine Months Ended September 30, 2005 vs. 2004 Change due to<sup>(1)</sup> Change due to<sup>(1)</sup> Increase Increase (Dollars in thousands) (Decrease) Volume Yield/Rate (Decrease) Volume Yield/Rate **Interest Income:** Consumer loans Domestic \$137,973 \$109,420 \$ 28,553 \$404,401 \$383,279 \$ 21,122 54,353 International 6,901 26,621 (19,720)60,514 6,161 Total loans 144,874 120,053 24,821 464,915 439,767 25,148 Securities available for sale 3,486 1,485 2,001 45,098 27,398 17,700 Other Domestic 20,135 73,036 (52,901)17,793 9,748 8,045 International 7,707 10,974 (3,267)19,887 235 19,652 Total 27,842 83,284 (55,442)37,680 34,074 3,606 Total interest income 176,202 162,400 13,802 547,693 496,340 51,353 **Interest Expense: Deposits** Domestic 18,353 12,113 6,240 54,063 46,402 7,661 International 9,909 263 33,172 32,081 1,091 9,646 7,997 Total 28,262 20,265 87,235 73,680 13,555 (22,857)Senior notes (8,570)(14,287)(53,011)(21,108)(31,903)Other borrowings Domestic 35,918 19,134 16,784 88,338 61,077 27,261 International 35 57 302 310 (22)(8) Total 35,953 19,270 16,683 88,640 61,453 27,187 Total interest expense 41,358 38,104 3,254 122,864 145,251 (22,387)

\$134,844

\$104,458

\$ 30,386

\$424,829

\$317,343

\$107,486

Net interest income(1)

<sup>(1)</sup> The change in interest due to both volume and rates has been allocated in proportion to the relationship of the absolute dollar amounts of the change in each. The changes in income and expense are calculated independently for each line in the table. The totals for the volume and yield/rate columns are not the sum of the individual lines.

# TABLE C - MANAGED CONSUMER LOAN PORTFOLIO

Table C summarizes the Company's managed consumer loan portfolio.

		nths Ended nber 30
(Dollars in thousands)	2005	2004
Period-End Balances:		
Reported consumer loans:		
Domestic	\$ 35,297,204	\$ 31,510,313
International	3,554,559	3,650,323
Total	38,851,763	35,160,635
Securitization adjustments:		
Domestic	39,049,176	34,773,552
International	6,866,744	5,522,644
Total	45,915,920	40,296,196
Managed consumer loan portfolio:		
Domestic	74,346,380	66,283,86
International	10,421,303	9,172,96
Total	\$ 84,767,683	\$ 75,456,832
Average Balances:		
Reported consumer loans:		
Domestic	\$ 34,532,826	\$ 31,144,773
International	4,022,749	3,627,71
Total	38,555,575	34,772,489
Securitization adjustments:		
Domestic	38,690,163	34,303,03
International	6,581,727	5,322,780
Total	45,271,890	39,625,812
Managed consumer loan portfolio:		
Domestic	73,222,989	65,447,805
International	10,604,476	8,950,490
Total	\$ 83,827,465	\$ 74,398,302
		nths Ended nber 30,
	2005	2004
Average Balances:		
Reported consumer loans:		
Domestic	\$ 34,103,733	\$ 30,088,462
International	4,228,835	3,562,482
Total	38,332,568	33,650,942
Securitization adjustments:		
Domestic	37,996,151	34,030,01
International	6,325,957	4,950,11
Total	44,322,108	38,980,13
Managed consumer loan portfolio:		
Domestic	72,099,884	64,118,47
International	10,554,792	8,512,600
Total	\$ 82,654,676	\$ 72,631,070

# TABLE D - DELINQUENCIES

Table D shows the Company's consumer loan delinquency trends for the periods presented on a reported and managed basis.

(Dollars in thousands)  Reported:  Loans outstanding	\$38,851,763 783,200	% of Total Loans 100.00%	Loans \$35,160,635	% of Total Loans
Reported: Loans outstanding	\$38,851,763			
<u> </u>		100.00%	\$35,160,635	100 00%
Tarana dalla assanta	783,200			100.0070
Loans delinquent:	783,200			
30-59 days		2.02%	701,974	2.00%
60-89 days	341,338	0.88%	304,855	0.87%
90-119 days	191,615	0.49%	182,997	0.52%
120-149 days	102,549	0.26%	131,148	0.37%
150 or more days	78,011	0.20%	86,323	0.24%
Total	\$ 1,496,713	3.85%	\$ 1,407,297	4.00%
Loans delinquent by geographic area:				
Domestic	1,415,513	4.01%	1,327,508	4.21%
International	81,200	2.28%	79,789	2.19%
Managed:				
Loans outstanding	\$84,767,683	100.00%	\$75,456,831	100.00%
Loans delinquent:				
30-59 days	1,396,010	1.65%	1,262,574	1.67%
60-89 days	719,101	0.85%	651,193	0.86%
90-119 days	478,624	0.56%	448,052	0.59%
120-149 days	322,183	0.38%	335,336	0.44%
150 or more days	247,859	0.29%	247,308	0.34%
Total	\$ 3,163,777	3.73%	\$ 2,944,463	3.90%

# TABLE E - NET CHARGE-OFFS

Table E shows the Company's net charge-offs for the periods presented on a reported and managed basis.

Three Months Ended		Nine Months Ended		
	September 30			oer 30
(Dollars in thousands)	2005	2004	2005	2004
Reported:				
Average loans outstanding	\$38,555,575	\$34,772,489	\$38,332,568	\$33,650,942
Net charge-offs	341,821	298,317	996,139	950,495
Net charge-offs as a percentage of average loans outstanding	3.55%	3.43%	3.46%	3.77%
Managed:				
Average loans outstanding	\$83,827,465	\$74,398,301	\$82,654,676	\$72,631,076
Net charge-offs	867,988	753,698	2,556,529	2,412,022
Net charge-offs as a percentage of average loans outstanding	4.14%	4.05%	4.12%	4.43%

# TABLE F - SUMMARY OF ALLOWANCE FOR LOAN LOSSES

Table F sets forth the activity in the allowance for loan losses for the periods indicated.

		nths Ended nber 30	d Nine Months Er September 3	
(Dollars in thousands)	2005	2004	2005	2004
Balance at beginning of period	\$1,405,000	\$1,425,000	\$ 1,505,000	\$ 1,595,000
Provision for loan losses:				
Domestic	346,788	250,181	796,053	683,001
International	27,379	17,614	129,345	70,718
Total provision for loan losses	374,167	267,795	925,398	753,719
Other	9,654	522	12,741	(3,224)
Charge-offs:				
Domestic	(401,559)	(376,949)	(1,193,370)	(1,192,590)
International	(50,183)	(34,906)	(141,851)	(102,059)
Total charge-offs	(451,742)	(411,855)	(1,335,221)	(1,294,649)
Recoveries:				
Consumer loans				
Domestic	99,338	102,059	306,932	310,608
International	10,583	11,479	32,150	33,546
Total recoveries	109,921	113,538	339,082	344,154
Net charge-offs	(341,821)	(298,317)	(996,139)	(950,495)
Balance at end of period	\$1,447,000	\$1,395,000	\$ 1,447,000	\$ 1,395,000
Allowance for loan losses to loans at period-end	3.72%	3.97%	3.72%	3.97%
Allowance for loan losses by loan category:				
Consumer loans				
Domestic	\$1,286,850	\$1,275,879	\$ 1,286,850	\$ 1,275,879
International	160,150	119,121	160,150	119,121

#### Item 3. Quantitative and Qualitative Disclosure of Market Risk

The information called for by this item is provided under the Corporation's Annual Report on Form 10-K for the year ended December 31, 2004, Item 7A "Quantitative and Qualitative Disclosures about Market Risk". No material changes have occurred during the three month period ended September 30, 2005.

# Item 4. Controls and Procedures

The Corporation's management carried out an evaluation of the effectiveness of the design and operation of the Corporation's disclosure controls and internal controls and procedures as of September 30, 2005 pursuant to Exchange Act Rules 13a-14 and 13a-15. These controls and procedures for financial reporting are the responsibility of the Corporation's management. Based upon that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that the Company's disclosure controls and procedures are effective in alerting them in a timely manner to material information relating to the Corporation (including consolidated subsidiaries) required to be included in the Corporation's periodic filings with the Securities and Exchange Commission. The Corporation has a Disclosure Committee consisting of members of senior management to assist in this evaluation.

#### **Part II Other Information**

## Item 1. Legal Proceedings

The information required by Item 1 is included in this Quarterly Report under the heading "Notes to Condensed Consolidated Financial Statements – Note 7 – Commitments and Contingencies."

#### Item 2. Changes in Securities, Uses of Proceeds and Issuer Purchases of Equity Securities.

Period	(a) Total Number of Shares Purchased <sup>(1)</sup>	(b) Average Price Paid per Share	(c) Total Number of Shares Purchased as Part of Publicly Announced Plans	(d) Maximum Number of Shares that May Yet Be Purchased Under the Plans
July 1-31, 2005	48,993	\$ 82.96	N/A	N/A
August 1-31, 2005	34,372	83.82	N/A	N/A
September 1-30, 2005	6,257	81.57	N/A	N/A
Total	89,622	83.19	N/A	N/A

<sup>(1)</sup> Shares purchased represent share swaps made in connection with stock option exercises and the withholding of shares to cover taxes on restricted stock lapses.

## Item 6. Exhibits and Reports on Form 8-K

## (a) Exhibits:

- 10.1 Services Agreement, dated August 5, 2005, between Capital One Financial Corporation, acting through its subsidiary Capital One Services, Inc. and Total System Services, Inc. (confidential treatment has been requested for portions of this document).
- 31.1 Certification of Richard D. Fairbank
- 31.2 Certification of Gary L. Perlin
- 32.1 Certification\* of Richard D. Fairbank
- 32.2 Certification\* of Gary L. Perlin

#### (b) Reports on Form 8-K:

On July 20, 2005, the Company filed under Item 2.02 — "Results of Operations and Financial Condition", Item 7.01 — "Regulation FD Disclosure", Item 8.01 – "Other Events", and Item 9.01— "Financial Statements, Pro Forma Financial Information and Exhibits" of Form 8-K, on Exhibit 99.1, a copy of its earnings press release for the second quarter 2005 that was issued July 20, 2005. This release, which is required under Item 2.02, "Results of Operations and Financial Condition," has been included under Item 7.01 pursuant to interim reporting guidance provided by the SEC. Additionally, the Company furnished the information in Exhibit 99.2, Second Quarter Earnings Presentation for the quarter ended June 30, 2005.

On July 20, 2005, the Company furnished under Item 7.01—"Regulation FD Disclosure" and Item 9.01 – "Financial Statements, Pro Forma Financial Information and Exhibits" of Form 8-K on Exhibit 99.1 the Monthly Charge-off and Delinquency Statistics—June 2005 for the month ended June 30, 2005.

On August 9, 2005, the Company filed under Item 1.01—"Entry into a Material Definitive Agreement" of Form 8-K that the Company acting through its subsidiary, COSI, entered into a processing services agreement with Total System Services, Inc., effective August 5, 2005.

On August 10, 2005, the Company furnished under Item 7.01—"Regulation FD Disclosure" and Item 9.01 – "Financial Statements and Exhibits" of Form 8-K, on Exhibit 99.1 the Monthly Charge-off and Delinquency Statistics—July 2005 for the month ended July 31, 2005.

On September 1, 2005, the Company filed under Item 7.01—"Regulation FD Disclosure" and Item 9.01 – "Financial Statements and Exhibits" of Form 8-K on Exhibit 99.1 a copy of its press release regarding the delay in closing of the merger of Hibernia Corporation with and into the Company, dated August 31, 2005.

One September 7, 2005, the Company filed under Item 5.02—"Departure of Directors or Principal Officers; Election of Directors; Appointment of Principal Officers" and Item 9.01 – "Financial Statements and Exhibits" of Form 8-K on Exhibit 99.1 a copy of its press release that announced the appointment of Pierre E. Leroy to its Board of Directors, dated September 6, 2005.

On September 7, 2005, the Company filed under Item 7.01—"Regulation FD Disclosure" and Item 9.01 – "Financial Statements and Exhibits" of Form 8-K on Exhibit 99.1 a copy of its press release dated September 7, 2005, regarding an amendment, dated as of September 6, 2005, to the Agreement and Plan of Merger, dated as of March 6, 2005, between the Company and Hibernia Corporation.

On September 8, 2005, the Company filed under Item 1.01—"Entry into a Material Definitive Agreement" and Item 9.01 – "Financial Statements and Exhibits" of Form 8-K on Exhibit 2.1, a copy of Amendment No. 1, dated as of September 6, 2005, to the Agreement and Plan of Merger, dated as of March 6, 2005, between the Company and Hibernia Corporation.

On September 12, 2005, the Company furnished under Item 7.01—"Regulation FD Disclosure" and Item 9.01 – "Financial Statements and Exhibits" of Form 8-K on Exhibit 99.1 the Monthly Charge-off and Delinquency Statistics—August 2005 for the month ended August 31, 2005.

On September 15, 2005, the Company filed under Item 8.01—"Other events" and Item 9.01 – "Financial Statements and Exhibits" of Form 8-K on Exhibit 10.1 a copy of the revised redacted Services Agreement with First Data Resources, Inc. dated November 8, 2004, which was previously filed as Exhibit 10.8 to the Company's From 10-K for the year ended December 31, 2004.

\* Information in this furnished herewith shall not be deemed to be "filed" for the purposes of Section 18 of the 1934 Act or otherwise subject to the liabilities of that section.

# **SIGNATURES**

Date: November 1, 2005

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

# CAPITAL ONE FINANCIAL CORPORATION

(Registrant)

/s/ GARY L.PERLIN

Gary L. Perlin Executive Vice President and Chief Financial Officer (Principal Financial Officer and duly authorized officer of the Registrant) [\*\*\*] indicates the omission of confidential portions for which confidential treatment has been requested. Such confidential information has been filed separately with the Securities and Exchange Commission.

# PROCESSING SERVICES AGREEMENT

BETWEEN

CAPITAL ONE SERVICES, INC.

AND

TOTAL SYSTEM SERVICES, INC.

Dated

August 4, 2005

# TABLE OF CONTENTS

1		BACKGROUND AND OBJECTIVES	1
	1.1	Background	1
	1.2	GOALS AND OBJECTIVES	1
	1.3	COLLATERAL AGREEMENTS	2
2		DEFINITIONS	3
	2.1	CERTAIN DEFINITIONS	3
	2.1	OTHER TERMS	13
	2.3	INCLUSION OF AFFILIATES IN DEFINITION OF CAPITAL ONE AND TSYS; MEANING OF "PARTY"	14
	2.4	RULES OF INTERPRETATION	14
3		SERVICES	15
	3.1	PROVISION OF SERVICES	15
	3.2	SERVICES PERFORMED BY CAPITAL ONE OR THIRD PARTIES	16
	3.3	PERMITTED USERS AND CAPITAL ONE ACCOUNTS	16
	3.4	RESPONSIBILITY FOR RESOURCES	18
	3.5	FACILITIES	18
	3.6	TRANSITION OF FUNCTIONS/CAPABILITIES TO [***] TS2	18
	3.7	SPECIAL TERMS AND CONDITIONS FOR [***] AND [***]	18
	3.8	OSFI Access to Data and Information	19
4		TERM	20
	4.1	INITIAL TERM	20
	4.2	RENEWAL	20
5		TSYS PERSONNEL	20
	5.1	KEY TSYS POSITIONS	20
		TCVC CRRVICES MANAGER	
	5.2	TSYS SERVICES MANAGER	21
	5.2 5.3	RESOURCE POOL	21 22
	5.3 5.4	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL	22 23
	5.3 5.4 5.5	RESOURCE POOL	22 23 23
	5.3 5.4 5.5 5.6	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES	22 23 23 23
	5.3 5.4 5.5 5.6 5.7	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT	22 23 23 23 23 25
	5.3 5.4 5.5 5.6	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES	22 23 23 23
6	5.3 5.4 5.5 5.6 5.7	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT	22 23 23 23 23 25
6	5.3 5.4 5.5 5.6 5.7	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS	22 23 23 23 25 26
6	5.3 5.4 5.5 5.6 5.7 5.8	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS	22 23 23 23 25 26 <b>26</b> 26
6	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT	22 23 23 25 26 <b>26</b> 27 27
6	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS	22 23 23 25 26 <b>26</b> 27 27 28
6	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4 6.5	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS OFFICE OF FOREIGN ASSETS CONTROL REGULATIONS	22 23 23 25 26 <b>26</b> 27 27 28 28
6	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS	22 23 23 25 26 <b>26</b> 27 27 28
6	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4 6.5	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS OFFICE OF FOREIGN ASSETS CONTROL REGULATIONS	22 23 23 25 26 <b>26</b> 27 27 28 28
	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4 6.5	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS OFFICE OF FOREIGN ASSETS CONTROL REGULATIONS USA PATRIOT ACT AND PROCEEDS OF CRIME (MONEY LAUNDERING) AND TERRORIST FINANCING ACT IN CANADA	22 23 23 25 26 <b>26</b> 26 27 27 28 28 28
7	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4 6.5	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS OFFICE OF FOREIGN ASSETS CONTROL REGULATIONS USA PATRIOT ACT AND PROCEEDS OF CRIME (MONEY LAUNDERING) AND TERRORIST FINANCING ACT IN CANADA  SERVICE LEVELS	22 23 23 25 26 <b>26</b> 26 27 27 28 28 28
7	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4 6.5 6.6	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS CRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS OFFICE OF FOREIGN ASSETS CONTROL REGULATIONS USA PATRIOT ACT AND PROCEEDS OF CRIME (MONEY LAUNDERING) AND TERRORIST FINANCING ACT IN CANADA  SERVICE LEVELS  PROPRIETARY RIGHTS CAPITAL ONE MATERIALS DEVELOPED MATERIALS DEVELOPED MATERIALS	22 23 23 23 25 26 26 26 27 27 28 28 28
7	5.3 5.4 5.5 5.6 5.7 5.8 6.1 6.2 6.3 6.4 6.5 6.6	RESOURCE POOL QUALIFICATIONS AND RETENTION OF TSYS PERSONNEL REMOVAL OF TSYS PERSONNEL COMPLIANCE WITH EACH PARTY'S REGULATIONS AND POLICIES USE OF SUBCONTRACTORS AND OTHER SUPPORT OVERSIGHT CONTROLS  COMPLIANCE WITH LAWS COMPLIANCE WITH LAWS GRAMM-LEACH-BLILEY ACT, AND SIMILAR LAWS BANK BRIBERY ACT AND FOREIGN CORRUPT PRACTICES ACT CONSUMER CREDIT PROTECTION LAWS OFFICE OF FOREIGN ASSETS CONTROL REGULATIONS USA PATRIOT ACT AND PROCEEDS OF CRIME (MONEY LAUNDERING) AND TERRORIST FINANCING ACT IN CANADA  SERVICE LEVELS  PROPRIETARY RIGHTS CAPITAL ONE MATERIALS	22 23 23 23 25 26 <b>26</b> 26 27 27 28 28 28 28

8.5

8.6

THIRD PARTY MATERIALS

WHOLLY NEW DEVELOPED MATERIALS

32

33

	8.7	Works Made for Hire	34
	8.8	PATENT RIGHTS	34
	8.9	[***] OF TSYS PATENTS	35
	8.10	BANKRUPTCY CODE SECTION 365(N)	35
	8.11	RESIDUAL KNOWLEDGE AND FURTHER RIGHTS	35
	8.12	EXPORT	36
	8.13	TS2 LICENSE OPTION	36
9		PROJECT AND CONTRACT MANAGEMENT	37
	9.1	STEERING COMMITTEE	37
	9.2	REPORTS AND MEETINGS	37
	9.3	PROCEDURES MANUAL	38
	9.4	CHANGE CONTROL	39
	9.5	TECHNOLOGY PLANNING	40
	9.6	TS2 DEVELOPMENT PRIORITIES	40
	9.7	QUALITY ASSURANCE AND IMPROVEMENT PROGRAMS	40
	9.8	COORDINATION OF ADDITIONAL MARKETING TO CAPITAL ONE	40
	9.9	NO AMENDMENT AND ORDER OF PRECEDENCE	41
10		CAPITAL ONE RESPONSIBILITIES	41
	10.1	RESPONSIBILITIES	41
	10.2	SAVINGS CLAUSE	42
11		CHARGES AND PAYMENT TERMS	42
12		AUDITS	43
	12.1	GENERALLY	43
	12.2	SAS 70 AUDITS	43
	12.3	REGULATORY AUDITS	44
	12.4	CAPITAL ONE AUDITS	44
	12.5	AUDIT FOLLOW-UP AND NOTIFICATION	47
	12.6	RECORDS RETENTION	47
	12.7	OVERCHARGES	48
13		SAFEGUARDING OF DATA; CONFIDENTIALITY	48
	13.1	SAFEGUARDING CAPITAL ONE DATA	48
	13.2	GLB COMPLIANCE	49
	13.3	CONFIDENTIALITY	50
	13.4	DISCLOSURES REQUIRED BY LAW	54
	13.5	NO IMPLIED RIGHTS	54
	13.6	RETURN OF CONFIDENTIAL INFORMATION	54
	13.7	CORPORATE RISK CONTROLS AND INFORMATION SECURITY REVIEW	55
14		REPRESENTATIONS AND WARRANTIES	56
	14.1	SYSTEM QUALITY AND COMPLIANCE	56
	14.2	[***] PERFORMANCE	56
	14.3	Work Standards	56
	14.4	EFFICIENCY AND COST EFFECTIVENESS	57
	14.5	TECHNOLOGY	57
	14.6	OPEN SOURCE SOFTWARE	57
	14.7	Non-Infringement	57
	14.8	OWNERSHIP OR AUTHORIZATION TO USE	57
	14.9	AUTHORIZATION AND OTHER CONTRACTS	58
	14.10	INDUCEMENTS	58
	14.11	USER DOCUMENTATION	58

58

14.12

VIRUSES

	14.13	DISABLING CODE	58
	14.14	Deliverables	59
	14.15	DISCLAIMER	59
15		INSURANCE	59
	15.1	Insurance Coverage	59
	15.2	Insurance Provisions	60
	15.3	RISK OF LOSS	61
16		INDEMNITIES	61
	16.1	INDEMNITY BY TSYS	61
	16.2	INDEMNITY BY CAPITAL ONE	62
	16.3	ADDITIONAL INDEMNITIES	63
	16.4	Infringement	63
	16.5	INDEMNIFICATION PROCEDURES	65
	16.6	SUBROGATION	66
17		LIABILITY AND FORCE MAJEURE	66
	17.1	General Intent	66
	17.2	LIABILITY RESTRICTIONS	67
	17.3	FORCE MAJEURE	68
	17.4	LIMITATION OF ACTIONS	69
18		DISPUTE RESOLUTION	69
	18.1	Informal Dispute Resolution	69
	18.2	Arbitration	70
	18.3	CONTINUED PERFORMANCE	71
	18.4	Injunctive Relief; Specific Performance	71
	18.5	GOVERNING LAW	71
	18.6	EXPEDITED DISPUTE RESOLUTION	72
	18.7	BINARY ARBITRATION	72
19		TERMINATION	73
	19.1	TERMINATION FOR CAUSE	73
	19.2	TERMINATION FOR CONVENIENCE	74
	19.3	TERMINATION DUE TO ELECTION OF LICENSE OPTION	74
	19.4	TERMINATION DUE TO FORCE MAJEURE EVENT	74
	19.5	TERMINATION DUE TO [***] CONSTRAINTS	75 
	19.6	TERMINATION DUE TO TSYS' FINANCIAL INABILITY TO PERFORM	75
	19.7 19.8	EXTENSION OF TERMINATION/EXPIRATION EFFECTIVE DATE TERMINATION/EXPIRATION ASSISTANCE	75 76
20			
20	20.4	GENERAL	77
	20.1	BINDING NATURE AND ASSIGNMENT	77
	20.2 20.3	No Collateral Documents Mutually Negotiated	77 78
	20.3	NOTICES	78 78
	20.4	COUNTERPARTS	78 79
	20.6	RELATIONSHIP OF PARTIES	80
	20.7	SEVERABILITY	80
	20.8	CONSENTS AND APPROVAL	80
	20.9	WAIVER OF DEFAULT; CUMULATIVE REMEDIES	80
	20.10	SURVIVAL	81
	20.11	Public Disclosures	81

81

20.12

SERVICE MARKS

20.13	THIRD PARTY BENEFICIARIES	81
20.14	NONSOLICITATION OF EMPLOYEES	81
20.15	COVENANT OF GOOD FAITH AND FAIR DEALING	82
20.16	ENTIRE AGREEMENT; AMENDMENT	82

# **SCHEDULES**

Schedule A: Services
Schedule B: Service Levels
Schedule C: Charges

Schedule D: Key TSYS Positions

Schedule E: [\*\*\*] List

Schedule F: Capital One Competitors
Schedule G: Permitted Subcontractors
Schedule H: Form of TS2 License Agreement

Schedule I: Form of Non-Disclosure/[\*\*\*] Agreements

Schedule J: Initial List of TSYS Personnel with Access to [\*\*\*] Capital One

[\*\*\*]

Schedule K: Third Party Materials

Schedule L: Initial Specifications for the TSYS System
Schedule M: Form of Commencement Date Acknowledgement
Schedule N: Capital One Initial Conversion Requirements

Schedule O: Disclosure Schedule Schedule P: Form of [\*\*\*]

Schedule Q: Form of [\*\*\*] Escrow Agreement

Schedule R: Reports and Meetings
Schedule S: Form of COF Guarantee
Schedule T: Form of [\*\*\*] Addendum
Schedule U: Travel and Expense Guidelines
Schedule V: Screening Requirements
Schedule W: TSYS Facilities

Schedule W: TSYS Facilities
Schedule X: List of [\*\*\*] Capital One

Schedule Y: Procedures Manual Table of Contents

#### PROCESSING SERVICES AGREEMENT

PROCESSING SERVICES AGREEMENT made and effective as of August 4, 2005 (the "Effective Date"), by and between CAPITAL ONE SERVICES, INC. ("Capital One"), a Delaware corporation having offices at 1680 Capital One Drive, McLean, Virginia 22102-3491, and TOTAL SYSTEM SERVICES, INC. ("TSYS"), a Georgia corporation having offices at 1600 First Avenue, Columbus, Georgia 31901-1804.

# 1 BACKGROUND AND OBJECTIVES

#### 1.1 Background

This Agreement is being made and entered into with reference to the following:

- (a) Capital One is (directly or indirectly) a subsidiary of Capital One Financial Corporation ("<u>COF</u>"), which through its banking subsidiaries is one of North America's largest consumer finance businesses, with large portfolios of credit cards and other consumer and small business finance accounts. Capital One currently performs the necessary data processing of such accounts through its own proprietary computer systems;
- (b) TSYS is one of North America's largest providers of account processing services for consumer finance companies, and provides such services through a number of computer systems, including its newest system known as TS2;
- (c) Capital One desires that certain transaction and account processing, software development, and related activities be provided and managed by an experienced and capable vendor skilled in the performance of these functions;
  - (d) Capital One has also licensed TSYS' [\*\*\*] system pursuant to a [\*\*\*] agreement; and
  - (e) TSYS wishes to perform the services provided for herein for Capital One.

## 1.2 Goals and Objectives

The goals and objectives set forth in this Section 1.2 are intended to be a general introduction to this Agreement and are not intended to expand the scope of the parties' obligations under this Agreement or to alter the plain meaning of the terms and conditions of this Agreement. Moreover, the parties acknowledge that such goals and objectives are not necessarily obligations of either party, and neither party shall be liable for failing to achieve any such goals or objectives except to the extent otherwise provided in other provisions of this Agreement, including to the extent a goal or objective is made an obligation of a party under other provisions of this Agreement. For purposes of the informal dispute resolution process under Section 18.1, the parties shall use the following goals and objectives as a guide. However, for purposes of any formal dispute resolution process under any other provision of Section 18, such goals and objectives shall be considered only to the extent the terms and conditions of this Agreement are unclear or ambiguous, and in such case such terms and conditions are to be interpreted and construed in light of the following goals and objectives.

Subject to the foregoing, Capital One and TSYS have agreed upon the following specific goals and objectives for the relationship resulting in this Agreement:

- (a) providing Capital One access to and use of a market competitive card transaction and account processing platform on a basis that does not enable Capital One to become a competitor to TSYS in the third party account processing business;
- (b) establishing and maintaining a joint development pool (of TSYS and Capital One resources) to implement changes to [\*\*\*] TS2, including (if requested by Capital One) new functionality developed by TSYS for its then-existing TS2 Processing Platform, custom enhancements and changes required to meet regulatory and Association requirements, so that Capital One resources develop and retain an understanding of [\*\*\*] TS2;
  - (c) providing Capital One a reliable processing platform and environment in which the Services are performed primarily on [\*\*\*] to Capital One;
- (d) establishing a relationship in which TSYS has responsibility for providing, maintaining, and supporting the Services, and recognizes the [\*\*\*] impact of Service disruptions on Capital One;
- (e) providing a pricing structure that is designed to: (i) provide Capital One competitive pricing in comparison to other appropriately comparable TSYS customers; (ii) motivate TSYS to disclose and provide to Capital One the benefit of any lower pricing provided to such TSYS customers without action on the part of Capital One; and (iii) provide a reasonable overall profit to TSYS;
  - (f) forming a relationship that is designed to enable Capital One to achieve faster time to market and more flexibility than it could achieve on its own;
- (g) achieving a contractual arrangement that is responsive to the changing demands of Capital One's businesses and Capital One's requests to implement new TSYS products and enhancements within [\*\*\*] TS2, while also accommodating material changes in volumes;
- (h) ensuring that Capital One is [\*\*\*] additional functionality and products provided or made available by TSYS to its other [\*\*\*] TS2, [\*\*\*], and [\*\*\*] customers;
- (i) creating incentive arrangements to encourage superior performance while providing Capital One the flexibility to change service vendors or resume internal operations;
  - (j) providing Capital One the right to license the software used by TSYS to provide the Services; and
  - (k) protecting the parties' respective intellectual property.

# 1.3 Collateral Agreements

Simultaneously with the execution and delivery of this Agreement, the parties or their respective Affiliates have executed and delivered the TS2 License Agreement, the [\*\*\*] Agreement, the COF Guarantee, and the [\*\*\*] Addendum. In addition, the parties shall use Commercially Reasonable Efforts to cause the third party escrow agent to execute the [\*\*\*] Escrow Agreement within [\*\*\*] days of the Effective Date. The parties have agreed to the terms of the [\*\*\*] Escrow Agreement included in Schedule Q and shall be reasonable in accommodating changes thereto that are required by the third party escrow agent.

#### 2 DEFINITIONS

#### 2.1 Certain Definitions

As used in this Agreement:

- (a) "AAA" shall mean the American Arbitration Association and its successors.
- (b) "Account" shall mean a unique representation of the data and current financial status of a customer account relationship for a credit, debit, private label, small business, line of credit, stored value or closed end account issued to a person who either currently has or at the time of origination had a United States or Canadian mailing address, and such other accounts as the parties may agree to from time to time.
  - (c) "Actual Migration Completion Date" shall have the meaning given in Section 13.4(b) of Schedule A.
- (d) "Affiliate" shall mean, with respect to any entity, any other entity Controlling, Controlled by, or under common Control with, such entity at the time in question.
- (e) "Agreement" shall mean this Processing Services Agreement, including the Main Body of the Agreement and all schedules, exhibits, and attachments hereto (other than the TS2 License Agreement, the [\*\*\*] Agreement, and the [\*\*\*] Escrow Agreement), as may be amended by the parties in accordance with Section 20.16.
  - (f) "Applicable Laws" shall mean:
    - (i) any law, statute, regulation, ordinance, or subordinate legislation in force from time to time to which a party or its Affiliates is subject;
    - (ii) the common law as applicable to the parties from time to time;
    - (iii) any court order, judgement, or decree that is binding on a party or its Affiliates; and
  - (iv) any directive, policy, rule, or order that is binding on a party or its Affiliates and that is made or given by any Association, Regulator, or other government or government agency, of, in the case of items (i) through (iv) above, any (x) country, or other national, federal, commonwealth, state, provincial, or local jurisdiction, (y) Association, or (z) exchange or association whose regulations are binding on either party pursuant to a self-regulating mechanism approved by a governmental entity.
- (g) "<u>Applications Software</u>" or "<u>Application</u>" shall mean Software (including the supporting documentation, media, on-line help facilities, and tutorials) that performs specific user related data processing and telecommunications tasks in connection with the Services. The Software portions of TS2, [\*\*\*], and [\*\*\*] are each Applications Software.
- (h) "Association(s)" shall mean Visa and MasterCard and any other card associations with which any of the Capital One Accounts receiving the Services may be associated from time to time, including American Express and Discover Card.
  - (i) "Bankruptcy Code" shall have the meaning given in Section 8.10.

- (j) "BIN/ISO" shall mean a bank identification number or prefix used to identify Association member transactions and groups of Accounts.
- (k) "Business Day" shall mean every day Monday through Friday other than those holidays when Capital One's corporate headquarters is not scheduled to be open for business. References in this Agreement to "days" that do not specifically refer to Business Days are references to calendar days.
- (l) "<u>Canadian Privacy Legislation</u>" shall mean the Personal Information and Electronics Documents Act, S.C. 2001, c-5, the Act Respecting the Protection of Personal Information in the Private Sector, R.S.Q. c. P-39.1, the Personal Information Protection Act, S.B.C. 2003, c-63, the Personal information Protection Act, R.S.A., c. P-65, and any similar legislation applicable in Canada.
  - (m) "Capital One Account" shall have the meaning given in Section 3.3(b).
  - (n) "Capital One Audit" shall have the meaning given in Section 12.4.
  - (o) "Capital One Auditors" shall have the meaning given in Section 12.1.
- (p) "Capital One Competitor" shall mean each of the entities listed in Schedule F and their respective Affiliates, as such Schedule may be updated from time to time by Capital One in its sole discretion, provided the list may not contain [\*\*\*] entities any [\*\*\*].
  - (q) "Capital One Confidential Information" shall have the meaning given in Section 13.3(a).
  - (r) "Capital One Contract Manager" shall have the meaning given in Section 10.1(a).
- (s) "Capital One Data" shall mean information of Capital One entered into the TSYS System (and/or, to the extent [\*\*\*] stores data separately from the TSYS System and TSYS is providing hosting services, [\*\*\*]) by or on behalf of Capital One and information derived from such information, including as stored in or processed through the TSYS System (or [\*\*\*], as the case may be).
  - (t) "Capital One Designee" shall have the meaning given in Section 19.8.
  - (u) "Capital One Developed Materials" shall mean the following:
  - (i) Developed Materials that (A) are modifications to, or derivative works, upgrades, or enhancements of, Capital One Materials and (B) do not incorporate any TSYS Materials and are not modifications to, or derivative works, upgrades or enhancements of, TSYS Materials; and
  - (ii) Non-Software Developed Materials that are predominantly comprised of pre-existing Capital One Materials. For example, pre-existing Capital One training materials, documentation, business plans, and similar non-Software materials that are modified within the scope of this Agreement using TSYS Materials would be Capital One Developed Materials.
  - (v) "Capital One Directions" shall have the meaning given in Section 6.1(b).
  - (w) "Capital One Initial Conversion Requirements" shall mean the functionality, scalability, and business requirements attached as Schedule N.

- (x) "Capital One [\*\*\*]" shall have the meaning given in Section 13.3(e).
- (y) "Capital One Materials" shall mean the following:
- (i) all Software and non-Software materials, such as business plans, credit risk models, optimization models, authorization models, marketing and product strategies, Capital One process and procedure descriptions, manuals, and training materials, owned or developed by Capital One or a Capital One Affiliate;
- (ii) Third Party Materials licensed to Capital One or any Capital One Affiliate and used by, or made available to, TSYS in connection with this Agreement; and
  - (iii) Capital One Developed Materials.
- (z) "Capital One Legal Requirements" shall have the meaning given in Section 6.1.
- (aa) "Capital One Operating Environment" shall mean the information technology software, equipment, systems, processes, and procedures used by any of Capital One and Capital One Affiliates, and their respective suppliers, to interface with the TSYS System, [\*\*\*], and [\*\*\*] and to access and utilize the Services. The Capital One Operating Environment shall not include any portion of the TSYS System or, so long as TSYS is providing hosting services in connection therewith, the portions of [\*\*\*] for which TSYS is providing hosting services.
- (bb) "[\*\*\*] TS2" shall mean the [\*\*\*] version of TS2 that TSYS uses (along with [\*\*\*]) to provide Services to Capital One under this Agreement. At the Commencement Date, [\*\*\*] TS2 shall be a [\*\*\*] of TS2 that has been [\*\*\*] for in the Initial Conversion Plan. Thereafter, [\*\*\*] TS2 shall be [\*\*\*] for [\*\*\*] during the Term as provided in this Agreement.
- (cc) "[\*\*\*] Failure" shall mean a failure of [\*\*\*] of the TSYS System and, as a result of [\*\*\*], the [\*\*\*], to be available for use for Capital One Accounts for a [\*\*\*].
  - (dd) "Change Control Procedure" shall have the meaning given in Section 9.4.
  - (ee) "Charges" shall have the meaning given in Section 1.1 of Schedule C.
  - (ff) "COF" shall have the meaning given in Section 1.1(a).
  - (gg) "COF Guarantee" shall mean the agreement between TSYS and COF in the form of Schedule S.
- (hh) "Commencement Date" shall mean the date on which the final Conversion event identified in the Initial Conversion Plan is completed. Within [\*\*\*] days after such date, the parties shall execute an acknowledgement thereof in the form of Schedule M.
- (ii) "Commercially Reasonable Efforts" shall mean taking such steps and performing in such a manner as a well managed business would when such business is acting in a determined, prudent, and reasonable manner to achieve a particular desired result for its own benefit.
  - (jj) "Compliance Enhancement" shall have the meaning given in Section 6.1(e).
  - (kk) "Confidential Information" shall have the meaning given in Section 13.3(a).

- (ll) "Control" and its derivatives mean with regard to any entity the legal, beneficial or equitable ownership, directly or indirectly, of more than [\*\*\*] percent ([\*\*\*]%) of the capital stock (or other ownership interest if not a stock corporation) of such entity ordinarily having voting rights.
  - (mm) "Conversion" and its derivatives (such as "Converted") shall mean:
  - (i) the transfer of data relating to Capital One Accounts from the processing system(s) in which that data currently resides and is processed to the TSYS System (and, as applicable, [\*\*\*]);
  - (ii) the migration from the current processing system(s) to the TSYS System (and, as applicable, [\*\*\*]) for the performance of the various processing functions;
  - (iii) implementation of all functionality contained in the applicable Specifications and completion of the activities in the applicable Conversion Plan; and
    - (iv) all activities necessary to complete the activities described in items (i), (ii), and (iii) above.
  - (nn) "Conversion Plan" shall have the meaning given in Section 3.1 of Schedule A.
  - (oo) "Customer Information" shall have the meaning given in Section 13.2.
  - (pp) "Default Rate" shall have the meaning given in Section 8.5(c) of Schedule C.
- (qq) "<u>Deliverables</u>" shall mean each of the Software, documentation, and other deliverables TSYS is required to deliver under this Agreement (including the Conversion Plans) and any other deliverables agreed upon by the parties in writing.
- (rr) "<u>Developed Materials</u>" shall mean all [\*\*\*] that are developed by TSYS and/or the Development Pool within the scope of this Agreement, which includes all of the development described in the Initial Conversion Plan. Developed Materials will [\*\*\*] (i) any materials developed by or on behalf of either party outside of the scope of this Agreement; (ii) any Wholly New Developed Materials; [\*\*\*] (iii) any modifications to the [\*\*\*].
  - (ss) "Development Pool" shall have the meaning given in Section 4.1 of Schedule C.
  - (tt) "Development Methodology" shall mean the TSYS development methodology set forth in the Procedures Manual applicable to TS2.
  - (uu) "Dispute Date" shall have the meaning given in Section 18.1(a).
  - (vv) "[\*\*\*] Account" shall have the meaning given in Section 3.3(e).

- (ww) "Documentation" shall mean TSYS documentation that:
- (i) instructs or assists users, operators, and system personnel in the operation and use of the TSYS System, [\*\*\*] (as applicable), and the Services; and/or
- (ii) describes the features, functions, and operation of the TSYS System, [\*\*\*], and the Services, as such documentation exists on the Effective Date and is updated from time to time, including the Procedures Manual.
- (xx) "Effective Date" shall have the meaning given in the preamble of this Agreement.
- (yy) "Equipment" shall mean the computer and telecommunications equipment (without regard to the entity owning or leasing such equipment) used by TSYS to provide the Services. Equipment includes any of the following that are used by TSYS to provide the Services: (i) computer equipment, including associated attachments, features, accessories, peripheral devices, and other computer equipment; (ii) telecommunications equipment, including private branch exchanges, multiplexors, modems, CSUs/DSUs, hubs, bridges, routers, switches, and other telecommunications equipment; and (iii) related services (e.g., maintenance and support services, upgrades, subscription services) provided by third parties (e.g., manufacturer and lessor) in the same contract covering the provision of such Equipment.
  - (zz) "[\*\*\*]" shall mean the third party software used to [\*\*\*] that Capital One licenses pursuant to a separate agreement with [\*\*\*].
  - (aaa) "Force Majeure Event" shall have the meaning given in Section 17.3.
- (bbb) "GLB Act" shall mean the Gramm-Leach-Bliley Act, 15 USC §6801 et seq., and the implementing regulations and regulatory interpretations thereto, as amended from time to time.
  - (ccc) "GLB Regulations" shall have the meaning given in Section 13.2.
  - (ddd) "Intellectual Property Rights" shall mean, collectively, Patent Rights and Non-Patent Intellectual Property Rights.
- (eee) "[\*\*\*] <u>List</u>" shall mean the list of functionalities set forth in the attached Schedule E as of the Effective Date and any other functionalities [\*\*\*] such list pursuant to Section 8.3.
  - (fff) "[\*\*\*] List Materials" shall mean TSYS Developed Materials that are developed to provide functionalities on the [\*\*\*] List.
  - (ggg) "Initial Conversion" shall have the meaning given in Section 3.4 of Schedule A.
  - (hhh) "Initial Conversion Plan" shall have the meaning given in Section 3.4 of Schedule A.
  - (iii) "Initial Term" shall have the meaning given in Section 4.1.
- (jjj) "[\*\*\*]" shall mean the [\*\*\*] held by [\*\*\*] (and their successors and assigns) listed in Schedule X and any additional [\*\*\*] which Capital One [\*\*\*] from either or both of them after the Effective Date.

- (kkk) "Key TSYS Positions" shall have the meaning given in Section 5.1.
- (lll) "Licensed Materials" shall have the meaning given in the TS2 License Agreement.
- (mmm) "List" shall mean the [\*\*\*] List or the [\*\*\*] List.
- (nnn) "<u>Losses</u>" shall mean all losses, liabilities, damages, and claims, and all related costs and expenses (including reasonable legal fees and disbursements and costs of investigation, litigation, settlement, judgment, interest, and penalties).
  - (000) "Main Body of the Agreement" shall mean, collectively, the preamble and Sections 1 through 20 of this Agreement.
- (ppp) "MasterCard" shall mean, individually or collectively, as appropriate, MasterCard® International Incorporated and its Affiliates and their respective successors and assigns.
  - (qqq) "Migration" shall have the meaning given in Section 13.1 of Schedule A.
  - (rrr) "Migration Plan" shall have the meaning given in Section 13.1 of Schedule A.
  - (sss) "[\*\*\*]" shall have the meaning given in Section 3.6(a) of Schedule C.
  - (ttt) "Monthly Performance Report" shall have the meaning given in Section 9.2(a).
  - (uuu) "New Additional Services" shall have the meaning given in Section 5.2 of Schedule C.
  - $(vvv) \ \hbox{$"$\underline{Non-Patent}$ \underline{Intellectual}$ \underline{Property}$ \underline{Rights}" shall mean, on a worldwide basis, any and all:$
  - (i) rights associated with works of authorship and literary property, including copyrights, moral rights of an author of a copyrightable work (including any right to be identified as the author of the work or to object to derogatory treatment of the work), and mask-work rights;
    - (ii) trade marks, service marks, logos, trade dress, trade names, whether or not registered, and the goodwill associated therewith;
  - (iii) rights relating to know-how or trade secrets, including ideas, concepts, methods, techniques, inventions (whether or not developed or reduced to practice);
    - (iv) rights in domain names, universal resource locator addresses, telephone numbers (including toll free numbers), and similar identifiers;
  - (v) other intellectual and industrial property rights (other than Patent Rights) of every kind and nature, however designated, whether arising by operation of law, contract, license or otherwise; and

- (vi) registrations, initial applications (including intent to use applications), renewals, extensions, continuations, divisions, or reissues thereof now or hereafter in force (including any rights in any of the foregoing).
- (www) "Notice of Election" shall have the meaning given in Section 16.5.
- (xxx) "Object Code" shall mean computer programming code (including microcode, as applicable) in binary form. Object Code is directly executable by a computer after processing, but without compilation or assembly.
- (yyy) "[\*\*\*] List" shall mean a [\*\*\*] number of [\*\*\*] designated by Capital One that Capital One desires to [\*\*\*] after the Commencement Date, that are [\*\*\*] to give Capital One a [\*\*\*], that at the time of [\*\*\*] Capital One [\*\*\*] the Development Pool [\*\*\*], and that at the time of [\*\*\*] TSYS' Development Pipeline or [\*\*\*] in the [\*\*\*]. Additionally, if any [\*\*\*] on the [\*\*\*] List is not [\*\*\*] by the Commencement Date it shall be automatically [\*\*\*] on the [\*\*\*] List.
- (zzz) "<u>Out-of-Pocket Expenses</u>" shall mean reasonable, demonstrable, and actual out-of-pocket expenses that are incurred by TSYS in connection with providing the Services, but not including TSYS' overhead costs (or allocations thereof), administrative expenses, or other mark-ups. Out-of-Pocket Expenses shall be calculated at TSYS' actual incremental expense and shall be net of all rebates and allowances received by TSYS or its Affiliates. Where TSYS incurs an Out-of-Pocket Expense from an Affiliate of TSYS, the amount charged to Capital One shall be no greater than would have been paid to such Affiliate in an arm's length transaction between unaffiliated companies at fair market value. Pass Through Expenses are not Out-of-Pocket Expenses.
  - (aaaa) "Pass-Through Expenses" shall have the meaning given in Section 8.1 of Schedule C.
- (bbbb) "Patent Rights" shall mean rights in patents and patentable subject matter, and patent registrations, initial applications, renewals, extensions, continuations, divisions, or reissues thereof now or hereafter in force (including any rights in any of the foregoing).
  - (cccc) "Permitted Subcontractors" shall mean those subcontractors of TSYS:
    - (i) identified in Schedule G;
    - (ii) whose approval by Capital One is not required pursuant to Section 5.7; and
    - (iii) any other subcontractor(s) approved by Capital One pursuant to Section 5.7,
    - unless and until (in each case of (i), (ii), and (iii)) approval of such subcontractor is rescinded by Capital One in accordance with Section 5.7.
  - (dddd) "Permitted Users" shall have the meaning given in Section 3.3(a).
- (eeee) "Personally Identifiable Information" shall mean any information that (i) is defined as "personal information" under the GLB Act or Canadian Privacy Legislation or (ii),

alone or in combination with other information, relates to a specific, identifiable individual person. Personally Identifiable Information includes individual names, social security numbers, telephone numbers, home address, driver's license number, account number, email address, and vehicle registration number. Any information that can be associated with Personally Identifiable Information shall also be Personally Identifiable Information. For example, an individual's age alone is not Personally Identifiable Information, but if such age were capable of being associated with one or more specific, identifiable, individuals then such age would be deemed Personally Identifiable Information.

(ffff) "[\*\*\*]" shall mean the strategy management system that TSYS sublicenses to Capital One pursuant to the [\*\*\*] Addendum.

(gggg) "[\*\*\*] <u>Addendum</u>" shall mean the addendum to this Agreement attached as Schedule T, which has been executed by the parties simultaneously with the execution and delivery of this Agreement.

(hhhh) "Portfolio" shall mean an aggregation of Accounts, as designated by Capital One.

(iiii) "Procedures Manual" shall have the meaning given in Section 9.3.

(jjjj) "Processing Year" shall mean each consecutive twelve (12) month period measured from (i) the Commencement Date if the Commencement Date is on the first day of a month or the first day of the month following the Commencement Date if the Commencement Date is not on the first day of a month, or (ii) an anniversary of the first day of the first Processing Year. For example, if the first Processing Year (Processing Year 1) began on June 1, 2007, it would end on May 31, 2008 and Processing Year 2 would begin on June 1, 2008 and end on May 31, 2009. If the final Processing Year is less than twelve (12) months, non-specific references to amounts for such Processing Year shall be appropriately and equitably pro-rated.

(kkkk) "[\*\*\*]" shall mean the web-based process management system licensed [\*\*\*]pursuant to the [\*\*\*] Agreement.

(Illl) "[\*\*\*] Agreement" shall mean the agreement entered into between Capital One and TSYS in the form of Schedule P.

(mmmm) "Regulator" shall mean a governmental authority, or a non-governmental entity such as the National Association of Securities Dealers (NASD) or New York Stock Exchange (NYSE), that is charged with monitoring or overseeing the business practices of the respective parties or their Affiliates, including (as applicable) the Federal Financial Institutions Examination Council (FFIEC), the Board of Governors of the Federal Reserve System (FRB), the Federal Deposit Insurance Corporation (FDIC), the National Credit Union Administration (NCUA), the US Office of the Comptroller of the Currency (OCC), the US Office of Thrift Supervision (OTS), the Canadian Office of the Superintendent of Financial Institutions (OSFI), and the Financial Consumer Agency of Canada (FCAC) or any successor bodies that regulate consumer finance companies and/or financial service providers.

(nnnn) "Renewal Term" shall have the meaning given in Section 4.2.

(0000) "Resource Pool" shall have the meaning given in Section 5.3.

(pppp) "SAS 70" shall have the meaning given in Section 12.2.

- (qqqq) "SAS 70 Auditor" shall have the meaning given in Section 12.2.
- (rrrr) "[\*\*\*] TS2" shall mean the [\*\*\*] platform used by TSYS to perform Services not performed using [\*\*\*] TS2, [\*\*\*], or [\*\*\*];
- (ssss) "Services" shall have the meaning given in Section 3.1.
- (tttt) "Service Levels" shall have the meaning given in Section 7.
- (uuuu) "<u>Software</u>" shall mean (unless a more specific reference is provided) software programs and programming (including the supporting documentation, media, on-line help facilities, and tutorials) as those terms are generally understood in the information technology industry.
  - (vvvv) "Source Code" shall mean computer programming code (including microcode, as applicable) other than Object Code, including:
  - (i) code that may be displayed in a form readable and understandable by a programmer of ordinary skill, as well as any enhancements, corrections, and documentation related thereto; and
    - (ii) all related Source Code level system documentation, comments, procedural code (such as job control language), and explanatory materials.
- (wwww) "[\*\*\*] <u>Escrow Agreement</u>" shall mean the agreement among TSYS, Capital One, and the third party escrow agent designated therein, in the form of Schedule Q. The parties acknowledge that the [\*\*\*] to [\*\*\*] is owned by an unaffiliated third party and will not be deposited under the [\*\*\*] Escrow Agreement.
- (xxxx) "<u>Specifications</u>" shall mean the written materials (including the relevant Documentation) that describe the functions and features of the TSYS System, [\*\*\*], or other Deliverables, and any of their component parts. The Specifications for the Initial Conversion consist of:
  - (i) the TS2, [\*\*\*], and [\*\*\*] Documentation;
  - (ii) the mutually agreed list of [\*\*\*] TSYS shall [\*\*\*] during the Initial Conversion to [\*\*\*] TS2, [\*\*\*], and [\*\*\*] to meet the Capital One Initial Conversion Requirements (the "[\*\*\*] List");
    - (iii) the mutually agreed list of [\*\*\*] necessary for Permitted Users to Access and Use the Services ("[\*\*\*] List");
    - (iv) the list of [\*\*\*] TSYS has committed to [\*\*\*] for TS2 prior to the Commencement Date ("[\*\*\*] List"); and
    - (v) the Initial Capital One [\*\*\*] Requirements designated as "[\*\*\*] Requirements".
- The [\*\*\*] List, [\*\*\*] List, [\*\*\*] List, and the [\*\*\*] Requirements as they exist on the Effective Date are attached as Schedule L.
  - (yyyy) " $\underline{\text{Steering Committee}}$ " shall have the meaning given in Section 9.1.
  - (zzzz) "Systems Software" shall mean those programs and programming (including the supporting documentation, media, on-line help facilities and tutorials) that

perform (i) tasks basic to the functioning of the Equipment and which are required to operate the Applications Software; or (ii) tasks, other than as performed by Applications Software, otherwise supporting the provision of the Services by TSYS. Software supporting the Services that are not Applications Software shall be deemed to be Systems Software. Systems Software includes mainframe and mid-range operating systems, server operating systems, network operating systems, systems utilities (including measuring and monitoring tools), data security software, middleware, database management systems, development tools (other than development tools specific to a particular item of Applications Software which is provided by the licensor of such Applications Software) and telecommunications monitors.

(aaaaa) "Technology Subcommittee" shall have the meaning given in Section 9.5(a).

(bbbbb) "Term" shall mean the Initial Term and any Renewal Terms.

(cccc) "Termination/Expiration Assistance" shall have the meaning given in Section 19.8.

(ddddd) "Third Party Materials" shall mean all Software and non-Software materials, such as business plans, process and procedure descriptions, manuals, and training materials owned by a third party and licensed to either Capital One or TSYS for use in connection with this Agreement.

(eeeee) "[\*\*\*]" shall mean the systems for processing and managing incentive rewards point calculations operated by TSYS through [\*\*\*].

(fffff) "TS2 License Agreement" shall mean the agreement entered into between TSYS and Capital One in the form of Schedule H.

(ggggg) "TS2 Processing Platform" or "TS2" shall mean the TOTAL SYSTEM II processing system, including all subsystems thereto and all related development and testing facilities, processes, procedures, and software, as the same may be enhanced, modified, and updated from time to time during the Term.

(hhhhh) "TSYS Confidential Information" shall have the meaning given in Section 13.3(a).

(iiii) "TSYS Developed Materials" shall mean all Developed Materials that are not Capital One Developed Materials.

(jjjjj) "TSYS' Development Pipeline" shall mean the written list of product development projects and ideas that TSYS is [\*\*\*] within the following [\*\*\*] year period as reflected in the [\*\*\*] by TSYS in the ordinary course of business, as the same is updated from time to time.

(kkkkk) "TSYS Facilities" shall have the meaning given in Section 3.5(a).

(lllll) "TSYS Materials" shall mean:

(i) all Software and non-Software materials, such as business plans, process and procedure descriptions, manuals, and training materials, that are not Developed Materials or Wholly New Developed Materials and that are owned or developed by TSYS or any of its Affiliates;

- (ii) Third Party Materials licensed to TSYS or any Affiliate of TSYS; and
- (iii) TSYS Developed Materials.

(mmmmm) "TSYS Personnel" shall mean employees and individual independent contractors of TSYS and its Permitted Subcontractors assigned to perform the Services.

(nnnnn) "TSYS Services Manager" shall have the meaning given in Section 5.2.

(00000) "TSYS System" shall mean all Equipment, Software, systems, interfaces, processes, and procedures used or necessary for use by TSYS to perform the Services. The TSYS System includes:

- (i) [\*\*\*] TS2;
- (ii) [\*\*\*] TS2;
- (iii) [\*\*\*];
- (iv) [\*\*\*];
- (v) so long as Capital One is acquiring hosting services with respect to [\*\*\*], or other system used to perform the Services, then all Equipment, Software, systems, interfaces, processes, and procedures used or necessary for use to perform such hosting; and
  - (vi) all Documentation relating to the TSYS System as a whole or the individual components making up the TSYS System.
- (ppppp) "[\*\*\*] Materials" shall have the meaning given in Section [\*\*\*].
- (qqqqq) "<u>Use</u>" shall mean, in the context of Intellectual Property Rights, to use, copy, maintain, modify, enhance, or create derivative works.
- (rrrrr) "Virus" shall mean:
- (i) program code or programming instruction or set of instructions intentionally designed to disrupt, disable, harm, interfere with or otherwise adversely affect computer programs, data files or operations; or
  - (ii) other code typically described as a virus or by similar terms, including Trojan horse, worm or backdoor.
- (sssss) "VISA" shall mean, individually or collectively, as appropriate, VISA U.S.A., Inc. or VISA International and their respective Affiliates, and their respective successors and assigns.
- (ttttt) "Wholly New Developed Materials" shall mean all Software and non-Software materials that are developed by TSYS and/or the Development Pool and that are not [\*\*\*] of, Capital One Materials or TSYS Materials.

## 2.2 Other Terms

Other capitalized terms used in this Agreement are defined where they are used and have the meanings there indicated. Those terms, acronyms, and phrases that are not defined in this

Agreement but that are utilized in the information technology services or credit card processing industry or other pertinent business context shall be interpreted in accordance with their generally understood meaning in such industry or business context.

# 2.3 Inclusion of Affiliates in Definition of Capital One and TSYS; Meaning of "party"

- (a) As used in this Agreement, references to "Capital One" include Affiliates of Capital One in accordance with the following:
  - (i) a reference includes Affiliates of Capital One where expressly so provided;
- (ii) references to Capital One in the following definitions include Affiliates of Capital One (unless expressly provided to the contrary): Capital One Data, Capital One Confidential Information, Capital One [\*\*\*], Customer Information, and Capital One Materials;
- (iii) references to sale, assignment, grant or the like by Capital One means Capital One will perform the act for itself or cause Affiliates of Capital One to perform the act themselves;
  - (iv) references to assets being in the name of Capital One include Affiliates of Capital One; and
- (v) references to the business, operations, policies, procedures and the like of Capital One include Affiliates of Capital One to the extent such Affiliates are receiving the Services.
- (b) As used in this Agreement, references to "TSYS" include Affiliates of TSYS where expressly so provided. References to TSYS in the definitions of TSYS Confidential Information and TSYS Materials include Affiliates of TSYS (unless expressly provided to the contrary).
- (c) References to a "party" means Capital One, on the one hand, and TSYS, on the other hand. References to "the parties" means Capital One and TSYS. References to third parties means individuals or entities other than Capital One, TSYS, Permitted Subcontractors, and any of their Affiliates. References to an entity include any branches of such entity in jurisdictions other than the jurisdiction of its incorporation.

## 2.4 Rules of Interpretation

- (a) Unless the context otherwise requires, words importing the singular include the plural and vice-versa, and words importing gender include both genders. Unless the context otherwise requires, references to "persons" includes individual natural persons and juridical legal entities.
- (b) References to articles, sections, and paragraphs shall be references to articles, sections, and paragraphs of the document in which they are contained (i.e., the Main Body of this Agreement, a Schedule, or an Exhibit), unless otherwise specifically stated.
- (c) The section headings in this Agreement are intended to be for reference purposes only and shall not be construed to modify or restrict any of the terms or provisions of this Agreement.

- (d) Where there is similar, but not identical, construction of phrases, sentences, or clauses of this Agreement no implication is made that a difference is intended or not intended, and each shall be construed separately, in accordance with its plain meaning.
- (e) The words "include", "includes", "including", and "e.g." when following a general statement or term, are not to be construed as limiting the general statement or term to any specific item or matter set forth or to similar items or matters, but rather as permitting the general statement or term to refer also to all other items or matters that could reasonably fall within its broadest scope.
  - (f) The word "may" (unless followed by "not") shall be construed as meaning "shall have the right, but not the obligation, to".
- (g) Any reference to a foreign, federal, state, local, or other statute or law shall be deemed to also refer to all rules and regulations promulgated thereunder, unless the context otherwise requires.
- (h) To the extent the same capitalized terms are defined differently in this Agreement, the [\*\*\*] Agreement, and the TS2 License Agreement, those differences are intentional.

## 3 SERVICES

#### 3.1 Provision of Services

- (a) <u>General</u>. TSYS shall provide the services, functions, and responsibilities described in this Agreement (including Schedule A), as such services, functions, and responsibilities may be supplemented, enhanced, modified, or replaced pursuant to this Agreement (collectively, the "<u>Services</u>"). The Services shall be limited to Capital One Accounts. No "Other Offered Service" listed in Schedule C shall be considered a Service until Capital One has elected to receive the service and a charge therefor has been established pursuant to Section 5.2 of Schedule C.
- (b) <u>Implied Services</u>. Except for Capital One's express obligations under this Agreement, if any services, functions, or responsibilities are required for the performance and provision of the Services by TSYS in accordance with this Agreement, regardless of whether they are specifically described herein, they shall be deemed to be implied by and included within the scope of the Services to be provided by TSYS to the same extent and in the same manner as if specifically described in this Agreement.
- (c) <u>Technology Currency</u>. TSYS will [\*\*\*] the TSYS System so that Capital One will receive the benefits of [\*\*\*] during the Term. Subject to Section [\*\*\*], TSYS shall make the [\*\*\*] to the Equipment and Systems Software elements of [\*\*\*] TS2 as it [\*\*\*] TS2.
- (d) <u>Services Variable In Scope</u>. The process for implementing variations in scope of the Services and the pricing impact of such variations are provided for in Schedules A and C. TSYS shall be responsible for adjusting the resources used to provide the Services to accommodate such variations in such a manner as to comply with all provisions of this Agreement. TSYS shall not be entitled to receive an adjustment to the Charges resulting from such variations except as set forth in Schedule C.

# 3.2 <u>Services Performed by Capital One or Third Parties</u>

- (a) No Exclusivity. This Agreement is non-exclusive and shall not be deemed to be a requirements contract. Subject to Section 3.6 of Schedule C, Capital One shall be free without obligation to TSYS to perform itself (or through an Affiliate) or acquire from third parties services similar or identical to the Services provided under this Agreement. Capital One may undertake or award other contracts for additional work to be performed in connection with, or that is operationally related to, the Services. In each case, TSYS shall cooperate with Capital One, such Affiliates, and its or their third party contractors as is reasonably necessary for Capital One, such Affiliates, and such third party contractors to interface such alternate services and additional work with the Services provided by TSYS. The Charges, if any, for such cooperation shall be determined in accordance with Section 5.2 of Schedule C. This cooperation will include:
  - (i) providing information regarding the Services, TSYS [\*\*\*], and other operating parameters (subject to Section 13);
  - (ii) coordinating the Services with such other activities and work to resolve differences that may arise, so that all work may be promptly and properly performed; and
  - (iii) working in good faith to integrate into [\*\*\*] TS2 components and systems developed independently by Capital One, its Affiliates, or third parties. The parties shall also work in good faith to address any compatibility, security, and Service Level issues arising from such integration. TSYS may withhold its approval of the integration of such components and systems only if such compatibility and security issues have not been adequately resolved. Moreover, if such Service Level issues have not been adequately resolved, TSYS shall be excused for its failure to comply with the Service Level as and to the extent such failure is caused by such integrated components or systems. Without limiting the prior sentences in this Section 3.2(a) (iii), Capital One may elect to obtain the equivalent functionality of [\*\*\*], and [\*\*\*] from third parties and TSYS shall allow these third party applications to run on [\*\*\*] TS2 and provide the assistance described in this Section 3.2(a).
- (b) <u>Third Party Compliance</u>. Third parties retained by Capital One and interfacing with TSYS under Section 3.2(a) above shall comply with TSYS' reasonable security and confidentiality requirements and with TSYS' reasonable work standards, methodologies, and procedures provided by TSYS. TSYS shall promptly notify Capital One if TSYS anticipates that an act or omission of such a third party is likely to cause a problem or delay in providing the Services and shall work with Capital One to prevent or circumvent such problem or delay.

# 3.3 Permitted Users and Capital One Accounts

(a) <u>Permitted Users</u>. The Services may be used with respect to Capital One Accounts by (i) Capital One, (ii) (as directed by Capital One) Capital One Affiliates, and (iii) any third party (such as a customer or service provider) permitted by Capital One who must have access to all or the portion of the Services designated by Capital One in order for the full functionality of such portions of the Services to be available with respect to the Capital One Accounts (collectively, those permitted by (i), (ii), and (iii), "<u>Permitted Users</u>"). Services provided to Permitted Users shall be deemed to be Services provided to Capital One. Capital One shall be responsible to TSYS for any breach of this Agreement caused by a Permitted User authorized by Capital One to use the Services under this Agreement.

- (b) <u>Definition of Capital One Accounts</u>. "Capital One Accounts" are those Accounts which are:
  - (i) identified by the BIN/ISOs provided in Exhibit A-1 to Schedule A; and/or
  - (ii) designated by Capital One that fall within one or more of the following categories:
    - (A) any Account [\*\*\*] of Capital One or a Capital One Affiliate;
    - (B) any Account that [\*\*\*] by Capital One or a Capital One Affiliate;
    - (C) any Account [\*\*\*] Capital One or a Capital One Affiliate;
    - (D) any Account for which Capital One or a Capital One Affiliate [\*\*\*] related to at least [\*\*\*] of the following three activities: (1) account origination; (2) portfolio management (e.g., retention programs, fee planning and administration, interest rate, and line of credit adjustments); and (3) collections;
      - (E) subject to Section 3.3(d), any Account acquired by Capital One or a Capital One Affiliate during the Term; and/or
      - (F) any other Account designated by Capital One for which TSYS agrees to provide the Services.
- (c) <u>Confirmation Procedure</u>. Prior to requesting that TSYS provide Services with respect to Accounts of any of the kinds described in Section 3.3(b) (ii)(D), Capital One shall provide to TSYS such information as is reasonably necessary to demonstrate that such Accounts qualify as Capital One Accounts. If TSYS believes an Account for which Capital One requests Services does not qualify as a Capital One Account pursuant to Section 3.3(b), (i) TSYS shall notify Capital One of its objection within [\*\*\*] Business Days of Capital One's written request for such Services and (ii) upon receipt of such notice, Capital One may seek Expedited Dispute Resolution pursuant to Section 18.6. Time is of the essence with respect to such [\*\*\*] Business Day period. Therefore, if TSYS does not object to Capital One's request for Services for a particular Account within [\*\*\*] Business Days after receiving the request, the Account shall be deemed a Capital One Account for the Term of this Agreement.
- (d) [\*\*\*]. If Capital One or a Capital One Affiliate [\*\*\*] or a [\*\*\*], and in connection therewith Capital One or such Capital One Affiliate has [\*\*\*] to (or [\*\*\*] by), [\*\*\*] the obligations under, or [\*\*\*] with [\*\*\*] (an "[\*\*\*] Agreement"), then the [\*\*\*] Agreement shall remain in effect until [\*\*\*] (and any [\*\*\*] provided for therein [\*\*\*]), and at [\*\*\*]:
  - (i) TSYS shall continue to provide services for such Accounts under the terms (including pricing) of the [\*\*\*] Agreement; and/or
  - (ii) provided such Accounts otherwise qualify as Capital One Accounts under Section 3.3(b) above, Capital One [\*\*\*] of this Agreement. In such case [\*\*\*] and shall pay any [\*\*\*] or [\*\*\*] provided for in the [\*\*\*] Agreement until the [\*\*\*] Agreement is terminated in accordance with its terms; provided that [\*\*\*] this [\*\*\*] with respect to any such accounts that [\*\*\*] of this Agreement shall be [\*\*\*] any [\*\*\*] under the [\*\*\*] Agreement.

- (e) [\*\*\*]. If Capital One or a Capital One Affiliate [\*\*\*] of Capital One Accounts ([\*\*\*]) (each, a "[\*\*\*] Account"), then:
- (i) if the [\*\*\*] Account is [\*\*\*], and [\*\*\*], or is [\*\*\*], to [\*\*\*] the [\*\*\*] Account from the TSYS System used to provide Services to Capital One [\*\*\*], the parties shall [\*\*\*] of such [\*\*\*] Accounts to TSYS' processing system for such other TSYS customer; and
- (ii) if the [\*\*\*] Account is [\*\*\*] that is [\*\*\*] or by a [\*\*\*], and is [\*\*\*], to [\*\*\*] Account from the TSYS System used to provide Services to Capital One, TSYS shall [\*\*\*] Account under this Agreement [\*\*\*] TSYS to perform such Services under this Agreement, [\*\*\*] such entity [\*\*\*] or [\*\*\*] Account from the TSYS System used to provide Services to Capital One (in which case the provisions of Section [\*\*\*] shall apply).

## 3.4 Responsibility for Resources

Except for Capital One's express obligations under this Agreement, TSYS shall be responsible for providing all resources (including Equipment and Software) necessary for TSYS to provide the Services and will only recover the costs related thereto through the Charges provided in Schedule C and in the other express payment provisions of this Agreement.

## 3.5 Facilities

- (a) Schedule W lists the TSYS facilities (the "TSYS Facilities") from which TSYS shall perform the Services.
- (b) TSYS shall obtain Capital One's written approval prior to performing the Services from any locations other than the TSYS Facilities. If the proposed location is outside of the United States, Capital One may withhold approval in its sole discretion, and if the proposed location is within the United States Capital One may not unreasonably withhold approval. It shall not be unreasonable for Capital One to withhold its approval if the relocation will result in an increase in Capital One's on-going costs. Notwithstanding the foregoing, in connection with the [\*\*\*] relocation of TSYS' disaster recovery facility as contemplated by Section 7.5(h)(v) of Schedule A, [\*\*\*] shall be responsible for [\*\*\*] one-time out-of-pocket costs (such as expenses for [\*\*\*], but only with respect to [\*\*\*], but [\*\*\*] shall be responsible for any one-time costs associated with [\*\*\*] of [\*\*\*] and all increased [\*\*\*] costs (such as increased [\*\*\*] expense). In addition, any use of or reliance on any resources other than facilities (e.g., personnel) outside of the United States by TSYS shall require Capital One's specific prior written approval, which approval [\*\*\*] in Capital One's [\*\*\*], and which approval Capital One hereby grants in the case of (i) programming personnel located in Canada and (ii) during the Initial Conversion and with respect to the [\*\*\*] Software only, personnel located in [\*\*\*] performing [\*\*\*], that have satisfied the Screening Requirements provided in Section 5.6(c).

## 3.6 Transition of Functions/Capabilities to [\*\*\*] TS2

At Capital One's expense (both one time and on-going) and request at any time during the Term, TSYS shall on a mutually agreed schedule transfer functions/capabilities then provided from the [\*\*\*] TS2, such as [\*\*\*], to [\*\*\*] TS2.

## 3.7 Special Terms and Conditions for [\*\*\*] and [\*\*\*]

(a) If it is established through the dispute resolution process in Section 18.6 that (1) TSYS materially misrepresented the [\*\*\*] of [\*\*\*], [\*\*\*], or [\*\*\*] or materially

misrepresented the [\*\*\*] required in [\*\*\*], or [\*\*\*] to meet the Capital One Initial Conversion Requirements or (2) TSYS cannot prior to [\*\*\*] either (x) complete the development required in order for [\*\*\*], and [\*\*\*] to meet the Specifications for the Initial Conversion and to function in an integrated manner within the TSYS System or (y) with respect to [\*\*\*], meet the Acceptance Criteria provided in Section 3.5(b)(ii) of Schedule A (which determination in any of the above cases the parties acknowledge may be made before [\*\*\*]), then TSYS shall either ([\*\*\*] subject to Section 3.7(c) below):

- (i) (A) allow Capital One to terminate the [\*\*\*] Agreement, the [\*\*\*] Addendum and Service, or the [\*\*\*] Service under this Agreement, as applicable, without payment of a termination charge or other liability; (B) negotiate [\*\*\*] Target Conversion Date for the Initial Conversion reasonably acceptable to both parties that allows Capital One time to obtain and implement a solution to replace the [\*\*\*], or [\*\*\*], as applicable; and (C) [\*\*\*] for (1) all one-time [\*\*\*] by [\*\*\*] in obtaining the [\*\*\*] that [\*\*\*], or [\*\*\*] (but not any additional [\*\*\*]) and that are above the [\*\*\*] by Capital One under the [\*\*\*] Agreement and Schedule C of this Agreement for the original [\*\*\*]; or
- (ii) allow Capital One to terminate this Agreement, the [\*\*\*] Addendum and Service, the [\*\*\*] Service, and/or the [\*\*\*] Agreement, each in whole, without payment of a termination charge or other liability.
- (b) If TSYS elects to offer the remedy in Section 3.7(a)(i) above to Capital One, Capital One shall use Commercially Reasonable Efforts to pursue an arrangement on competitive market terms with a third party for [\*\*\*] to replace [\*\*\*], or [\*\*\*], as applicable.
- (c) If TSYS elects to offer the remedy in Section 3.7(a)(i) above and (i) following a good faith negotiation the parties are unable to agree upon a new Target Conversion Date for the Initial Conversion and (ii) the last date proposed by TSYS during such negotiation is later than the date proposed by Capital One, then Capital One may elect to pursue the remedy in either 3.7(a)(i) (and agree to the last date proposed by TSYS) or 3.7(a)(ii) above.
- (d) If Capital One elects to pursue the remedy in either 3.7(a)(i) or 3.7(a)(ii) above, such remedy shall be Capital One's sole and exclusive remedy with respect to the events giving rise to such remedy under Section 3.7(a). As an alternative to pursuing such remedies, Capital One may elect to continue with the Initial Conversion, and if the conditions provided in Section 2.2 of Schedule C are met, to [\*\*\*] specified in such Section.

## 3.8 OSFI Access to Data and Information

With respect to Capital One Accounts for Canadians, except if Capital One has given Capital One Directions to the contrary, in order to enable Capital One and Capital One Affiliates to comply with the Section 597(1)(c) of the Canadian Bank Act, S.C. 1991, c. 46, (i) each day TSYS shall transmit to Capital One sufficient records for Capital One to maintain in Canada records showing for each such Capital One Account particulars of the transactions with that account holder and the balance owing to or by Capital One's branch (or the Capital One Affiliate in Canada, as the case may be) in respect of such account holder and (ii) TSYS shall maintain the capability to isolate the information of Capital One's Canadian branch (or the Capital One Affiliate in Canada, as the case may be) without undue difficulty.

#### 4 TERM

#### 4.1 Initial Term

The term of this Agreement shall begin on the Effective Date and shall continue for five (5) years from the Commencement Date, unless terminated earlier or extended in accordance with the provisions of this Agreement (the "Initial Term").

#### 4.2 Renewal

Upon expiration of the Initial Term, this Agreement shall automatically renew for [\*\*\*] number of successive [\*\*\*] year periods (each, a "Renewal Term") unless Capital One notifies TSYS in writing at least [\*\*\*] days prior to the expiration of the Initial Term or the then-current Renewal Term of Capital One's desire to terminate this Agreement. This Section 4.2 does not provide TSYS with the right to terminate (or object to the renewal of) this Agreement at the end of the Initial Term or any Renewal Term. TSYS' only rights to terminate are as provided in Section 19.1(b).

## 5 TSYS PERSONNEL

## 5.1 Key TSYS Positions

- (a) <u>General</u>. Schedule D lists certain TSYS positions (the "<u>Key TSYS Positions</u>") and (to the extent determined as of the Effective Date) the individuals filling such positions as of the Effective Date. TSYS shall cause each individual filling a Key TSYS Position to devote substantially full time and effort to the provision of the Services for at least twelve (12) months from the date she or he is assigned to the Key TSYS Position unless such individual:
  - (i) leaves TSYS' employment;
  - (ii) undertakes a long term leave of absence; or
  - (iii) without encouragement from TSYS notifies TSYS that he or she is unwilling to continue in TSYS' employment unless removed from the Capital One Account.
- (b) <u>Changes In Designation</u>. After consultation with TSYS, Capital One may from time to time change the positions that are designated as Key TSYS Positions. Without TSYS' consent, the number of Key TSYS Positions may not exceed the number of positions initially specified in Schedule D as of the Effective Date.
  - (c) Review. Before assigning an individual to a Key TSYS Position, whether as an initial assignment or a subsequent assignment, TSYS shall:
    - (i) notify Capital One of the proposed assignment;
    - (ii) introduce the individual to appropriate Capital One representatives;
    - (iii) provide such representatives upon request with the opportunity to interview the individual; and
    - (iv) provide Capital One with a resume and other information about the individual reasonably requested by Capital One.

If Capital One objects in good faith, and for reasons which do not impermissibly relate to any individual's status as a member of a "protected class" under applicable employment law, to the

proposed assignment, the parties shall attempt to resolve Capital One's concerns on a mutually agreeable basis. If the parties have not been able to resolve Capital One's concerns within [\*\*\*] Business Days, TSYS shall not assign the individual to that position and shall propose to Capital One the assignment of a different individual of suitable ability and qualifications. Except with Capital One's prior written consent (which may be arbitrarily withheld) (or the condition in Section 5.1(a)(iii) applies) individuals filling Key TSYS Positions may not be transferred or re-assigned by TSYS to other positions with TSYS or its Affiliates until (A) suitable replacements have been approved by Capital One and (B) the replacements have been trained to perform substantially all of their respective responsibilities in such Key TSYS Positions, and no such transfer shall occur at a time or in a manner that would have an adverse impact on delivery of the Services (other than an insubstantial impact).

## 5.2 TSYS Services Manager

- (a) <u>General</u>. One of the Key TSYS Positions shall be for an individual whose responsibility is to serve as the TSYS services manager for the Capital One account ("<u>TSYS Services Manager</u>"). Unless agreed by Capital One, the TSYS Services Manager will never be more than [\*\*\*] organizational reporting levels removed from TSYS' [\*\*\*] with overall responsibility for performance of the Services.
  - (b) <u>Responsibilities</u>. The TSYS Services Manager (in conjunction with the applicable TSYS Conversion personnel during a Conversion) shall:
  - (i) on a regular basis (and at least weekly prior to and during each Conversion) report to the Capital One Contract Manager as to whether the work is proceeding in accordance with the requirements of this Agreement and the applicable Conversion Plan, and if not, what steps are being taken (or should be taken) by either TSYS or Capital One to restore the Conversion or other work to such status as soon as possible;
  - (ii) serve as the TSYS principal point of contact regarding the Services for Capital One and otherwise be responsible for all Services-related communications on behalf of TSYS;
    - (iii) oversee the management of performance of TSYS' obligations under this Agreement;
  - (iv) meet with the Capital One Contract Manager regularly, and at the Capital One Contract Manager's reasonable request on an ad hoc basis, over the course of the Term to discuss the status and quality of the Services;
    - (v) oversee the day-to-day activities of any Services staff dedicated to Capital One;
    - (vi) serve as the point of escalation for any issues or questions related to the Services;
  - (vii) ensure Capital One is briefed in accordance with Section 9.2(a)(ii)(A) about each anticipated material change to the technology or functionality contained in TS2, [\*\*\*], or other components of the TSYS System;
    - (viii) monitor status of all development projects; and

- (ix) monitor status of all issues and monitor the other support provided to Capital One under this Agreement.
- (c) [\*\*\*]. The TSYS [\*\*\*] shall include the opportunity to [\*\*\*] that are [\*\*\*] based [\*\*\*] on [\*\*\*] the [\*\*\*] under this Agreement and the [\*\*\*] Agreement.

## (d) Location.

- (i) The TSYS Services Manager shall be located in the Columbus, Georgia area but he or she shall travel to Capital One's offices in the Richmond, Virginia area on Capital One's reasonable request for meetings and the like.
- (ii) If the parties agree to [\*\*\*] any of the TSYS Personnel in the Development Pool [\*\*\*], and such personnel are [\*\*\*], TSYS shall [\*\*\*] Capital One for incremental [\*\*\*] TSYS may [\*\*\*] as a result of [\*\*\*] up to [\*\*\*] ([\*\*\*]) such TSYS Personnel [\*\*\*].

## 5.3 Resource Pool

- (a) Subject to the adjustment provided in Section 5.3(b), TSYS shall provide a pool of at least [\*\*\*] TSYS Personnel dedicated to provide Services to Capital One for the Non-Development Pool and [\*\*\*] TSYS Personnel dedicated to Capital One for the Development Pool (the Non-Development Pool and the Development Pool are collectively the "Resource Pool"). Except as permitted by Section 5.3(b) below, Section 3.6(c) of Schedule C, and Section 4.5 of Schedule A, TSYS may not [\*\*\*] of TSYS Personnel in the Resource Pool without Capital One's prior written consent, which Capital One may withhold in its sole discretion. Except for [\*\*\*] under Section [\*\*\*] and Section [\*\*\*] of Schedule C, if Capital One agrees to have the Resource Pool [\*\*\*] the parties shall agree upon an equitable [\*\*\*] the Charges to reflect the [\*\*\*] will derive from the [\*\*\*] personnel.
- (b) The Non-Development Pool shall be increased or decreased every calendar quarter beginning in the calendar quarter [\*\*\*] after the Commencement Date based upon the [\*\*\*] number of Active Accounts in the prior calendar quarter in accordance with the following table (and no adjustment to the Development Pool shall be required if an adjustment in the Non-Development Pool pursuant to this Section 5.3(b) causes the total number of TSYS Personnel in the Resource Pool to be less than [\*\*\*]):

Monthly Active Accounts	Dedicated Support Staff (FTE)
Less than [***]	[***]
[***] or greater	[***]

## 5.4 Qualifications and Retention of TSYS Personnel

- (a) TSYS shall assign an adequate number of TSYS Personnel to perform the Services. TSYS Personnel shall be properly educated, trained, and qualified for the Services they are to perform. Unless otherwise approved, at least [\*\*\*] of the TSYS members of the Resource Pool shall have either:
  - (i) except as provided otherwise in Exhibit A-7 for Development Pool Personnel, a [\*\*\*] year college degree or equivalent bankcard business experience; or
    - (ii) at least [\*\*\*] or more years of appropriate experience as a TSYS employee.
- (b) TSYS shall provide TSYS Personnel continuing education and training throughout the Term at least comparable to the continuing education and training provided to TSYS employees performing similar duties for TSYS. TSYS [\*\*\*] TSYS Personnel to become familiar with Capital One's account, business, or requirements.
- (c) The parties agree that it is in their best interests to keep the turnover rate of TSYS Personnel assigned to Capital One's account (or otherwise providing the Services) to a reasonably low level. Accordingly, if Capital One believes that TSYS' turnover rate has been excessive and so notifies TSYS, TSYS shall provide data concerning its turnover rate and meet with Capital One to discuss the reasons for, and impact of, the turnover rate. TSYS shall use Commercially Reasonable Efforts to keep the turnover rate to a reasonably low level. Notwithstanding transfer or turnover of TSYS Personnel, TSYS remains obligated to perform the Services in accordance with this Agreement.

## 5.5 Removal of TSYS Personnel

If Capital One determines in good faith, and for reasons which do not impermissibly relate to any individual's status as a member of a "protected class" under applicable employment law, that the continued assignment to Capital One's account of one or more of the TSYS Personnel is not in the best interests of Capital One, then Capital One may give TSYS notice to that effect. After receipt of such notice, TSYS shall have a reasonable period of time in which to discuss Capital One's concerns with Capital One. If following such period Capital One continues to request replacement of any such person, TSYS shall replace that person with another person of suitable ability and qualifications. However, where Capital One initially notifies TSYS that Capital One has determined that the nature of the concern is such that such TSYS Personnel should be removed immediately (albeit temporarily) from Capital One's account, TSYS shall immediately remove such individual(s) from Capital One's account. Any request by Capital One to remove an individual from Capital One's account shall not be deemed to constitute a termination of such individual's employment by TSYS and in no event shall Capital One be deemed an employer of any such person. This provision shall not operate or be construed to limit TSYS' responsibility for the acts or omissions of TSYS Personnel.

# 5.6 Compliance with Each Party's Regulations and Policies

- (a) <u>General Compliance</u>. While on the premises of the other party, each party's personnel shall comply with the other party's directions, policies, rules, and regulations that have been provided to the other party in writing regarding:
  - (i) personal and professional conduct (including the wearing of an identification badge and adhering to regulations and general safety, dress, behavior, and security practices or procedures) generally applicable to such premises;

- (ii) protection of Confidential Information;
- (iii) security and network access; and
- (iv) all other rules and regulations generally applicable to the other party's employees, contractors, and invitees on such premises.
- (b) <u>Communication of Obligations</u>. Each party shall regularly brief all its applicable personnel on the scope of restrictions on it and its personnel concerning the safeguarding of Confidential Information. In addition, TSYS shall make TSYS Personnel aware of Applicable Laws against improper disclosure or use of Customer Information.

## (c) Personnel Screening.

- (i) TSYS shall apply [\*\*\*] and [\*\*\*] screening requirements ("<u>Screening Requirements</u>") to all TSYS Personnel to whom TSYS grants access to (A) Customer Information, (B) systems that they could use to modify, change, or manipulate Customer Information, or (C) Capital One's premises. The Screening Requirements existing as of the Effective Date are set forth in Schedule V of this Agreement. The parties may revise the Screening Requirements from time to time during the Term as provided in Section 5.6(c)(iii).
- (ii) Capital One shall have the right to validate TSYS' compliance with the Screening Requirements on an [\*\*\*] by requiring TSYS to make available to Capital One at TSYS' premises the following information for all TSYS Personnel whom TSYS is required to screen using the Screening Criteria: (A) employee identification numbers, (B) the results of any background checks required by the Screening Requirements, (C) a list of criteria used by TSYS to assess background checks that are found to contain [\*\*\*], which is set forth in Schedule V of this Agreement, and (D) any other information reasonably necessary for Capital One to validate that all TSYS Personnel whom TSYS is required to screen using the Screening Criteria were screened in compliance with the Screening Requirements. In addition, Capital One may conduct follow-up validations at TSYS' premises as is reasonably necessary to ensure any issues or problems identified during the [\*\*\*] validation are corrected.
- (iii) If, during the Term, Capital One revises its [\*\*\*] and [\*\*\*] screening requirements for Capital One employees or third party service providers in order to be in compliance with any changes in Applicable Laws effected after the Effective Date or to remain current with common practice in the card processing industry, then Capital One shall provide written notice to TSYS of such changes, which notice shall be accompanied by documentation evidencing the changes in Applicable Law or in the card processing industry that have necessitated Capital One's changes to its internal personnel background and security screening requirements. Provided TSYS is able to validate that Capital One was [\*\*\*], to implement such changes, or that such changes are [\*\*\*] the card processing industry, TSYS shall revise the Screening Requirements accordingly and shall thereafter comply (to the extent it may legally and practically do so) with the Screening Requirements as revised at [\*\*\*] Capital One. If Capital One [\*\*\*] and [\*\*\*] screening requirements for reasons not governed by the foregoing, TSYS shall consider in good faith [\*\*\*] the Screening Requirements to [\*\*\*], and any Out-of-Pocket Expenses associated with TSYS' [\*\*\*] shall be [\*\*\*].

(iv) Capital One may, at its own expense, [\*\*\*] and conduct [\*\*\*] screening of TSYS Personnel, to the extent permitted by Applicable Law and required under policies and procedures generally applicable to Capital One employees or third party service providers. Such screening is in addition to, and not a substitute for, the screening TSYS is required to perform under this Section 5.6(c). Capital One may not obtain [\*\*\*] of TSYS Personnel unless the Federal Bureau of Investigation or other third party has agreed to perform background checks for Capital One using [\*\*\*].

## 5.7 <u>Use of Subcontractors and Other Support</u>

- (a) TSYS shall remain responsible for obligations, services, and functions performed by its subcontractors to the same extent as if such obligations, services, and functions were to be performed by TSYS and for purposes of this Agreement such work shall be deemed work performed by TSYS' own employees. TSYS shall be Capital One's sole point of contact regarding the Services, including with respect to payment. TSYS shall not disclose Capital One Confidential Information to a subcontractor unless and until such subcontractor has agreed in writing to protect the confidentiality of such Confidential Information in a manner substantially equivalent to that required of TSYS under this Agreement.
- (b) Subject to Section 5.7(c), except as and to the extent Capital One may agree otherwise in writing, TSYS may subcontract performance of the Services only in accordance with the following:
  - (i) The TSYS Affiliates and other subcontractors listed on Schedule G are hereby approved for the scope of the efforts described.
  - (ii) Prior to entering into any subcontract with a potential subcontractor (including a TSYS Affiliate) not listed on Schedule G or for additional scope beyond that described in Schedule G, TSYS shall give Capital One reasonable prior notice specifying (A) the components of the Services affected, (B) the [\*\*\*] of the proposed subcontract, (C) Capital One's audit rights with respect to such subcontractor, (D) the amount and type of insurance carried by the subcontractor, and (E) the [\*\*\*] and [\*\*\*] of the proposed subcontractor. At Capital One's request, TSYS shall forward to Capital One a description of the [\*\*\*] and material terms (other than [\*\*\*], except if it directly affects the [\*\*\*]) of the subcontract or proposed subcontract. [\*\*\*].
  - (iii) Capital One may revoke its approval of a subcontractor and direct TSYS to replace such subcontractor if (A) the subcontractor's performance is materially deficient; (B) good faith doubts exist concerning the subcontractor's ability to render future performance because of changes in the subcontractor's ownership, management, financial condition, or otherwise; (C) the subcontractor violates any of the terms and conditions of this Agreement; (D) [\*\*\*] that [\*\*\*] of the [\*\*\*] or [\*\*\*] the [\*\*\*] to [\*\*\*]; or (E) there have been material misrepresentations by or concerning the subcontractor. Approval of a subcontractor that at the time of approval is an Affiliate of TSYS shall be automatically revoked if such subcontractor ceases to be such an Affiliate. However, if Capital One removes the subcontractor providing a New Additional Service for any of the reasons provided in the first sentence above and such subcontractor was designated by Capital One, then such New Additional Service shall not be included for purposes of [\*\*\*] under Schedule C, Capital One shall bear the cost of replacing the subcontractor, TSYS shall not be obligated to provide the New Additional Service

until the subcontractor is replaced, and TSYS shall not be liable for any [\*\*\*] from the disruption caused by the replacement.

(c) TSYS may in the ordinary course of business subcontract for third party services or products that are not dedicated to Capital One, that are not material to a particular function constituting a part of the Services, and that do not result in a material change in the way TSYS conducts its business, provided such subcontract does not adversely affect Capital One, whether in performance of or Charges for the Services or otherwise. If Capital One expresses concerns to TSYS about a subcontract covered by this Section 5.7(c), TSYS shall discuss such concerns with Capital One and work in good faith to resolve Capital One's concerns on a mutually acceptable basis.

#### 5.8 Oversight Controls

TSYS shall implement oversight and monitoring procedures reasonably designed to detect, prevent, investigate, and report to Capital One any misuse or misappropriation of Capital One assets, systems, funds, or information by TSYS Personnel. TSYS shall promptly report to Capital One any such misuse or misappropriation identified or reasonably suspected.

## 6 COMPLIANCE WITH LAWS

## 6.1 Compliance with Laws

(a) <u>Mutual Responsibilities</u>. Each party will comply with all Applicable Laws that apply to such party (including, in the case of TSYS, those applicable to it as a third party provider of processing service and other services similar to the Services), including identifying and procuring required permits, certificates, approvals and inspections. If a party receives formal notice from the government or written notice from any other entity of a charge of non-compliance of that party with any Applicable Law in any way related to its performance of this Agreement, that party will promptly notify the other party of such charge in writing.

## (b) Capital One Legal Requirements and Directions.

- (i) Subject to the provisions of this Section 6, and without limiting TSYS' obligations under Section 13, Capital One shall be responsible for compliance with all Applicable Laws applicable to the Accounts authorized by Capital One to receive the Services (the "<u>Capital One Legal Requirements</u>").
- (ii) Capital One shall be responsible for selecting the parameter settings and programming features and options within the TSYS System that will apply to the Accounts receiving the Services and, subject to TSYS' compliance with its obligations under this Section 6.1, for determining that such selections are consistent with the Capital One Legal Requirements and with the terms and conditions of the various agreements between Capital One and its customers/cardholders. In making such determinations, Capital One may rely on the accuracy and completeness of the descriptions of such settings, features, and options provided by TSYS, as well as TSYS' compliance with and performance of its obligations under this Section 6. TSYS shall review with and explain to Capital One the TSYS System parameter settings, features, functionality, and options and shall provide clarification on such as is requested by Capital One.
- (iii) Capital One shall from time to time provide directions to TSYS, either in writing or by such methodology as is provided by the Procedures Manual or the

Documentation (e.g., by email or on-line database and parameter updates), based on Capital One's interpretation of the Capital One Legal Requirements and Capital One's selection of parameter settings and programming features and options (the "Capital One Directions").

## (c) TSYS Compliance. TSYS shall:

- (i) perform the Services in a manner that complies with the Capital One Directions; and
- (ii) not make any changes which would cause the Services to be performed other than in accordance with the Capital One Directions; provided that in the event of conflicts among Capital One Directions, the more recent Capital One Directions govern.
- (d) <u>TSYS Obligation To Inform Capital One</u>. Notwithstanding that Capital One is responsible for ensuring that the Capital One Directions are sufficient to enable Capital One to be [\*\*\*], TSYS shall promptly notify Capital One if at any time TSYS believes in good faith that based on the actual knowledge (without duty of inquiry) of the involved individuals that the Capital One Directions may [\*\*\*]. TSYS shall also inform Capital One, and seek clarification from Capital One, if the Capital One Directions are [\*\*\*]. TSYS shall have no obligation to detect or inform Capital One of conflicts among Capital One Directions. Whenever relevant information is known by TSYS, subject to TSYS' confidentiality obligations to third parties, TSYS shall share with Capital One and its Affiliates information concerning the various regulatory compliance practices and interpretive positions taken by TSYS with respect to the services TSYS provides to its customers generally related to or in connection with the use of the Services and any functionality available on [\*\*\*] TS2.
- (e) <u>Compliance Enhancements</u>. TSYS shall maintain the TSYS System so that it enables TSYS to comply with the Capital One Directions, as they may change from time to time during the Term, including by adding functionality and making the necessary modifications and enhancements to the TSYS System (collectively, "<u>Compliance Enhancements</u>"). Such Compliance Enhancements shall be implemented in accordance with the Development Methodology. The Charges, if any, for such Compliance Enhancements are set forth in Schedule C.
- (f) Quality Checks. TSYS will monitor its compliance with the requirements of this Section 6 and shall notify Capital One if it [\*\*\*] of this Section 6.

## 6.2 Gramm-Leach-Bliley Act, and Similar Laws

Each party shall comply with all Applicable Laws that apply to it protecting the confidential material and privacy rights of the parties, their Affiliates, and/or their customers and consumers, including Title V, Subtitle A of the GLB Act, the U.S. Economic Espionage Act (18 USC §1831 et seq.), and Canadian Privacy Legislation and the Canadian Security Information Act (R.S. 1985, c. O-5).

## 6.3 Bank Bribery Act and Foreign Corrupt Practices Act

Each party shall comply with the (a) Federal Bank Bribery Act of 1984 (12 USC §215) and (b) Foreign Corrupt Practices Act of 1977 (15 USC §§ 78dd-1, et seq.), to the extent applicable to that party.

#### 6.4 Consumer Credit Protection Laws

Each party shall comply with Applicable Laws that apply to it governing collection of debt and consumer credit reporting, including the Fair Debt Collection Practices Act (FDCPA) (16 USC §1601 et seq.), the Fair Credit Reporting Act (FCRA) (15 USC §1681 et seq.), the Equal Credit Opportunity Act (16 USC §1691 et seq.), the Telephone Consumer Protection Act, (47 USC §227 et seq.), the Telephone Sales Rule (16 CFR §310.1 et seq.), the Canadian Financial Consumer Agency of Canada Act (S.C. 2001), c.9), Section 52.1 of the Canadian Competition Act (R.S. 1985, c. C-34), applicable telemarketing rules established by the Canadian Radio-Television and Telecommunications Commission, the Truth in Lending Act (Pub. L. 90-321, Title I, May 29, 1968, 82 Stat. 146)1) ("TILA") and any federal, provincial, state, or local equivalents of the foregoing.

## 6.5 Office of Foreign Assets Control Regulations

Each party shall comply, to the extent applicable to it, with the economic and trade sanctions administered by the Office of Foreign Assets Control ("OFAC") of the U.S. Department of the Treasury and OSFI, including all Executive Orders and implementing regulations. In the case of TSYS, this obligation includes not engaging to perform the Services any persons or businesses on (a) OFAC's Specifically Designated Nationals and Blocked Persons List or (b) OSFI's Consolidated List, to perform the Services.

# 6.6 USA PATRIOT Act and Proceeds of Crime (Money Laundering) and Terrorist Financing Act in Canada

Each party shall comply, to the extent applicable to it, with the anti-money laundering requirements and other Applicable Laws designed to combat terrorist financing, including those requirements contained in the Bank Secrecy Act and the USA PATRIOT Act of 2001 (Pub L. 107-56) in the United States and the Proceeds of Crime (Money Laundering) and Terrorist Financing Act in Canada, and all implementing regulations. TSYS will ensure all appropriate personnel are adequately trained on anti-money laundering procedures, with refresher training conducted no less than annually. TSYS will additionally ensure that it has appropriately identified and verified those persons performing Services such that it has a sufficiently reasonable belief as to the identity of those persons to be in compliance with such requirements.

## 7 SERVICE LEVELS

In performing the Services TSYS shall meet or exceed the [\*\*\*] service levels provided in Schedule B ("Service Levels"). Schedule B also describes the tools and processes TSYS shall use to measure and report on its performance against the Service Levels, and provides a methodology for [\*\*\*] by [\*\*\*] for [\*\*\*] meet certain Service Levels designated therein. If Capital One elects to [\*\*\*] a [\*\*\*] in respect of a Service Level [\*\*\*] and does not revoke such election within [\*\*\*] days as required by Section 4.3(c) of Schedule B, such [\*\*\*] shall be Capital One's [\*\*\*] for such [\*\*\*]. To the extent such [\*\*\*] is a breach of this Agreement Capital One shall retain the rights provided in Section 19.1(a) with respect to such breach regardless of whether or not it elects to receive a corresponding [\*\*\*]. If Capital One revokes its election pursuant to the prior sentence, it shall [\*\*\*] to TSYS.

## 8 PROPRIETARY RIGHTS

This Section addresses the parties' respective rights in Capital One Materials, TSYS Materials, and Developed Materials.

#### 8.1 Capital One Materials

- (a) General. As between the parties, subject to Section 8.1(b), Capital One retains all right, title, and interest in and to Capital One Materials.
- (b) <u>Grant of Rights</u>. Capital One grants to TSYS a [\*\*\*] license to Use (and to allow its Permitted Subcontractors to Use) Capital One Materials and other Capital One Confidential Information during the Term solely to the extent necessary for performing the Services. Capital One Materials will be made available to TSYS in such form and on such media as exists on the Effective Date or as is later obtained by Capital One, together with available documentation and any other related materials.
- (c) <u>Use Restrictions and Compliance</u>. TSYS may not Use Capital One Materials for the benefit of any entities other than Capital One and its Affiliates without the prior written consent of Capital One, which may be withheld at Capital One's discretion. TSYS shall comply with any written directions furnished by Capital One regarding restrictions on the Use or disclosure of Capital One Materials that consist of Third Party Materials, as may be necessary to comply with Capital One's contractual obligations with respect to such Third Party Materials.
- (d) <u>Cessation of Use Upon Expiration or Termination of this Agreement</u>. Except as otherwise requested or approved by Capital One, TSYS shall cease all use of Capital One Materials upon expiration or termination of this Agreement except to the extent such materials are required to provide Termination/Expiration Assistance.

## 8.2 <u>Developed Materials</u>

- (a) Use Restrictions and Compliance.
- (i) Subject to Section 8.2(a)(ii) and 8.2(a)(iii) below, TSYS may not Use [\*\*\*] List Materials or any other [\*\*\*] for the benefit of [\*\*\*] Capital One and its Affiliates without [\*\*\*], which [\*\*\*].
  - (ii) TSYS may Use for itself or other customers TSYS Developed Materials that are:
    - (A) not [\*\*\*] List Materials;
    - (B) developed [\*\*\*] or [\*\*\*] the [\*\*\*] of the Development Methodology; and
  - (C) either developed [\*\*\*], or developed [\*\*\*] the Commencement Date and documented in Schedule L (which shall be updated by the parties to document any such [\*\*\*] that are [\*\*\*] the parties before [\*\*\*] and that are not listed on Schedule L at the Effective Date) between the [\*\*\*] the TSYS System as of the Effective Date and the [\*\*\*] by Capital One,

(TSYS Developed Materials that meet [\*\*\*] of the criteria in (A), (B), and (C), are referred to as "[\*\*\*] Materials").

(iii) TSYS may Use TSYS Developed Materials that are part of the [\*\*\*] without restriction other than as provided in Sections 8.2(c), 8.11(b), and 8.11(c) below.

- (iv) This Agreement does not restrict TSYS from Using for itself and/or other customers any TSYS Materials developed by or on behalf of TSYS or any of its Affiliates outside the scope of this Agreement (including [\*\*\*] materials that [\*\*\*] or [\*\*\*] the [\*\*\*] for [\*\*\*] List Materials).
- (b) <u>Cessation of Use Upon Expiration or Termination of this Agreement</u>. Except as otherwise requested or approved by Capital One, TSYS shall cease all Use of [\*\*\*], other than the [\*\*\*] Date Materials, upon expiration or termination of this Agreement except to the extent such materials are required to provide Termination/Expiration Assistance.
- (c) <u>Exclusivity</u>. The following limitations on TSYS' development of certain [\*\*\*] are in addition to the restrictions on use of TSYS Developed Materials provided above:

#### (i) Pre-Commencement Date.

- (A) Subject to Section 8.2(c)(i)(B) below, for [\*\*\*] months after the Commencement Date, TSYS and its [\*\*\*] to an item on the [\*\*\*] List [\*\*\*] TSYS or its Affiliates.
- (B) During the period provided in Section 8.2(c)(i)(A), TSYS may [\*\*\*] (or, in the case of a [\*\*\*] that is part of the [\*\*\*], [\*\*\*]) [\*\*\*] an item on the [\*\*\*] List [\*\*\*] ([\*\*\*] by TSYS [\*\*\*]), regardless of the format of the [\*\*\*]; provided that (1) if TSYS provides [\*\*\*], TSYS may follow its normal practice of generally [\*\*\*] to suggest [\*\*\*] and (2) TSYS [\*\*\*] any Restricted Employees (defined below) to [\*\*\*] until the [\*\*\*] of (x) if the Restricted Employee is not [\*\*\*] the Development Pool, [\*\*\*] months after the Commencement Date and (y) if the Restricted Employee is [\*\*\*] the Development Pool, [\*\*\*] months after the date such individual is [\*\*\*] the Development Pool. "Restricted Employee" shall mean (a) any of an aggregate of [\*\*\*] TSYS Personnel that Capital One may [\*\*\*] (none of which may be [\*\*\*] level or higher and no more than [\*\*\*] of which may be [\*\*\*]) and (b) any TSYS Personnel selected by TSYS to be [\*\*\*] the Development Pool, in each case who were involved in the [\*\*\*] of the [\*\*\*] the applicable item on the [\*\*\*] List.
- (C) After the period provided in Section 8.2(c)(i)(A), this Agreement shall not restrict TSYS from [\*\*\*] (and [\*\*\*] and/or [\*\*\*]) [\*\*\*] an item on the [\*\*\*] List provided it complies with its obligations in Section 13.

## (ii) Post-Commencement Date.

- (A) Subject to Section 8.2(c)(ii)(B) below, for [\*\*\*] months after the [\*\*\*] is first [\*\*\*] for [\*\*\*], TSYS and its Affiliates [\*\*\*] an item on the [\*\*\*] List for any other customer of TSYS or its Affiliates.
- (B) During the period provided in Section 8.2(c)(ii)(A), TSYS may [\*\*\*] (or, in the case of a [\*\*\*] that is part of the [\*\*\*], [\*\*\*]) [\*\*\*] an item on the [\*\*\*] List [\*\*\*] a [\*\*\*] ([\*\*\*] by TSYS [\*\*\*] to [\*\*\*]), regardless of the format of the [\*\*\*]; provided that (1) if TSYS provides [\*\*\*], TSYS may follow its normal practice of generally [\*\*\*] to suggest [\*\*\*] and (2) TSYS [\*\*\*] any individual involved in the [\*\*\*] of such [\*\*\*] for Capital One to [\*\*\*] such [\*\*\*] until after such individual [\*\*\*] the Development Pool.
- (C) After the period provided in Section 8.2(c)(ii)(A), this Agreement shall not restrict TSYS from [\*\*\*] (and [\*\*\*] and/or [\*\*\*]) [\*\*\*] an item on the [\*\*\*] List provided it complies with its obligations in Section 13.

- (iii) <u>Information Bulletins</u>. If pursuant to Section 8.2(c)(i)(B) or 8.2(c)(ii)(B) TSYS is permitted to [\*\*\*] on the [\*\*\*] List or [\*\*\*] List for [\*\*\*], then TSYS may (A) list the [\*\*\*] in a standard TS2 information bulletin, (B) respond to [\*\*\*], and [\*\*\*] the [\*\*\*], any [\*\*\*] such information bulletin and ([\*\*\*] by TSYS) [\*\*\*] such [\*\*\*], and (C) [\*\*\*] the [\*\*\*] for any [\*\*\*] that [\*\*\*]. TSYS may not otherwise [\*\*\*] or [\*\*\*] the [\*\*\*] to or for [\*\*\*] during the [\*\*\*] month periods referenced above.
- (d) Access to [\*\*\*]. Notwithstanding any [\*\*\*] TSYS has with [\*\*\*], if Capital One requests [\*\*\*] contained in [\*\*\*] that TSYS [\*\*\*] for [\*\*\*], TSYS shall [\*\*\*] that [\*\*\*] to Capital One [\*\*\*] (other than [\*\*\*] pursuant to Schedule [\*\*\*] for any [\*\*\*] that may be required to integrate such [\*\*\*] with [\*\*\*] TS2).
- (e) <u>Incorporation of Capital One Confidential Information in TSYS Developed Materials</u>. TSYS shall create and maintain a written record of the TSYS Developed Materials that contain Capital One Confidential Information. The record shall be sufficiently detailed to enable TSYS to identify and remove all Capital One Confidential Information prior to using such TSYS Developed Materials (if and to the extent permitted by this Section 8) for a purpose other than providing Services to Capital One and Capital One's Affiliates.

## 8.3 Process For [\*\*\*] the Lists

- (a) The parties recognize that all key [\*\*\*] Capital One may desire to be [\*\*\*] to the Commencement Date that [\*\*\*] the [\*\*\*] provided by the TSYS System and the [\*\*\*] by Capital One [\*\*\*] as of the Effective Date. Accordingly, Capital One may [\*\*\*], prior to the Commencement Date, up to [\*\*\*] that in Capital One's reasonable judgment should [\*\*\*] the [\*\*\*] the [\*\*\*] on the [\*\*\*] List, provided such [\*\*\*] are: (i) reasonably expected to give Capital One a [\*\*\*]; (ii) not generally [\*\*\*] or [\*\*\*] in the [\*\*\*]; and (iii) not at the time [\*\*\*] (other than due to TSYS' [\*\*\*] for Capital One). Such [\*\*\*] shall be [\*\*\*] the [\*\*\*] List in accordance with the procedure specified in Section 8.3(b) and the restrictions applicable to the related [\*\*\*] List Materials shall apply. The parties acknowledge that items such as Capital One's [\*\*\*] would be considered a [\*\*\*] "[\*\*\*]" for purposes of this Section. The parties acknowledge that items such as Capital One's "[\*\*\*]" would be considered [\*\*\*] "[\*\*\*]" for purposes of this Section.
- (b) Within [\*\*\*] days after Capital One notifies TSYS that it desires to [\*\*\*] an [\*\*\*] to a List (time being of the essence), TSYS may notify Capital One either (i) that Capital One's designation is [\*\*\*] for TSYS to [\*\*\*] (TSYS shall not have [\*\*\*] for items it is [\*\*\*] for Capital One) or (ii) that the [\*\*\*] does not meet the [\*\*\*] for [\*\*\*] on the [\*\*\*] List provided in Section 8.3(a) or in the definition of [\*\*\*] List, as applicable. If TSYS does not so notify Capital One, the item shall be automatically [\*\*\*] to the [\*\*\*] List. If TSYS objects on the basis specified in clause (i) or (ii) of the first sentence of this paragraph, TSYS shall, within a reasonable period of time following delivery of the notice to Capital One, reasonably document the basis for [\*\*\*]. The parties shall meet to discuss TSYS' [\*\*\*] within [\*\*\*] Business Days following delivery of such documentation to Capital One, and if they cannot resolve the dispute, the dispute may be submitted by either party to Expedited Dispute Resolution under Section 18.6.
- (c) TSYS shall provide a copy of its then current [\*\*\*] every [\*\*\*] months (or earlier, within [\*\*\*] days of request, as Capital One may reasonably request). Any objection by TSYS to an [\*\*\*] on a List on the grounds that it is already [\*\*\*] shall only be valid if such

[\*\*\*] was on the last list provided by TSYS or TSYS can demonstrate by business records maintained in the ordinary course was [\*\*\*] after the last list was prepared but prior to the date of Capital One's request.

## 8.4 TSYS Materials

(a) <u>General</u>. As between the parties, TSYS retains all right, title, and interest in and to TSYS Materials, subject to Section 8.4(b), the TS2 License Agreement, and the [\*\*\*] Agreement.

## (b) Grant of Rights.

- (i) TSYS grants to Capital One, during the Term, and for any period during which Capital One receives Termination/Expiration Assistance and the Development Pool is providing development services, a right to allow Development Pool Personnel to Use the TSYS System and TSYS Materials, including all [\*\*\*], Documentation, other materials necessary for the Use thereof, to [\*\*\*] TS2. For the avoidance of doubt, except where authorized by TSYS, personnel who are not members of the Development Pool shall not be permitted [\*\*\*] the [\*\*\*] of the TSYS System.
- (ii) TSYS grants Capital One a [\*\*\*] worldwide license to Use any non-Software TSYS Materials incorporated in Capital One Developed Materials, subject to Capital One's obligations of confidentiality with respect to any TSYS Confidential Information that TSYS identifies prior to incorporation into such Capital One Developed Materials.
- (c) <u>Licensed Materials and [\*\*\*] Software</u>. For purposes of the TS2 [\*\*\*] Agreement, the TSYS Developed Materials relating to [\*\*\*] TS2 shall be deemed part of the Licensed Materials. For purposes of the [\*\*\*] Agreement, the TSYS Developed Materials relating to the [\*\*\*] Software shall be deemed part of the [\*\*\*] Software [\*\*\*].

## 8.5 Third Party Materials

- (a) Representation and Warranty. TSYS represents and warrants, as of the Effective Date, as follows:
- (i) Section 1 of Schedule K lists all Third Party Materials that are incorporated into the TSYS System (as amended from time to time, the "TPM Required to Provide Services") and Section 2 of Schedule K lists all other Third Party Materials required for Capital One and the Permitted Users to receive the Services in the manner contemplated by this Agreement (as amended from time to time, the "TPM Required to Receive Services" and, together with the TPM Required to Provide Services, the "Schedule K Materials"); and
  - (ii) all Third Party Materials listed in Schedule K are commercially available to the general public.
- (b) Allocation of Responsibility for Schedule K Materials.
- (i) Subject to Section 8.5(b)(vii), TSYS may change and add to the Schedule K Materials at any time without Capital One's consent. TSYS shall give Capital One written notice of each such change and addition prior to the effective date of such change or addition (each, a "<u>Schedule K Change Notice</u>").

- (ii) If TSYS [\*\*\*] the TPM Required to Receive Services without [\*\*\*], and such [\*\*\*] and [\*\*\*] in connection with the TPM Required to Receive Services, then TSYS will [\*\*\*] for such [\*\*\*].
- (iii) Except as expressly provided in Section 8.5(b)(iv), Capital One is solely responsible for obtaining, at Capital One's sole expense, all of the TPM Required to Provide Services that Capital One may require if Capital One exercises the License Option. Capital One acknowledges that TSYS recommended that Capital One make arrangements prior to the Effective Date with the vendors of all TPM Required to Provide Services listed on Schedule K at the Effective Date so that such TPM Required to Provide Services can be obtained by Capital One on terms acceptable to Capital One if Capital One exercises the License Option.
- (iv) If TSYS changes or adds to any of the TPM Required to Provide Services, and the changed or additional item is not a new version of an item of the TPM Required to Provide Services and is not commercially available, then in the applicable Schedule K Change Notice TSYS will so advise Capital One and will recommend one or more commercially available alternatives. Capital One shall be solely responsible for obtaining, at Capital One's sole expense, any such alternative that Capital One may require if it exercises the License Option. If Capital One exercise the License Option, TSYS shall, at TSYS' sole expense, make such modifications to the Licensed Materials as may be necessary so that the Licensed Materials are compatible with the alternative that was recommended by TSYS (as it exists at the time of the recommendation) and that was selected by Capital One.
- (v) During the Term, TSYS shall license and maintain, at TSYS' sole expense, the TPM Required to Provide Services and any additional Third Party Materials that are (A) part of the TSYS System or (B) required to operate the TSYS System or otherwise perform the Services.
- (vi) During the Term, Capital One shall license and maintain, at Capital One's sole expense except as provided in Section 8.5(b)(ii), the TPM Required to Receive Services.
- (vii) TSYS shall not [\*\*\*] any TSYS [\*\*\*] that is part of the Licensed Materials with [\*\*\*] without Capital One's consent, which Capital One may withhold in its sole discretion.
- (viii) If pursuant to Section 3.2(a) Third Party Software is added to the TSYS System at the request of Capital One (regardless of whether such Third Party Software is Application Software or Systems Software), unless TSYS was otherwise responsible to provide the required functionality and failed to do so, Capital One shall be responsible for the charges imposed by the relevant third party. None of such Third Party Software shall be included in the Licensed Materials.

## 8.6 Wholly New Developed Materials

Wholly New Developed Materials are outside the scope of this Agreement. Neither TSYS nor the Development Pool shall develop any Wholly New Developed Materials except pursuant to a separate written agreement entered into by the parties.

#### 8.7 Works Made for Hire

(a) Capital One Developed Materials shall be deemed "Works Made For Hire" for Capital One for purposes of copyright law. If, and to the extent, any of the Capital One Developed Materials is not deemed a "Work Made For Hire" under relevant law, TSYS hereby irrevocably assigns, transfers and conveys to Capital One without further consideration all of its Non-Patent Intellectual Property rights, title and interest in such Capital One Developed Materials (excluding any pre-existing TSYS Materials incorporated in such Capital One Developed Materials). Capital One and its assigns may obtain and hold in their own name all such Non-Patent Intellectual Property Rights in and to such Capital One Developed Materials. TSYS agrees to (and shall cause its employees and subcontractors to) execute any documents or take any other actions as may reasonably be necessary, or as Capital One may reasonably request, to perfect Capital One's ownership of such Non-Patent Intellectual Property Rights, without additional consideration and regardless of whether during or after the Term. TSYS shall maintain appropriate proprietary rights agreements with employees, subcontractors, and third parties sufficient to grant and transfer to Capital One all of the rights granted hereunder.

(b) Capital One hereby irrevocably assigns, transfers and conveys to TSYS without further consideration all of its Non-Patent Intellectual Property rights, title and interest in the TSYS Developed Materials (excluding any pre-existing Capital One Materials incorporated in such TSYS Developed Materials). TSYS and its assigns may obtain and hold in their own name all such Non-Patent Intellectual Property Rights in and to such materials. Capital One agrees to (and shall cause its employees and subcontractors to) execute any documents or take any other actions as may reasonably be necessary, or as TSYS may reasonably request, to perfect TSYS' ownership of such TSYS Developed Materials, without additional consideration and regardless of whether during or after the Term. Capital One shall maintain appropriate proprietary rights agreements with employees, subcontractors, and third parties sufficient to grant and transfer to TSYS all of the rights granted hereunder.

## 8.8 Patent Rights

(a) Patent Rights shall be owned by the party who was the employer of (or contracting party with) the inventor(s) of the patentable subject matter; provided that (i) Patent Rights in any new patentable subject matter that is invented (A) by the [\*\*\*] or (B) by TSYS Personnel, [\*\*\*] TSYS Personnel, on [\*\*\*] the Initial Conversion Plan shall be owned by TSYS and (ii) [\*\*\*] TSYS [\*\*\*] will be considered a "contracting party with" an employee or contractor [\*\*\*]. Capital One shall have a [\*\*\*], non-exclusive license (with right to sublicense to entities providing services to Capital One and its Affiliates) to practice (including the right to make and have made) under any of such patents for the purpose of providing services to Capital One and its Affiliates without further action being necessary for the term of any such patents.

(b) If TSYS does not [\*\*\*] with the [\*\*\*] and [\*\*\*] for a [\*\*\*] by [\*\*\*] (A) or (B) of Section 8.8(a) within [\*\*\*] months after the date such [\*\*\*] is [\*\*\*] by the parties (e.g., in [\*\*\*] or [\*\*\*]), then as between the parties, Capital One will have the [\*\*\*] to [\*\*\*] and [\*\*\*] a [\*\*\*] with respect to such [\*\*\*], at Capital One's sole cost and expense, and Capital One will [\*\*\*] and [\*\*\*] any [\*\*\*] that may [\*\*\*] of [\*\*\*]. TSYS shall have a [\*\*\*], non exclusive license (with the right to sublicense to entities receiving services or software from TSYS and its Affiliates) to practice (including the right to make and have made) under any patents Capital One obtains in such patentable invention for the purpose of providing services or software to TSYS

and its Affiliates, and their respective customers and licensees, without further action being necessary for the term of any such patents.

(c) Except as expressly provided in this Section 8, nothing in this Agreement is intended to grant, and no grant shall be implied of, any right, title, or other interest in any Patent Right of one party to the other party.

## 8.9 [\*\*\*] of TSYS Patents

At no time will TSYS or its Affiliates [\*\*\*] Capital One or any Capital One Affiliate or any of their service providers or vendors (solely to the extent that such service providers or vendors are providing services to Capital One or a Capital One Affiliate), any Covered Patent [\*\*\*] (a) [\*\*\*] such Covered Patent and of the portion of [\*\*\*] TS2 to [\*\*\*], which may be given during the Term and (b) [\*\*\*] the effective date of termination of this Agreement. "Covered Patent" shall mean any patent that relates to the Services provided to Capital One under this Agreement and that is in existence or applied for as of the date of termination of this Agreement or within the two year period thereafter. This Section 8.9 shall not apply to U.S. Patent 6,817,008, issued November 9, 2004.

## 8.10 Bankruptcy Code Section 365(n)

All rights and licenses to Developed Materials, TSYS Materials, and Capital One Materials granted under or pursuant to this Agreement will be deemed to be, for the purposes of Section 365(n) of the United States Bankruptcy Code (the "Bankruptcy Code"), licenses to rights in "intellectual property" as defined under the Bankruptcy Code. Accordingly, the licensee of such rights shall retain and may fully exercise all of its rights and elections under the Bankruptcy Code. In the event of the commencement of bankruptcy proceedings by or against either party under the Bankruptcy Code, the other party shall be entitled to retain all of its rights in such Developed Works, TSYS Materials, and Capital One Materials under this Agreement.

## 8.11 Residual Knowledge and Further Rights

- (a) Nothing contained in this Agreement shall restrict a party from the use of any general ideas, concepts, know-how, methodologies, processes, technologies, algorithms, or techniques relating to the Services that are retained in the unaided mental impressions of such party's personnel, which either party, individually or jointly, develops or discloses under this Agreement, provided that in doing so such party does not [\*\*\*] under Section 13 or otherwise infringe the Intellectual Property Rights of the other party or third parties who have licensed or provided materials to the other party. For the avoidance of doubt, this Section 8.11(a) is not intended to restrict TSYS' right to use TSYS Personnel that may have access to Capital One Confidential Information from working for other TSYS customers, provided (i) use of such personnel is not otherwise prohibited under this Agreement and (ii) in doing so TSYS does not [\*\*\*] under Section 13 or otherwise infringe the Intellectual Property Rights of Capital One.
- (b) TSYS' rights of use in this Section 8 are in all cases subject to Capital One Patent Rights (i.e., TSYS may not infringe upon any Capital One patent even if use of a material or functionality is permitted in this Section 8).
- (c) TSYS Developed Materials and information pertaining thereto that TSYS has a right to use under this Section 8 for a purpose other than providing Services for Capital One or that are derivative of Capital One Confidential Information shall have all Capital One

Confidential Information removed by TSYS prior to TSYS using or disclosing such TSYS Developed Materials for any purpose other than providing Services for Capital One.

(d) Except for the license rights contained in this Section 8, the TS2 License Agreement, and the [\*\*\*] Agreement, neither this Agreement nor any disclosure made under this Agreement grants any license to either party in the Intellectual Property Rights of the other party. This Section 8.11 shall survive termination/expiration of this Agreement.

#### 8.12 Export

Certain goods and technology to be provided under this Agreement, including certain software and technical data, as well as certain transactions under this Agreement may be subject to export controls under Applicable Laws. The parties shall ensure that all exports and re-exports of all goods and technology or any direct product thereof are made consistent with Applicable Laws, including all applicable export controls. Neither party shall undertake any transaction in violation of any such laws or regulations. To the extent within a party's control, that party shall be responsible for, and shall coordinate and oversee, compliance with such export laws in respect of such items exported or imported under this Agreement.

## 8.13 TS2 License Option

- (a) Pursuant to the terms of the TS2 License Agreement, TSYS has licensed the Licensed Materials to Capital One ("<u>TS2 License</u>"). At any time following either (i) completion of the Processing Year [\*\*\*] (except if the Agreement is being terminated by TSYS pursuant to Section 19.1(b)); or (ii) the date this Agreement is terminated by either party (except where such termination is (A) by Capital One for convenience pursuant to Section 19.2 effective prior to the end of Processing Year [\*\*\*]; (B) by TSYS pursuant to Section 19.1(b); or (C) by either party prior to the Commencement Date), Capital One may exercise an option to activate the TS2 License (the "License Option").
- (b) If Capital One exercises the License Option the parties shall cooperate to transition the Capital One Accounts from TSYS' to Capital One's (or its designee's) facility in accordance with the appropriate portions of Schedule A, applicable portions of Section 19.8, if any, and the TS2 License Agreement.
  - (c) If Capital One exercises the License Option, this Agreement will terminate on the Actual Migration Completion Date.
- (d) If Capital One exercises the License Option, and the effective date of termination of this Agreement pursuant to Section 8.13(c) is during the [\*\*\*] or [\*\*\*] Processing Years, then Capital One will pay the applicable Termination Charge specified in Section 6 of Schedule C.
- (e) On the date Capital One exercises the License Option, it shall elect whether or not to include [\*\*\*] in the Licensed Materials. Capital One may not elect to include [\*\*\*] in the Licensed Materials if Capital One did not [\*\*\*] to include [\*\*\*] in the scope of Services under this Agreement as of the Commencement Date, or did so elect but subsequently removed [\*\*\*] from the scope of Services under this Agreement, unless the parties reach agreement on new financial terms for the inclusion of [\*\*\*] in the Licensed Materials. The parties acknowledge that the [\*\*\*] in the TS2 License Agreement with [\*\*\*] included are based on the assumption that [\*\*\*] will be [\*\*\*] the Services provided by TSYS throughout the Term.

- (f) At any time prior to the Actual Migration Completion Date, Capital One may rescind the exercise of the License Option and continue receiving Services under this Agreement as if the License Option had not been exercised. Without modifying the parties' obligations under Section 7.4 of Schedule C, Capital One shall [\*\*\*] TSYS for its [\*\*\*] in [\*\*\*] for the [\*\*\*] (and any [\*\*\*] to [\*\*\*] to [\*\*\*] to [\*\*\*] the Services under this Agreement), such as [\*\*\*] to [\*\*\*] which had been [\*\*\*]) unless the reason for the [\*\*\*] is TSYS' [\*\*\*] to [\*\*\*] with its [\*\*\*] under the [\*\*\*].
- (g) Subject to the exceptions in Section 8.13(a), the right for Capital One to exercise the License Option will survive for [\*\*\*] days following the effective date of termination or expiration of this Agreement.

## PROJECT AND CONTRACT MANAGEMENT

## 9.1 Steering Committee

The parties shall form a steering committee to facilitate communications between them (the "<u>Steering Committee</u>"). The Steering Committee initially shall be composed of the persons set forth in Schedule R. Each party shall be entitled to half the seats. Either party may substitute other individuals, either temporarily due to the unavailability of a designated individual or permanently; provided that a proposed permanent substitution by TSYS shall require the prior approval of Capital One where such proposed substitution has a less senior position in the TSYS organization than the prior Steering Committee member.

## 9.2 Reports and Meetings

- (a) Reports. Schedule R contains a list of reports that TSYS will issue to Capital One. TSYS shall issue such reports at the frequency and in the format specified in Schedule R.
  - (i) As one such report, TSYS shall provide a monthly performance report, which shall be delivered to Capital One within [\*\*\*] Business Days after the end of each month, describing TSYS' performance of the Services in the prior month (the "Monthly Performance Report"). Such Monthly Performance Report shall:
    - (A) separately address TSYS' performance in each area of the Services, including, as applicable, conversion, development activities, and hosting and processing Services;
    - (B) for each area of the Services, assess the degree to which TSYS has attained or failed to attain the pertinent objectives in that area, including with respect to the Service Levels;
    - (C) include the information required by Section 3.3(b) of Schedule B;
    - (D) describe the status of problem resolution efforts, ongoing projects (including Conversion and development projects), and other initiatives;
    - (E) set forth a record of the material Equipment, Software and personnel changes that pertain to the Services and describe planned changes during the upcoming month that may affect the Services; and

- (F) set forth the number of Active Accounts and the utilization of capacity for the month and report on Account and utilization trends and statistics.
- (ii) As another such report, TSYS shall provide to Capital One regular information bulletins (no later than provided to any other TSYS customer) concerning the operation and development projects related to the [\*\*\*] TS2 and other components of TSYS' standard TS2 Processing Platform. These information bulletins shall:
  - (A) provide advance notice regarding upcoming enhancements to the [\*\*\*] TS2; and
  - (B) provide implementation dates, features, and specific directions on where to locate more detailed information.
- (b) Meetings. Schedule R contains a list of meetings that shall be held between representatives of Capital One and TSYS. Either party may prepare and circulate an agenda in advance of each such meeting to give participants an opportunity to prepare for the meeting. The parties shall establish procedures for preparing and circulating minutes promptly after each meeting, although the party not preparing the minutes shall not be bound thereto or obligated to correct or object to any errors therein. As of the Effective Date, such meetings shall include the following:
  - (i) an operations call [\*\*\*] to discuss the operation and performance of the Services since the last such call;
  - (ii) a [\*\*\*] meeting of the Capital One Contract Manager and the TSYS Services Manager to discuss the status of any on-going Conversion, day-to-day operations during the Term, and such other matters as appropriate;
  - (iii) a [\*\*\*] meeting among operational personnel representing Capital One and TSYS to discuss the Monthly Performance Report, daily performance, planned or anticipated activities and changes that might adversely affect performance, and otherwise to address, review and discuss matters specific to Capital One;
  - (iv) a [\*\*\*] management meeting of the Steering Committee to review the reports for the quarter, review TSYS' overall performance under the Agreement, review progress on the resolution of issues, provide a strategic outlook for new TS2 developments and discuss how they can be incorporated into [\*\*\*] TS2, and discuss such other matters as appropriate;
    - (v) a [\*\*\*] meeting by the parties to review relevant contract and performance issues; and
  - (vi) such other meetings between Capital One representatives and TSYS Personnel, including meetings between Capital One and TSYS' Chief Executive Officer and Chief Operating Officer, reasonably requested by either party as necessary to address performance of the Services.

#### 9.3 Procedures Manual

(a) On the Effective Date, the parties have approved the version of the procedures manual for this Agreement (the "<u>Procedures Manual</u>") by signing the acknowledgement page thereof. The table of contents for the Procedures Manual as of the Effective Date is provided in Schedule Y.

- (b) If the Procedures Manual is inconsistent with the requirements provided in the Main Body of the Agreement or Schedule A, then the applicable provisions of the Main Body of the Agreement or Schedule A, as applicable, shall control. The Procedures Manual may be amended as mutually agreed by the parties.
- (c) TSYS shall periodically (including prior to the Commencement Date) update the Procedures Manual to reflect changes in the operations or procedures described therein, subject to the following: (i) all updates of the Procedures Manual shall be provided to Capital One for review and comment and (ii) all updates of the issues resolution methodology, Change Control Procedure, and Development Methodology shall require Capital One's approval prior to becoming effective. TSYS shall perform the Services in accordance with the Procedures Manual, to the extent applicable.

## 9.4 Change Control

TSYS shall comply with the following change control requirements when making any changes to the TSYS System:

- (a) prior to implementing or changing any technology or functionality, in the TSYS System, or using any new Software or new Equipment to provide the Services, TSYS shall have complied with all applicable provisions of the Development Methodology and of the Change Control Procedure in the Procedures Manual (the "Change Control Procedure");
- (b) TSYS shall schedule, and shall use Commercially Reasonable Efforts to perform, all development work under this Agreement and all changes described in Section 9.4(a) so as not to have an adverse impact on the Services;
- (c) TSYS shall not make the following changes, including implementing or changing any technology or functionality in the applicable components of the TSYS System, without first obtaining Capital One's approval pursuant to the applicable provisions of the Development Methodology and/or the Change Control Procedure:
  - (i) a change adversely affecting the function or performance of, or decreasing to any significant degree the resource efficiency of, the [\*\*\*] or the [\*\*\*] in more than insubstantial ways;
    - (ii) a change materially reducing the functionality, parameters, capabilities, or processing speed of the [\*\*\*] TS2; or
    - (iii) a [\*\*\*] which has the effect of increasing Capital One's [\*\*\*] under this Agreement or other [\*\*\*] or [\*\*\*] of Capital One.

However, TSYS may make temporary changes required by an emergency if it has been unable to contact an appropriate Capital One representative to obtain such approval after making reasonable efforts. TSYS shall document and promptly report such emergency changes to Capital One, which changes shall then be subject to Capital One's approval under this Agreement.

(d) TSYS shall perform Conversion activities and move enhancements and programs from development and test environments to production environments in accordance with the Development Methodology and the Change Control Procedure.

## 9.5 Technology Planning

- (a) Each party will designate an equal number of individuals to serve on a subcommittee of the Steering Committee (the "<u>Technology</u> <u>Subcommittee</u>") to review and assess the performance of [\*\*\*] TS2, other components of the TSYS System, and [\*\*\*] and to develop plans for enhancing [\*\*\*] TS2, other components of the TSYS System, [\*\*\*], and the Services over time. Such plans may include:
  - (i) the identification of new developments and enhancements to improve the quality of [\*\*\*] TS2;
  - (ii) a description of the types and volumes of personnel skills and abilities needed to respond to any recommended changes or upgrades in technology and to changes, increases and decreases in Capital One's demand;
  - (iii) general plans for improving Service Level performance through improvements in [\*\*\*] TS2, other components of the TSYS System, and [\*\*\*]; and
  - (iv) a general plan and a projected time schedule for developing and achieving the recommended elements in accordance with the Development Methodology.
- (b) The Technology Subcommittee shall meet on a [\*\*\*] basis or more frequently if recommended by the Steering Committee to support Capital One's business planning cycles or to address changes in the business of Capital One that materially impact [\*\*\*] TS2, [\*\*\*], and other information technology systems.

## 9.6 TS2 Development Priorities

Capital One will be provided significant input into TSYS' development priorities for the [\*\*\*] TS2 Processing Platform and related technologies, including 'a seat at the table' for all user group and similar meetings. Capital One will have at least the same opportunity to beta test new developments on the [\*\*\*] TS2 Processing Platform as is provided to any other TSYS customer who has a dedicated version of a system that evolved from TS2.

## 9.7 Quality Assurance and Improvement Programs

As part of its total quality management process, TSYS shall provide quality assurance and quality improvement through: (a) the identification and application of proven techniques and tools from [\*\*\*]; and (b) the implementation of programs, practices and measures designed to improve compliance with the Service Levels. Such procedures shall include [\*\*\*], and other procedures for Capital One to confirm the quality of TSYS' performance. TSYS shall utilize project management tools, including productivity aids and project management systems, as appropriate in performing the Services.

## 9.8 Coordination of Additional Marketing to Capital One

TSYS shall coordinate all of its and its Affiliates marketing efforts with, and comply with the rules of marketing engagement provided from time to time by, the Capital One Contract Manager.

#### 9.9 No Amendment and Order of Precedence

Operational documentation (including the Documentation, Change Control Procedure and other parts of the Procedures Manual) shall not be used to, and shall not, amend this Agreement. Except to the extent otherwise provided in this Agreement, if there is a conflict between the provisions of this Agreement and such documentation, the provisions of this Agreement shall control.

## CAPITAL ONE RESPONSIBILITIES

## 10.1 Responsibilities

In addition to Capital One's responsibilities as expressly set forth elsewhere in this Agreement, Capital One shall be responsible for the following:

- (a) <u>Capital One Contract Manager</u>. Capital One shall designate one (1) individual to whom TSYS may address operational communications concerning this Agreement (the "<u>Capital One Contract Manager</u>"). The Capital One Contract Manager's responsibilities will include:
  - (i) serving as Capital One's principal point of contact regarding the Services and otherwise being responsible for all Services-related communications on behalf of Capital One;
    - (ii) overseeing and managing the performance of Capital One's obligations under this Agreement; and
    - (iii) providing any necessary input and making required decisions in a timely manner.
- (b) <u>Capital One Cooperation</u>. Capital One shall cooperate with TSYS, including by making available management decisions, information, approvals, and acceptances, as reasonably requested by TSYS so that TSYS may accomplish its obligations and responsibilities under this Agreement. In providing the Services, TSYS shall be entitled to rely on Capital One Data, information, and instructions provided by Capital One or its Affiliates to TSYS.
- (c) <u>Capital One Quality Control Notification</u>. Capital One shall check information produced by TSYS, including, as applicable, information used to produce periodic Capital One Account statements, embossed cards and reports, to determine if such information is correct. Capital One shall notify TSYS if Capital One determines as a result of such checking there has been a non-insubstantial error in the Services. TSYS shall not be liable to Capital One with respect to any damages arising from an error in the Services during a period when (i) TSYS is not and should not reasonably be aware of the error and (ii) a Capital One representative at the appropriate level is or reasonably should be aware of the error for more than [\*\*\*] Business Days without notifying TSYS. The prior sentence shall only relieve TSYS of damages arising from the error during the period covered by items (i) and (ii) and shall not relieve TSYS of responsibility to correct the error once identified or of damages arising before or after the period covered by items (i) and (ii). For example, if Capital One discovers an inaccuracy in the TSYS System and waits [\*\*\*] Business Days to report it to TSYS, then, as long as TSYS is not and should not reasonably be aware of the error, TSYS shall be relieved of damages arising from the

error during the period beginning on the [\*\*\*] Business Day and ending on the [\*\*\*] Business Day when the error is reported.

- (d) <u>Telecommunications Access to TSYS Data Center(s)</u>. Capital One shall be responsible for providing the Capital One Telecom Infrastructure described in Section 6.12 of Schedule A. Capital One shall ensure that all telecommunications equipment used by it conform to operational interface specifications that are mutually agreed from time to time. TSYS shall not unreasonably withhold its approval to changes in such equipment proposed by Capital One, and Capital One [\*\*\*] TSYS for any of its [\*\*\*] and [\*\*\*] resulting from such changes [\*\*\*] and until TSYS [\*\*\*] for any of its [\*\*\*].
- (e) <u>Equipment</u>. Capital One shall provide and maintain the Equipment in the Capital One Operating Environment. TSYS shall not change the configuration of [\*\*\*] TS2 in a manner which shifts the processing burden to Equipment required to be provided by Capital One in the Capital One Operating Environment without Capital One's prior written consent (which may be withheld in Capital One's discretion) and an appropriate adjustment of the charges to reflect the reallocation of responsibility.
- (f) <u>Security</u>. Capital One shall use Commercially Reasonable Efforts to implement security procedures designed to prevent unauthorized access to the TSYS System through the Capital One Operating Environment.
- (g) <u>Compliance With TSYS Rules</u>. Capital One will use Commercially Reasonable Efforts to cause its personnel assigned to the Development Pool to comply with TSYS' reasonable operational practices and procedures applicable to the Development Pool.
- (h) <u>Contractor Personnel in Development Pool</u>. Capital One shall be responsible for the obligations, services, and functions performed by personnel of third party contractors designated by Capital One who are Development Pool Personnel to the same extent as if such obligations, services, and functions were performed by Capital One. Capital One shall not disclose TSYS Confidential Information to a subcontractor unless and until such subcontractor agrees in writing to comply with obligations substantially similar to those in Section 13.3.

## 10.2 Savings Clause

- (a) Due to the material adverse impact termination of this Agreement or suspension of performance of the Services would have on Capital One's business, [\*\*\*].
- (b) TSYS ACKNOWLEDGES THAT CAPITAL ONE WOULD NOT BE WILLING TO ENTER INTO THIS AGREEMENT WITHOUT ASSURANCE THAT THIS AGREEMENT [\*\*\*] AND THAT TSYS [\*\*\*], AND ONLY TO THE EXTENT, EXPLICITLY PROVIDED HEREIN.
- (c) Without limiting any provisions hereof that expressly excuse TSYS from performance (including provisions relating to a Force Majeure Event), TSYS' nonperformance of its obligations under this Agreement [\*\*\*]: (i) [\*\*\*] the extent [\*\*\*]; and (ii) TSYS provides Capital One with [\*\*\*]. If requested by Capital One, TSYS shall use [\*\*\*] Capital One's, its Affiliates' (or such third party's) [\*\*\*] (with Capital One being responsible to [\*\*\*]).

## 11 CHARGES AND PAYMENT TERMS

ALL AMOUNTS TSYS MAY CHARGE FOR THE SERVICES ARE SET FORTH IN SCHEDULE C OR IN OTHER EXPRESS PAYMENT PROVISIONS OF THIS AGREEMENT. [\*\*\*] HAVE THE RIGHT TO [\*\*\*],

AND [\*\*\*] SHALL NOT BE REQUIRED TO [\*\*\*], [\*\*\*] AMOUNTS FOR THE SERVICES OTHER THAN OR IN ADDITION TO THOSE PAYABLE TO TSYS UNDER SCHEDULE C AND UNDER THE OTHER EXPRESS PAYMENT PROVISIONS OF THIS AGREEMENT. SCHEDULE C ALSO CONTAINS THE INVOICING AND PAYMENT TERMS APPLICABLE TO THE CHARGES.

## 12 AUDITS

## 12.1 Generally

TSYS shall allow Capital One and Capital One Affiliates, and their respective internal and external auditors and Regulators (collectively, "<u>Capital One Auditors</u>") reasonable access to TSYS Personnel and TSYS facilities, Equipment, Software, and Documentation used in providing Services to Capital One, and to the Capital One Data and Capital One-related records held by or under the control of TSYS at no additional charge and subject to the conditions set out in this Section 12, for the purpose of performing the audits described in this Agreement, including Section 6 and this Section 12. Capital One shall make good faith efforts to coordinate audits requested under this Section 12 so as to minimize the number of audits of the same subject matter performed separately by Capital One and the various Capital One Affiliates.

## 12.2 SAS 70 Audits

- (a) <u>General</u>. Annually during the Term, TSYS shall retain a third party auditor (the "<u>SAS 70 Auditor</u>") to perform an audit or series of audits, each of which audit(s) shall conform to the American Institute of Certified Public Accountants Statements of Auditing Standards No. 70 Type II (collectively, "<u>SAS 70</u>" audits) or any successor or substitute statement adopted by the American Institute of Certified Public Accountants, of the control activities and processes maintained by TSYS to provide the Services.
- (b) Scope of Audit. The scope of the SAS 70 audit(s) shall at a minimum include TSYS' [\*\*\*], [\*\*\*] TS2, [\*\*\*], ([\*\*\*] only), and other elements of the TSYS System and [\*\*\*] used to provide the Services to Capital One and shall be sufficient to satisfy audit requirements generally applicable to financial and card processing institutions under Applicable Laws. At least [\*\*\*] days prior to the initiation of each SAS 70 audit, TSYS shall provide in writing the proposed objectives and scope of the SAS 70 audit to Capital One. To the extent Capital One has any comments relative to such scope, it shall provide its comments in writing to TSYS at least [\*\*\*] days prior to the SAS 70 audit being performed, and if in such comments Capital One in good faith identifies [\*\*\*] or [\*\*\*] issues (other than an increase in the [\*\*\*] covered by the SAS 70 audit that is not required by Applicable Law) that the parties agree require a reasonable addition or change to the scope of the SAS 70 audit, then TSYS shall revise the scope of the SAS 70 audit that the parties do not agree to make pursuant to the previous sentence (including an increase in the [\*\*\*] covered by the SAS 70 audit that is not required by Applicable Law), then (i) TSYS shall make such additions or changes and (ii) Capital One shall reimburse TSYS for any incremental Out-of-Pocket Expenses it incurs in making such additions or changes. Promptly after TSYS' receipt of each such SAS 70 audit (but no later than [\*\*\*] weeks), TSYS shall provide Capital One with a copy of the SAS 70 audit report, except such [\*\*\*] week period shall be extended to [\*\*\*] days if requested by TSYS and approved by Capital One, which approval shall not be unreasonably withheld.

- (c) <u>Addressing Capital One Questions and Concerns</u>. Capital One may ask questions of TSYS concerning the SAS 70 audits and the SAS 70 reports, which TSYS shall promptly answer. If in Capital One's opinion TSYS' response does not satisfactorily answer Capital One's questions, or if TSYS fails to respond promptly, then at Capital One's request TSYS shall request the involvement of the SAS 70 Auditor to answer Capital One's questions.
- (d) <u>Deficiencies</u>. TSYS shall, as soon as reasonably practicable, remedy any material weakness noted in the SAS 70 audit, and regularly report on its progress in remedying such weaknesses (such reports shall be provided no less frequently than [\*\*\*], but if material deficiencies are noted or Capital One otherwise requests TSYS shall report [\*\*\*]). If the scope of the weakness requires efforts by TSYS beyond those that it is otherwise required to make in order to provide the Services, the effort to correct the same may be a New Additional Service in which event the charges therefor shall be established in accordance with Schedule C. Within [\*\*\*] weeks following correction of the weakness TSYS: (i) will have the SAS 70 Auditor verify that the weakness has been corrected in a manner that meets SAS 70 requirements; and (ii) provide Capital One with a copy of the verification report.
- (e) <u>Change of Auditor</u>. Upon any TSYS change of independent auditor acting as SAS 70 Auditor, TSYS shall provide Capital One written notice of such change as soon as it may legally do so. Any replacement SAS 70 Auditor shall be a nationally (United States) recognized auditing firm.

## 12.3 Regulatory Audits

- (a) If any Association or Regulator requires that it be permitted to review or audit the Services or information concerning such Services held by TSYS under this Agreement, TSYS upon reasonable advance notice from Capital One (to the extent Capital One is able or permitted to provide such notice) shall at Capital One's expense to the extent the amount of effort required is greater than that experienced by TSYS in regulatory audits for its other customers and the reason for such greater effort is caused by factors peculiar to Capital One, not to those involving TSYS, allow such Association or Regulator to conduct the required audit. At no time shall TSYS deny any such Association or Regulator access to facilities, records, Services, or materials necessary for it to perform a required audit.
- (b) The Parties acknowledge that performance of the Services in or for Capital One Accounts in Canada is subject to examination by Canadian regulatory agencies, including the Office of the Superintendent of Financial Institutions ("OSFI"), and that each party is and shall be authorized to submit or furnish to any such regulatory agency reports, information or other data as may be required or requested by them under Applicable Laws to otherwise permit access by such agencies to information and data relevant to the Services. TSYS, to the extent legally permissible, shall promptly notify Capital One of, and provide details with respect to, any Capital One Confidential Information it provides to OSFI. TSYS also agrees to execute undertakings in favor of OSFI regarding the inspection of such data or information by OSFI, the undertakings to be in such form and contain such terms as the OSFI may reasonably require for that purpose from time to time.

## 12.4 Capital One Audits

(a) Capital One may conduct one (1) Capital One Audit (as defined below) on an annual basis for any reason. Additionally, if at any time, including during a disaster situation,

Capital One has reasonable material concerns or a reasonable basis for requesting information, Capital One shall first address such concerns or request such information from TSYS' management, and to the extent its reasonable material concerns or reasonable basis for requesting information remain unsatisfied after such discussions and the other audits permitted hereunder, then Capital One shall have the right to [\*\*\*] Capital One Audits. Each "Capital One Audits" may consist of an audit of:

- (i) general controls and security practices and procedures used to protect Capital One Confidential Information;
- (ii) the TSYS System and [\*\*\*] Software and the integrity of Capital One Data processed, stored, supported and transmitted by the TSYS System or [\*\*\*] Software;
- (iii) books and records of TSYS which relate to the Services, including all supporting financial records and invoices upon which the Charges are based; provided that: (A) Capital One shall not be entitled to see any [\*\*\*] of TSYS except to the extent such [\*\*\*] is required to support a Charge for a [\*\*\*] Expense, [\*\*\*] Expense, or any Service for which this Agreement expressly provides that the Charge is to be computed by reference to [\*\*\*]; and (B) in the case of [\*\*\*] of the type described in the exception to item (A) that TSYS may not disclose to Capital One because of its obligations of confidentiality to third parties even after using Commercially Reasonable Efforts to obtain the right to disclose such information to Capital One, in lieu of disclosing such information to Capital One TSYS shall [\*\*\*] an [\*\*\*] to [\*\*\*] to Capital One that the Charges based on such [\*\*\*] are accurate);
  - (iv) supporting information and calculations regarding compliance with Service Levels;
  - (v) disaster recovery and back-up procedures;
- (vi) information regarding the [\*\*\*] and [\*\*\*] of TSYS in performing the Services to the extent directly affecting Charges for, or timing of, Services, provided that (A) Capital One shall not be entitled to see any [\*\*\*] information of TSYS except to the extent such [\*\*\*] information is required to support a Charge for a [\*\*\*] Expense, [\*\*\*] Expense, or any Service for which this Agreement expressly provides that the Charge is to be computed by reference to [\*\*\*]); and (B) in the case of [\*\*\*] of the type described in the exception to item (A) that TSYS may not disclose to Capital One because of its obligations of confidentiality to third parties even after using Commercially Reasonable Efforts to obtain the right to disclose such information to Capital One, in lieu of disclosing such information to Capital One TSYS shall [\*\*\*] an [\*\*\*] to [\*\*\*] to Capital One that the Charges based on such [\*\*\*] are accurate); and
- (vii) subject to Section 12.3, any other information or materials reasonably necessary to enable Capital One to meet, or to confirm that TSYS is meeting its obligations and responsibilities under Section 6.
- (b) Each Capital One Audit shall be subject to the following conditions:
- (i) Notice and Consent. Capital One will give TSYS not less than [\*\*\*] days' prior written notice of the date(s) on which Capital One proposes to conduct a Capital One Audit (this restriction shall not apply to Associations and Regulators). TSYS shall not unreasonably withhold its consent to Capital One's conduct of a Capital One Audit on date(s) proposed by Capital One, but reserves the right to withhold its consent as to date(s) on which: (A) the relevant TSYS Personnel are not reasonably available due to their other work-related

duties (but TSYS shall maintain sufficient staff that this shall not be a material impediment or persistently cause undue delay); or (B) other audits or regulatory examinations are being conducted on the same date(s) and the conduct of the Capital One Audit on such date(s) would unreasonably burden the TSYS Personnel and materially interfere with TSYS' business activities. If TSYS reasonably withholds its consent, TSYS shall work in good faith to schedule the Capital One Audit as soon as is reasonably practicable. This will include TSYS promptly proposing different date(s) for the Capital One Audit, which shall be no more than [\*\*\*] days after the date of Capital One's original desired date.

- (ii) [\*\*\*]. The notice referenced in Section 12.4(b)(i) will be accompanied by a copy of Capital One's proposed objective and scope of the Capital One Audit. Prior to Capital One Auditors arriving on-site, Capital One and representatives of TSYS will discuss and agree upon reasonable rules and administrative procedures applicable to Capital One's proposed [\*\*\*] and the logistics of conducting the Capital One Audit. The Capital One Audit must be conducted substantially in accordance with such agreement. Notwithstanding such agreement, Capital One shall be entitled to change the [\*\*\*] and [\*\*\*] of the Capital One Audit if at any time before or during such audit Capital One finds or receives information which reasonably raises additional concerns for Capital One, provided if Capital One changes the [\*\*\*] or [\*\*\*] of a Capital One Audit in a material way and TSYS is unable to accommodate the changed [\*\*\*] or [\*\*\*] on the scheduled audit dates, then TSYS may require Capital One to provide notice and obtain consent pursuant to Section 12.4(b)(i) prior to conducting the changed portion of the Capital One Audit. This requirement shall not apply to Associations and Regulators.
- (iii) <u>Minimize On-Site Presence</u>. No more than [\*\*\*] Capital One Auditors may be on-site at TSYS' facilities at any time, and no more than [\*\*\*] of the [\*\*\*] Capital One Auditors may be external auditors unless they are Capital One's official outside auditors or are performing what would otherwise be internal audit functions. This requirement shall not apply to Associations and Regulators.
- (iv) Minimize Disruption. Capital One will conduct the Capital One Audit in such a way as to minimize unnecessary disruption of TSYS' business activities, and the right of Capital One to review TSYS' systems will be limited to the specific access necessary to perform the Capital One Audit. Capital One shall use reasonable efforts to minimize the number of Capital One Auditors required to conduct the Capital One Audit in an efficient and timely manner. This requirement shall not apply to Associations and Regulators. Capital One may not install, add or execute Software, or install, add or attach hardware, on any TSYS system in connection with any Capital One audit; provided, however, that Capital One may [\*\*\*] and [\*\*\*] of TSYS' information technology infrastructure in accordance with Section [\*\*\*] of this Agreement.
- (v) <u>Record Removal Restrictions</u>. The Capital One Audit shall be conducted on site at TSYS' premises under the observation of TSYS, and Capital One will not remove original copies of TSYS records from TSYS' premises, although Capital One may remove its audit notes and, with TSYS' consent (which consent TSYS shall not unreasonably withhold), copies of such records, and such copies shall constitute TSYS Confidential Information. This requirement shall not apply to Associations and Regulators.

- (vi) <u>TSYS Security</u>. At all times while on site at TSYS' premises, Capital One will comply with TSYS' standard written security policies and procedures applicable to its general customer population of which Capital One is given prior written notice. This requirement shall not apply to Associations and Regulators.
- (vii) <u>Costs</u>. Each of Capital One and TSYS will bear its own costs associated with any Capital One Audit. There shall be [\*\*\*] to Capital One by TSYS in connection with TSYS' compliance with any of the audit provisions in this Section 12 unless Capital One spends more than [\*\*\*] Business Days during the first Processing Year or [\*\*\*] Business Days during any subsequent Processing Year, on all Capital One audits during that year. If Capital One spends more Business Days than are permitted above during any Processing Year on all Capital One audits during that year, then Capital One will pay reasonable costs associated with data reports, handler costs, and other audit related expenses, including the cost of TSYS Personnel occupied with supervising and facilitating the Capital One Audit.

# 12.5 Audit Follow-up and Notification

- (a) Following an audit or examination performed in connection with this Agreement, Capital One or the Capital One Auditors may conduct an exit conference with TSYS to obtain factual concurrence with issues identified in the review.
- (b) At Capital One's request, TSYS and Capital One shall promptly meet to review each Capital One Audit report after its issuance and to mutually agree upon the appropriate manner, if any, in which to respond to the changes suggested by the audit report.
- (c) If TSYS or its Affiliates become aware (through internal or external audits or otherwise) of any deficiency in its operating practices or procedures that may be reasonably expected to have a material and adverse impact on the Services or Capital One, then: (i) TSYS shall provide Capital One with prompt formal notice thereof; (ii) TSYS executive management shall promptly meet with Capital One's designated personnel to discuss the situation (to the extent legally permissible) and the steps TSYS will take to promptly remediate such deficiency; and (iii) TSYS shall perform such steps discussed by the parties and frequently report on the status of such efforts until the deficiency has been corrected.
- (d) TSYS shall provide Capital One with prompt notice of any audit of TSYS operations conducted by a Regulator, excluding health and safety audits.

# 12.6 Records Retention

- (a) TSYS shall maintain and provide, during each Capital One Audit, access upon request to the following records, documents, and other information, and any other records, documents, and other information reasonably requested by Capital One, required to meet Capital One's audit rights under this Agreement:
  - (i) letters of intent;
  - (ii) Specifications;
  - (iii) project documentation;
  - (iv) copies of audits performed by or for TSYS under Section 12.2 and relevant audit information arising out of TSYS' reporting obligation under Section 12.5(c);
    - (v) insurance policies;

- (vi) invoices and supporting documentation;
- (vii) non-disclosure and [\*\*\*] agreements that TSYS is required to obtain pursuant to Section 13.3(f); and
- (viii) Procedures Manual and related updates.
- (b) TSYS shall maintain such records until the latest of:
- (i) [\*\*\*] after the date the record was created (except with respect to audit reports which shall be retained for [\*\*\*] after creation and TSYS Personnel agreements and contracts, safety and health procedures, and insurance policies which shall be retained for [\*\*\*] after the date they are last in effect);
  - (ii) the information is no longer required to comply with Capital One Directions; and
  - (iii) the information is no longer required for TSYS to comply with Applicable Laws applicable to TSYS.

provided, however, that when such records pertain to a pending matter relating to this Agreement (e.g., a dispute) they shall be retained until the matter is closed.

# 12.7 Overcharges

- (a) If as a result of an audit or otherwise it is determined that TSYS has overcharged Capital One, TSYS shall credit Capital One's account an amount equal to the overcharge plus [\*\*\*] at the [\*\*\*] calculated from the date of payment by Capital One to TSYS; provided that if TSYS discloses such overcharges to Capital One and offers to refund such overcharges in full prior to Capital One's indication of the intention to audit and such overcharge was not the result of an intentional act to conceal such overcharges from Capital One, no interest shall be charged thereon if the amount of the overcharge does not exceed [\*\*\*] percent ([\*\*\*]%) of the Charges during the period audited.
- (b) If the audit discloses that TSYS' overcharges exceeded [\*\*\*] percent ([\*\*\*]%) of the Charges during the prior [\*\*\*] months, then TSYS shall also reimburse Capital One for the reasonable cost of the portion of the audit related to such overcharges.

# SAFEGUARDING OF DATA; CONFIDENTIALITY

# 13.1 Safeguarding Capital One Data

- (a) Capital One Data shall be and remain, as between the parties, the property of Capital One. TSYS shall not: (i) use Capital One Data for any purpose other than in connection with providing the Services; (ii) sell, assign, lease or otherwise dispose of Capital One Data to third parties; (iii) possess or assert any lien or other right against Capital One Data; or (iv) commercially exploit Capital One Data.
- (b) TSYS shall establish and maintain safeguards against the destruction, loss or alteration of Capital One Data in the possession or control of TSYS which are no less rigorous than those maintained by TSYS for its other customers' information of a similar nature.
  - (c) Without limiting the generality of Section 13.1(a):
  - (i) If TSYS Personnel gain, or if TSYS believes the TSYS Personnel might have gained, access to Capital One Data that they are not permitted to access under this

Agreement, TSYS shall promptly report such incident to Capital One, describe in detail the accessed Capital One Data, and if requested return to Capital One any such Capital One Data.

- (ii) TSYS shall utilize Commercially Reasonable Efforts, including systems security measures, to guard against the unauthorized access, alteration, or destruction of Capital One Data. Such measures shall include the use of Software which:
  - (A) requires all users to enter a user identification and password prior to gaining access to the information systems;
  - (B) controls and tracks the addition and deletion of users authorized to access Capital One Data; and
  - (C) controls and tracks user access to areas and features of the information systems.

# 13.2 GLB Compliance

- (a) TSYS acknowledges that Capital One is subject to Title V of the GLB Act and the Interagency Guidelines Establishing Standards for Safeguarding Customer Information adopted by federal bank regulatory agencies, including the OTS and Board of Governors of the Federal Reserve System, as each may be amended from time to time (collectively, the "GLB Regulations") and may be subject to Canadian Privacy Legislation pursuant to which Capital One is required to comply with certain regulations and implement certain procedures and processes with respect the privacy, use and protection of Personally Identifiable Information of Capital One's customers or prospective customers, and, as a result of engaging TSYS to perform the Services is required to obtain certain undertakings from TSYS with regard to the privacy, use and protection of Personally Identifiable Information of Capital One's customers or prospective customers. Therefore, notwithstanding anything to the contrary contained in this Agreement, and in addition to (and not in substitution for) TSYS' other obligations under this Agreement, TSYS shall, in accordance with the GLB Act and the GLB Regulations, for so long as it retains Customer Information:
  - (i) protect and keep confidential all Personally Identifiable Information about or pertaining to Capital One's customers, prospective customers or other individuals about whom Capital One has collected Personally Identifiable Information that is disclosed by Capital One or otherwise obtained by TSYS in the performance of its duties under this Agreement (collectively, the "<u>Customer Information</u>"). Customer Information includes all data considered "nonpublic personal information" as defined in Title V of the GLB Act and applicable regulations promulgated thereunder. During the Term, TSYS shall collect and use Customer Information only to the extent necessary to exercise its rights or perform its obligations under this Agreement for which such Customer Information was disclosed to TSYS or as otherwise directed by Capital One;
  - (ii) implement process and procedures designed to ensure that TSYS Personnel shall not attempt to access, or allow access to, any Customer Information that they are not permitted to access under this Agreement;
    - (iii) have administrative, technical, and physical safeguards to:
      - (A) ensure the security and confidentiality of such Customer Information;

- (B) protect against any anticipated threats or hazards to the security or integrity of such Customer Information;
- (C) protect against unauthorized access to or use of such Customer Information; and
- (D) without limiting the generality of items (A) through (C) preceding, meet the objectives of the GLB Act and the GLB Regulations.

Such safeguards shall include using systems and controls that: (1) requires all users to enter a user identification and password prior to gaining access to Customer Information; (2) controls and tracks the addition and deletion of users authorized to access Customer Information; and (3) controls and tracks user access to Customer Information:

- (iv) implement and maintain, at TSYS' own expense, a [\*\*\*] compliance with this Section 13.2 and appropriate mitigation strategies to address issues identified as a result of such [\*\*\*]; and
- (v) notify Capital One if TSYS discovers there has been a material breach of, or a serious attempt to breach, its security safeguards required by this Section 13.2 such that the security of Customer Information has been, or TSYS has a reasonable basis to believe may have been, compromised (including unauthorized access to and unauthorized attempts to access Customer Information). TSYS shall provide such notice promptly, and in all cases within [\*\*\*] hours, after discovery of the breach or attempted breach. TSYS shall take appropriate actions to address incidents of unauthorized access to or use of Customer Information. Capital One may take all reasonable and appropriate steps to protect Customer Information in such event, including investigating TSYS' security safeguards required under this Section 13.2.
- (b) TSYS shall not retain Customer Information unless it is required to do so to perform the Services or by applicable law.
- (c) Upon reasonable notice to TSYS, Capital One and its representatives may (at Capital One's option) review either TSYS' information technology infrastructure security or an independent audit provided by TSYS in order to monitor TSYS' compliance with the terms of this Section 13.2. TSYS shall implement, and maintain, at TSYS' own expense, appropriate mitigation strategies to address issues identified as a result of these reviews.
- (d) The obligations set forth in this Section 13.2 are material for purposes of determining a breach of the Agreement, and they shall survive termination of this Agreement.

# 13.3 Confidentiality

(a) <u>Confidential Information</u>. TSYS and Capital One each acknowledge that they may be furnished with, receive or otherwise have access to information of or concerning the other party that such party considers to be confidential, a trade secret or otherwise restricted. "<u>Confidential Information</u>" shall mean all information, in any form, furnished or made available directly or indirectly by one party to the other that is marked confidential, restricted, or with a similar designation. Confidential Information furnished or made available by Capital One is "<u>Capital One Confidential Information</u>". Confidential Information furnished or made available by TSYS is "<u>TSYS Confidential Information</u>". The terms and conditions of this Agreement shall be deemed Confidential Information of each party.

- (b) <u>Capital One Confidential Information</u>. Capital One Confidential Information also shall include, whether or not designated "Confidential Information":
  - (i) Capital One Data and Customer Information;
  - (ii) Capital One Materials, including credit risk models, optimization models, authorization models, and marketing and product strategies;
  - (iii) the specifications, designs, documents, correspondence, software, documentation, data and other materials and work products furnished by Capital One to TSYS for use in the course of performing the Services;
  - (iv) all information concerning the operations, affairs and businesses of Capital One, the financial affairs of Capital One, and the relations of Capital One with its customers, employees and service providers (including customer lists, customer information, account information and consumer markets); and
    - (v) any other information reasonably designated as Capital One Confidential Information in this Agreement.
  - (c) TSYS Confidential Information. TSYS Confidential Information also shall include, whether or not designated "Confidential Information":
    - (i) Third Party Materials licensed to TSYS;
    - (ii) the TSYS System;
  - (iii) the specifications, designs, documents, software, documentation, data and other materials and work products disclosed by TSYS or its Affiliates to Capital One within the Development Pool;
  - (iv) the specifications, designs, documents, software, documentation, data and other materials and work products disclosed by TSYS or its Affiliates to Capital One before Conversion that (A) was either covered by the Confidentiality Agreement between the parties dated May 27, 2004 or (B) a reasonably prudent business person would have considered to be of a confidential nature;
  - (v) all information concerning (A) TSYS employees and individual subcontractor personnel that a reasonably prudent employer would treat as confidential and/or (B) the relations of TSYS and its Affiliates with their service providers; and
    - (vi) any other information reasonably designated as TSYS Confidential Information in this Agreement.

# (d) General Obligations.

- (i) Neither party shall use the Confidential Information of the other party for any purpose other than in connection with providing or receiving, as applicable, the Services in accordance with this Agreement.
- (ii) Each party shall use at least the same degree of care as it employs to avoid unauthorized disclosure of its own information, but in any event no less than Commercially Reasonable Efforts, to prevent disclosing to unauthorized persons the Confidential Information of the other party, provided that TSYS may disclose such information to properly authorized entities as and to the extent necessary for performance of the Services, and Capital

One may disclose such information to its Affiliates and to third parties as and to the extent necessary for Permitted Users' receipt of the Services, where in each such case, the receiving entity first agrees in writing to comply with obligations substantially similar to those provided in this Section 13.3.

- (iii) In addition, if a party to this Agreement requires the other party to use or exchange information with a third party in a manner contemplated by this Agreement, and the third party requires such other party to execute a non-disclosure agreement, then such other party shall not unreasonably refuse to agree to execute the form of non-disclosure agreement required by such a third party if such other party's obligations under the provisions thereof, taken as a whole, are not substantially greater than the party's obligations hereunder.
- (iv) In the event of any actual or suspected misuse, disclosure or loss of, or inability to account for, any Confidential Information of the furnishing party, the receiving party shall:
  - (A) notify the furnishing party promptly after becoming aware thereof;
  - (B) furnish to the other party full details of the unauthorized possession, use, or knowledge, or attempt thereof, and use reasonable efforts to assist the other party in investigating or preventing the reoccurrence of any unauthorized possession, use, or knowledge, or attempt thereof; and
  - (C) promptly take such actions as may be necessary or reasonably requested by the furnishing party to minimize the violation and any damage resulting therefrom.
- (v) Neither party shall commence any legal action or proceeding against a third party with respect of any unauthorized possession, use, or knowledge, or attempt thereof, of Confidential Information by any person or entity which action or proceeding identifies the other party or the other party's Confidential Information without the other party's prior written consent.
- (vi) The parties' obligations respecting Confidential Information shall continue for a period of [\*\*\*] years following expiration or termination of this Agreement, provided that the obligation with respect to Capital One Data, Customer Information, and the TSYS System shall continue in perpetuity, without limitation.
- (e) <u>Additional Obligations with Regard to Capital One</u> [\*\*\*]. This Section 13.3(e) provides certain additional confidentiality obligations that apply to "Capital One [\*\*\*]". Capital One [\*\*\*] shall mean any [\*\*\*] of Capital One [\*\*\*] based on the [\*\*\*], including [\*\*\*] (e.g., [\*\*\*]), [\*\*\*] and [\*\*\*], and [\*\*\*].
  - (i) Prior to the Commencement Date, TSYS shall provide (and update Schedule J to include) a complete list of the TSYS Personnel that the parties believe require access to Capital One [\*\*\*] to perform the Services. The parties will update such list during the Term as necessary to maintain a complete list of the TSYS Personnel who have had or who are expected to require access to Capital One [\*\*\*], and will cooperate to minimize the number of TSYS Personnel requiring access to Capital One [\*\*\*] in performance of the Services. Without each party's consent such list shall not exceed [\*\*\*] individuals.

# (ii) TSYS shall:

- (A) use Commercially Reasonable Efforts to cause TSYS Personnel with access to Capital One [\*\*\*] to not disclose such information to other TSYS Personnel not listed on Schedule J (or the current list mutually maintained), and TSYS shall promptly (and in all cases within [\*\*\*] hours) notify Capital One of any unauthorized disclosure of which TSYS has actual knowledge; and
- (B) inform Capital One in writing, upon Capital One's request, of the [\*\*\*] of other [\*\*\*] of TSYS for whom each such individual is [\*\*\*] during the [\*\*\*] months following his or her last access to Capital One [\*\*\*].

# (f) Non-Disclosure Agreements.

- (i) TSYS shall require that (A) all TSYS Personnel who have access to Capital One [\*\*\*] execute the [\*\*\*] agreement attached as "Form A" in Schedule I; and (B) subject to the last sentence of this Section 13.3(f)(i), all TSYS Personnel who are authorized to have access to Capital One Data or Capital One Confidential Information execute the non-disclosure agreement attached as "Form B" in Schedule I (Form A and Form B collectively, "Non-Disclosure Agreements"). TSYS shall provide the Capital One Contract Manager with copies of all Non-Disclosure Agreements executed by TSYS Personnel prior to the date such TSYS Personnel begin work on the Capital One account. In the case of TSYS Personnel who are not TSYS employees, another form may be used lieu of Form B so long as it provides substantially equivalent protection to Capital One and TSYS complies with its other obligations under Section 5.7.
- (ii) TSYS shall [\*\*\*] TSYS Personnel. If TSYS changes any of its standard non-disclosure agreements (i.e., form agreements used generally for other TSYS accounts) in order to address changes in Applicable Laws, TSYS shall inform Capital One of the changes it is making and the rationale for such changes and will not unreasonably refuse to consent to any changes to the Non-Disclosure Agreements desired by Capital One to comply with changes in Applicable Law.
- (iii) If the requirement to have individuals who are to have access to Capital One [\*\*\*] execute Form A causes TSYS [\*\*\*] to [\*\*\*] the Services in accordance with this Agreement, TSYS shall inform Capital One and the parties shall attempt to ameliorate the situation. Until such situation is ameliorated TSYS shall be [\*\*\*] any [\*\*\*] to [\*\*\*] the Services to the [\*\*\*] is [\*\*\*] by such requirement.
- (iv) Capital One may change the list of companies in Attachment A to Form A at any time to conform to the Capital One Competitor list, effective only as to individuals who have not yet at that time executed the Form A. If an individual who has executed Form A and has left (or is about to leave) the employ of TSYS notifies TSYS that he or she has [\*\*\*], TSYS shall promptly provide Capital One a copy of the notice. If the TSYS [\*\*\*] or other TSYS [\*\*\*] have actual knowledge of circumstances that indicate that an individual who has executed Form A and/or Form B is in breach of his or her obligations thereunder TSYS shall promptly notify Capital One. If an individual who has executed Form A gives notice pursuant to Section 2(b) of Form A that he or she intends to [\*\*\*] by Section 2(a) of Form A, and Capital One indicates that it desires that TSYS [\*\*\*] pursuant to Section [\*\*\*], TSYS shall [\*\*\*] by Section [\*\*\*] and [\*\*\*] to [\*\*\*] on [\*\*\*] (including the [\*\*\*] of any [\*\*\*] on TSYS).

- (v) Subject to the last sentence of this Section 13.3(f)(v), Capital One shall [\*\*\*] execute the non-disclosure agreement attached as "Form C" in Schedule I. Capital One shall provide TSYS with [\*\*\*]. In the case of [\*\*\*], another form may be used in lieu of Form C so long as it provides substantially equivalent protection to [\*\*\*].
- (g) Exclusions. Section 13.3(d) shall not apply to any particular information (other than Capital One Data and any Customer Information) which TSYS or Capital One can demonstrate:
  - (i) was, at the time of disclosure to it, in the public knowledge;
  - (ii) after disclosure to it, is published or otherwise becomes part of the public knowledge through no fault of the receiving party;
  - (iii) was lawfully in the possession of the receiving party at the time of disclosure to it without obligation of confidentiality;
  - (iv) was received after disclosure to it from a third party who had a lawful right to disclose such information to it without any obligation to restrict its further use or disclosure; or
    - (v) was independently developed by the receiving party without reference to Confidential Information of the furnishing party.

# 13.4 <u>Disclosures Required by Law</u>

A party shall not be considered to have breached its obligations by disclosing Confidential Information (including Capital One Data and Customer Information) of the other party as required to satisfy any Applicable Laws, provided that, promptly upon receiving any request or demand therefor (to the extent that it may legally do so) such party advises the other party of the request prior to making such disclosure in order that the other party may interpose an objection to such disclosure, take action to assure confidential handling of the Confidential Information, and take such other action as it deems appropriate to protect the Confidential Information. A party receiving legal process (such as subpoenas, civil investigative demands, etc.) from a third party related to a claim against or investigation of the other party, shall be entitled to be reimbursed by the other party for all reasonable Out-of-Pocket Expenses incurred in connection with responding thereto, including reasonable attorneys' fees.

# 13.5 No Implied Rights

Nothing contained in this Section 13 shall be construed as obligating a party to disclose its Confidential Information to the other party, or as granting to or conferring on a party, expressly or impliedly, any rights or license to the Confidential Information of the other party, and any such obligation or grant shall only be as provided by other provisions of this Agreement.

# 13.6 Return of Confidential Information

(a) As requested by Capital One, TSYS shall return or destroy, as Capital One may direct, all material in any medium that contains, refers to, or relates to Capital One Confidential Information, including Capital One Confidential Information contained in TSYS Developed Materials and retain no copies (other than archival data files which shall not be used for any purpose and shall be destroyed in the ordinary course of TSYS' computer file backup

procedures and, provided that TSYS so notifies Capital One, one (1) copy may be retained by internal or external legal counsel for the purpose of evidentiary preservation).

(b) Each party shall, upon termination or expiration of this Agreement and completion of TSYS' Termination/Expiration Assistance obligations, return or destroy the Confidential Information of the other party that is in its possession or under its control; provided, however, that (i) each party may retain a reasonable number of copies of this Agreement, and such other Confidential Information of the other party (other than Personally Identifiable Information) as is mutually agreed, or is required for such party to comply with Applicable Laws; (ii) Capital One shall not be required to return any Software, Documentation, or other materials to which it has received a license that survives termination or expiration of this Agreement; and (iii) each party's obligation under this paragraph with respect to non-critical Confidential Information shall be limited to using Commercially Reasonable Efforts to return or destroy such information.

# 13.7 Corporate Risk Controls and Information Security Review

- (a) TSYS shall work in good faith to achieve compliance with all information [\*\*\*] provided in [\*\*\*] and, once achieved, to maintain such compliance at all times during the Term.
- (b) Upon reasonable advance notice to TSYS, Capital One and its representatives may meet at TSYS' offices in Columbus, Georgia, or other TSYS facilities used to provide the Services during normal business hours with TSYS Personnel responsible for information security and business continuity matters within TSYS' audit department for purposes of:
  - (i) reviewing TSYS' information security and business continuity policies, procedures and related documentation;
  - (ii) physically inspecting the TSYS facilities and information technology and business continuity infrastructure used to provide the Services, provided (A) TSYS Personnel shall accompany Capital One at all times during such inspections and (B) such inspections may not adversely impact, or compromise the confidentiality of, the operations of another customer of TSYS; and
  - (iii) making inquiries of such personnel with regard to such items, as they relate to providing Services to Capital One, (collectively, a "Capital One Security Review").
- (c) TSYS shall develop, implement, and maintain, at [\*\*\*], mitigation strategies to address any deficiencies identified as a result of the Capital One Security Reviews that the parties agree may be reasonably expected to have a material and adverse impact on the Services or Capital One. If TSYS does not agree that a deficiency identified during a Capital One Security Review exists or may be reasonably expected to have a material and adverse impact on the Services, then at Capital One's request, TSYS shall: (A) engage a third party to review the deficiency in accordance with the process described below in Section 13.7(d); (B) add a review of the deficiency to the scope of the next SAS 70 audit to be provided pursuant to Section 12.2; or (C) require TSYS promptly to take action specified by Capital One to address the deficiency and [\*\*\*] for the [\*\*\*] TSYS [\*\*\*].

(d) Capital One shall not have the right to perform [\*\*\*] and [\*\*\*] reviews of TSYS' IT infrastructure except in accordance with this Section 13.7 and as permitted under Section 12.4. If Capital One has reasonable material concerns that a deficiency exists in TSYS' IT infrastructure that may be reasonably expected to have a material and adverse impact on the Services or Capital One, and such concerns are not addressed to Capital One's reasonable satisfaction in the SAS 70 audit provided pursuant to Section 12.2 or through the process described above in Section 13.7(b), then at Capital One's written request (but no more than once in any calendar year unless an event of unauthorized access to Capital One Information through TSYS' IT infrastructure has occurred in which case Capital One may request such [\*\*\*] immediately following any such event), TSYS will engage, on [\*\*\*] and [\*\*\*], TSYS' independent auditor (which must be a nationally (United States) recognized auditing firm) to conduct [\*\*\*] of TSYS' IT infrastructure in accordance with mutually agreed upon procedures, and, subject to TSYS' [\*\*\*] to its other customers, [\*\*\*]. Upon completion of its report for any such [\*\*\*], the independent auditor will provide copies to [\*\*\*] TSYS and [\*\*\*]. TSYS shall develop, implement, and maintain, at TSYS' expense, appropriate mitigation strategies to address any deficiencies identified as a result of these [\*\*\*] that may be reasonably expected to have a material and adverse impact on the Services or Capital One.

(e) If Capital One spends more than [\*\*\*] Business Days at a TSYS facility during any Processing Year on all [\*\*\*] conducted pursuant to this Section 13.7 during that year, then Capital One will pay reasonable costs associated with data reports, handler costs, and other review related expenses of TSYS, including the cost of TSYS Personnel occupied with supervising and facilitating the [\*\*\*]. The [\*\*\*] Business Day threshold in the prior sentence shall exclude time spent performing (i) inspections of third party facilities and materials pursuant to a separate agreement and (ii) initial inspections of new facilities for which TSYS is required to obtain Capital One approval pursuant to Section 3.5

# REPRESENTATIONS AND WARRANTIES

# 14.1 System Quality and Compliance

TSYS represents and warrants that:

- (a) as of the Effective Date, the TSYS System and [\*\*\*] conform to the Documentation in all material respects;
- (b) as of the Commencement Date, the TSYS System and [\*\*\*] will conform to the [\*\*\*] for the Initial Conversion; and
- (c) as of the Effective Date there are, and as of the Commencement Date there will be, no known errors in the TSYS System or [\*\*\*] that will have a material adverse effect on Capital One's use of the Services in accordance with the Documentation.

# 14.2 [\*\*\*]Performance

TSYS represents and warrants that it has achieved the levels of performance [\*\*\*] using the TSYS System [\*\*\*] and [\*\*\*] shown in Schedule O.

# 14.3 Work Standards

(a) TSYS represents, warrants, and covenants that the Services shall be rendered with promptness and diligence and shall be executed in a workmanlike manner, in

accordance with the practices and professional standards used in well-managed operations performing services similar to the Services.

(b) For all aspects of the Services as to which there is no specific Service Level provided herein, TSYS will promptly re-perform any such Services that do not comply with the foregoing warranty or, upon its failure to do so, will, at Capital One's option, [\*\*\*] for [\*\*\*] Services. If Capital One elects to [\*\*\*] a [\*\*\*] in respect of [\*\*\*] Services and does not revoke such election within [\*\*\*] days, such refund shall be Capital One's [\*\*\*] for such failure, although to the extent such failure is a breach of this Agreement Capital One shall retain the rights provided in Section 19.1(a) with respect to such breach.

#### 14.4 Efficiency and Cost Effectiveness

TSYS represents, warrants, and covenants that to the extent affecting the Charges or Capital One's retained costs it shall use Commercially Reasonable Efforts to (i) use efficiently the resources or services necessary to provide development Services and (ii) perform the development Services in a cost-effective manner consistent with the required level of quality and performance.

#### 14.5 Technology

TSYS represents, warrants, and covenants that it shall provide the Services using proven technology that is kept current in a manner consistent with the Change Control Procedure.

# 14.6 Open Source Software

TSYS represents, warrants, and covenants that no "open source" Source Code is included in the Licensed Materials as of the Effective Date, or will be included in the Licensed Materials during the Term, that TSYS is required to license or distribute for free or for a limited charge to the public.

# 14.7 Non-Infringement

- (a) TSYS represents that, as of the Effective Date, it does not have any actual knowledge (without any duty of inquiry beyond asking those employees responsible for intellectual property rights management whether they have actual knowledge of any such claims) of any infringement or misappropriation claim or potential claim that [\*\*\*] TS2 or other component of the TSYS System infringes, or constitutes an infringement or misappropriation of, any Intellectual Property Right of any third party.
- (b) Capital One represents that, as of the Effective Date, it does not have any actual knowledge (without any duty of inquiry beyond asking those employees responsible for intellectual property rights management whether they have actual knowledge of any such claims) of any infringement or misappropriation claim or potential claim that the Capital One Materials infringe, or constitute an infringement or misappropriation of, any Intellectual Property Right of any third party.

# 14.8 Ownership or Authorization to Use

TSYS represents and warrants that it is either the owner of, or authorized to Use as provided in this Agreement, the TSYS System, TSYS Materials, Documentation, Deliverables, and other materials provided by TSYS under this Agreement.

# 14.9 Authorization and Other Contracts

Each party represents and warrants to the other that:

- (a) it has the requisite corporate power and authority to enter into this Agreement and to carry out the transactions contemplated by this Agreement;
- (b) the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated by this Agreement have been duly authorized by the requisite corporate action on the part of such party and will not constitute a violation of any judgment, order, or decree binding on such party or its assets:
- (c) the execution, delivery, and performance of this Agreement and the consummation of the transactions contemplated by this Agreement will not constitute a material default under any material contract by which it or any of its material assets are bound, or an event that would, with notice or lapse of time or both, constitute such a default; and
- (d) there is no proceeding to which it is a party, or, to the knowledge of the party, that is threatened, that challenges or may have a material adverse affect on this Agreement or the transactions contemplated by this Agreement.

#### 14.10 Inducements

TSYS represents and warrants to Capital One that it has not violated any Applicable Laws regarding the offering of unlawful inducements in connection with this Agreement.

# 14.11 <u>User Documentation</u>

TSYS represents, warrants, and covenants that all Documentation delivered to Capital One in connection with the Services will accurately describe the functions, features and, if any, [\*\*\*] on the use of the Services and the procedures for an end-user exercising such functions and features, except for the [\*\*\*] on [\*\*\*] provided in Schedule O.

# 14.12 **Viruses**

Each party represents, warrants, and covenants that it shall use Commercially Reasonable Efforts so that no Viruses are coded or introduced into the TSYS System. If a Virus is found to have been introduced into the TSYS System, TSYS shall use Commercially Reasonable Efforts at no additional charge to reduce the effects of the Virus on the TSYS System; provided that if the Virus is found to have been introduced by Capital One, Capital One shall be responsible for TSYS' expenses in complying with the foregoing.

# 14.13 Disabling Code

TSYS represents, warrants, and covenants that:

- (a) without the prior written consent of Capital One, which Capital One may withhold in its sole discretion, TSYS shall not intentionally insert into the Software resident on Capital One's computers or other resources any code that would have the effect of disabling or otherwise shutting down or limiting any portion of the Services ("Disabling Code"); and
- (b) with respect to any Disabling Code that may be part of the Software resident on Capital One's computers or other resources, TSYS shall not intentionally invoke such Disabling Code at any time for any reason, including upon expiration or termination of this Agreement for any reason. If any Disabling Code is invoked, TSYS shall use its Commercially

Reasonable Efforts to remediate the effects of such Disabling Code. With respect to matters not specifically addressed by the Disaster Recovery Plan, such Commercially Reasonable Efforts shall include TSYS acting to restore the full functionality of the Services in the same manner that it would act if it were the owner of both its business and the affected Capital One business.

Because of the irreparable injury such actions would have on Capital One's business, Capital One shall be entitled to seek and obtain (and TSYS irrevocably consents to Capital One obtaining) injunctive relief with respect to any breach of this Section 14.13. In doing so, Capital One may proceed on an ex parte basis and shall not be required to post bond or prove damages. Such right is non-exclusive and shall not constitute an election of remedies.

#### 14.14 Deliverables

TSYS represents, warrants, and covenants that each Deliverable provided to Capital One shall not, after final acceptance of the Deliverable by Capital One, materially deviate from the applicable Specifications.

# 14.15 Disclaimer

THE WARRANTEES SET FORTH IN THIS AGREEMENT ARE EXCLUSIVE. OTHER THAN AS PROVIDED IN THIS AGREEMENT, THERE ARE NO EXPRESS WARRANTIES AND THERE ARE NO IMPLIED WARRANTIES, INCLUDING THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

# 15 INSURANCE

# 15.1 Insurance Coverage

TSYS represents that as of the Effective Date it has, and agrees that during the Term it shall maintain in force, at least the following insurance coverages:

- (a) Worker's Compensation Insurance and other similar social insurance in accordance with the laws of the country, state or territory exercising jurisdiction over the employee with minimum limits required by law;
- (b) Employer's Liability Insurance (Primary and Umbrella/Excess), including coverage for occupational injury, illness and disease, with minimum limits per employee and per event of [\*\*\*] Dollars (\$[\*\*\*]) and a minimum aggregate limit of [\*\*\*] Dollars (\$[\*\*\*]) or the minimum limits required by law, whichever limits are greater;
- (c) Commercial General Liability Insurance, including Products Liability, Completed Operations, Premises Operations, Personal Injury, and Contractual and Broad Form Property Damage Liability coverages (Primary and Umbrella/Excess), on an occurrence basis, with a minimum per occurrence combined single limit of [\*\*\*] Dollars (\$[\*\*\*]) and a minimum aggregate combined single limit of [\*\*\*] Dollars (\$[\*\*\*]). This coverage shall be endorsed to name Capital One and its Affiliates as additional insureds;
- (d) Comprehensive Automotive Liability Insurance covering use of all owned, non-owned and hired automobiles used to perform the Services or on Capital One's property for purposes related to the Services for bodily injury, property damage, uninsured motorist and underinsured motorist liability with a minimum combined single limit per accident of [\*\*\*]

Dollars (\$[\*\*\*]) or the minimum limit required by law, whichever limit is greater. This coverage shall be endorsed to name Capital One and its Affiliates as additional insureds:

- (e) Commercial Crime Insurance, including blanket coverage for Employee Dishonesty and Computer Fraud, for loss or damage arising out of or in connection with any fraudulent or dishonest acts, including wrongful conversion of: (i) Capital One's, or any of its Affiliates' or Users', property; (ii) the property of any customers or patrons of such persons; or (iii) the property of others in TSYS' possession, care, custody or control, committed by TSYS' employees, agents or subcontractors, acting alone or in collusion with others, with a minimum limit per event of [\*\*\*] Dollars (\$[\*\*\*]). This coverage shall be endorsed to name Capital One and its Affiliates as loss payees;
- (f) Professional Liability Insurance for Errors and Omissions covering liability for loss or damage due to an act, error, omission, or negligence, with a minimum limit per event of [\*\*\*] Dollars (\$[\*\*\*]);
- (g) Electronic Data Processing Insurance providing coverage for all risks of loss or damage to equipment, data, media and valuable papers that are in the possession, care, custody or control of TSYS pursuant to this Agreement. Such insurance shall include Extra Expense and Business Income coverage and have a minimum limit adequate to cover such risks on a replacement costs basis. This coverage shall be endorsed to name Capital One and its Affiliates as loss payees;
- (h) Property Insurance, including Extra Expense and Business Income coverage, for all risks of physical loss of or damage to buildings, business personal property or other property that is in the possession, care, custody or control of TSYS pursuant to this Agreement. Such insurance shall have a minimum limit adequate to cover risks on a replacement costs basis. This coverage shall be endorsed to name Capital One and its Affiliates as loss payees; and
- (i) Excess or Umbrella Liability Insurance with a minimum limit of [\*\*\*] Dollars (\$[\*\*\*]) in excess of the insurance coverage described in Sections 15.1(b), 15.1(c), and 15.1(d) above.

# 15.2 <u>Insurance Provisions</u>

- (a) To the extent allowed by the individual insurers and policies, TSYS shall cause its insurers to [\*\*\*], and [\*\*\*] and [\*\*\*] and [\*\*\*] regarding Capital One.
- (b) The amounts of insurance required in Section 15.1 may be satisfied by TSYS purchasing primary coverage in the amounts specified or by TSYS buying a separate Excess or Umbrella Liability policy together with lower limit primary underlying coverage. The structure of the coverage is at TSYS' option, so long as the total amount of insurance meets the requirements in 15.1.
- (c) With regard to the Services performed under this Agreement, the coverages TSYS is required to carry under Section 15.1 shall be [\*\*\*], and all coverage shall be [\*\*\*] with respect to any other insurance or self-insurance that may be maintained by Capital One. The insurance coverage under Sections 15.1(a), 15.1(b), 15.1(c), 15.1(d), and 15.1(i) shall be written on an occurrence form basis. If any other coverage is written on a claims-made basis, it shall have a retroactive date no later than the Effective Date and, notwithstanding the termination of this Agreement, either directly or through 'tail' coverage shall be maintained for a period of at

least [\*\*\*] following completion of the Term, or it shall allow for reporting of claims until the later of [\*\*\*] after the end of the Term or the period of the applicable [\*\*\*] has expired.

- (d) TSYS shall cause its insurers to [\*\*\*], and [\*\*\*], and [\*\*\*], and [\*\*\*] on the insurance coverages under Sections 15.1(c), 15.1(d), and 15.1(i).
- (e) TSYS shall cause its insurers to issue certificates of insurance evidencing that the coverages and policy endorsements required under this Agreement are maintained in force. TSYS shall [\*\*\*] Capital One not less than [\*\*\*] days' written notice prior to any adverse modification, cancellation, or non-renewal of the policies. The insurers selected by TSYS shall be of good standing and authorized to conduct business in the jurisdictions in which Services are to be performed. When the policy is issued each such insurer shall have at least an A.M. Best rating of at least A-IX or with such other companies as may be approved in advance by Capital One.
- (f) In the case of loss or damage or other event that requires notice or other action under the terms of any insurance coverage specified in this Section 15, TSYS shall be solely responsible to take such action. TSYS shall provide Capital One with contemporaneous notice and with such other information as Capital One may request regarding the event.
- (g) TSYS' obligation to maintain insurance coverage shall be in addition to, and not in substitution for, TSYS' other obligations under this Agreement and TSYS' liability to Capital One for any breach of an obligation under this Agreement that is subject to insurance under this Agreement shall not be limited to the amount of coverage required under this Agreement.

# 15.3 Risk of Loss

Each party will be responsible for the risk of loss of, and damage to, any Equipment, Software, or other resources or materials in transit from such party to the other party until such resources or materials have been received by the other party (i.e., the sending party is responsible for in-transit risk of loss).

If there is a conflict between this Section 15 and the insurance requirements in any of the Schedules or Exhibits to this Agreement, this Section 15 shall control.

# 16 INDEMNITIES

# 16.1 Indemnity by TSYS

TSYS shall indemnify, defend, and hold harmless Capital One and its Affiliates and their respective officers, directors, employees, agents, successors, and assigns from and against any and all Losses arising from, in connection with, or based on allegations of, any of the following:

- (a) any claim or action by subcontractors arising out of TSYS' breach or violation of TSYS' subcontracting arrangements;
- (b) TSYS' intentional or criminal misconduct or conduct that evinces reckless disregard for the rights of others;
- (c) TSYS' breach of Section 13;

- (d) TSYS' repudiation of this Agreement, including a failure to perform during a dispute and any improper or wrongful termination or abandonment of the work under this Agreement by TSYS;
  - (e) failure of TSYS to comply with or perform its obligations under Section 6;
  - (f) any claim or action arising out of:
  - (i) a violation of Applicable Laws for the protection of persons or members of a protected class or category of persons by TSYS or its employees, subcontractors or agents;
    - (ii) sexual, racial, age, gender or other prohibited discrimination or harassment by TSYS, its employees, subcontractors or agents; or
  - (iii) work-related injury or death caused by TSYS, its employees, subcontractors, or agents (except to the extent covered by Capital One's workers compensation coverage);
- (g) any claim or action by any employee or contractor of TSYS or its Affiliates or subcontractors that Capital One is liable to such person as the employer or joint employer of such person, including and any claim for employee benefits as a result thereof; and
- (h) any and all liens, encumbrances, security interests, or other lien rights of any kind asserted by any of TSYS' subcontractors or suppliers, or by subcontractors or suppliers of TSYS' subcontractors, which in whole or in part, are based on any work, Services, material, or equipment to be provided under this Agreement.

# 16.2 <u>Indemnity by Capital One</u>

Capital One shall indemnify, defend, and hold harmless TSYS and its Affiliates and their respective officers, directors, employees, agents, successors, and assigns from any and all Losses arising from, in connection with, or based on allegations of, any of the following:

- (a) Capital One's intentional or criminal misconduct or conduct that evinces reckless disregard for the rights of others;
- (b) Capital One's breach of Section 13;
- (c) failure of Capital One to comply with or perform its obligations under Section 6;
- (d) any claim or action arising out of:
- (i) a violation of Applicable Laws for the protection of persons or members of a protected class or category of persons by Capital One or its employees, third party contractors or agents;
- (ii) sexual, racial, age, gender or other prohibited discrimination or harassment by Capital One, its employees, third party contractors or agents; or
- (iii) work-related injury or death caused by Capital One, its employees, third party contractors (other than TSYS and its Affiliates and contractors), or agents (except to the extent covered by TSYS' workers compensation coverage); and

(e) any claim or action by any employee or contractor of Capital One or its Affiliates that TSYS is liable to such person as the employer or joint employer of such person, including and any claim for employee benefits as a result thereof.

# 16.3 Additional Indemnities

Each party (indemnitor) agrees to indemnify, defend, and hold harmless the other, and its Affiliates, officers, directors, employees, agents, successors, and assigns (collectively, the indemnitee), from any and all Losses arising from, in connection with, or based on allegations of, any of the following:

- (a) the death or bodily injury of any agent, employee (other than an employee of the indemnitor), customer, business invitee, or business visitor or other person caused by the tortious conduct of the indemnitor;
  - (b) the damage, loss or destruction of any real or tangible personal property caused by the tortious conduct of the indemnitor; and
- (c) any claim, demand, charge, action, cause of action, or other proceeding asserted against the indemnitee but resulting from an act or omission of the indemnitor in its capacity as an employer of a person.

# 16.4 Infringement

- (a) <u>TSYS Indemnity</u>. TSYS shall indemnify, defend, and hold harmless Capital One and its Affiliates and their respective officers, directors, employees, agents, successors, and assigns from and against any and all Losses arising from, in connection with, or based on any third party claims of infringement or misappropriation of any Patent Right (other than a [\*\*\*]) related to a patent issued by the United States or Canada or any Non-Patent Intellectual Property Right, alleged to have occurred because of the TSYS System, TSYS Materials, Documentation, Deliverables or other materials provided by TSYS under this Agreement, or based upon performance of the Services by TSYS or a TSYS Affiliate; provided that the indemnity in this Section 16.4(a) shall not apply (and TSYS shall not be obligated to indemnify Capital One and its Affiliates) to the extent an infringement or misappropriation is caused by:
  - (i) specifications provided by or on behalf of Capital One for [\*\*\*] TS2 or other Deliverable;
  - (ii) Capital One business processes, methods, business plans, and procedures that Capital One requires TSYS to use or follow;
  - (iii) Capital One's failure to permit TSYS or the Development Pool to implement a change to [\*\*\*] TS2 if TSYS: (A) notifies Capital One that the change is necessary to cure an infringement or misappropriation and (B) allows Capital One a reasonable amount of time to authorize TSYS or the Development Pool to implement the change;
  - (iv) combinations by Capital One with items, including Third Party Materials and the Capital One Operating Environment, that were not (A) provided or approved by TSYS or (B) required in order for Capital One to receive the Services;
    - (v) Capital One's use of the TSYS System in a manner not permitted by this Agreement (unless permitted by the Documentation); or

- (vi) TSYS' use of Capital One Materials as contemplated in this Agreement.
- (b) <u>Capital One Indemnity</u>. Capital One shall indemnify, defend, and hold harmless TSYS and its Affiliates and their respective officers, directors, employees, agents, successors, and assigns from and against any and all Losses arising from, in connection with, or based on third party claims of infringement or misappropriation of any Patent Right (other than any [\*\*\*] by [\*\*\*] (and their successors and assigns) which are not within the definition of a [\*\*\*] hereunder) related to a patent issued by the United States or Canada or Non-Patent Intellectual Property Right, alleged to have occurred in connection with TSYS' provision of the Services to Capital One, Capital One Affiliates, or any Permitted Users where such infringement occurred because of the Capital One Materials or specifications provided by Capital One under this Agreement (and such infringement would not have otherwise occurred absent such Materials or specifications); provided, the indemnity in this Section 16.4(b) shall not apply (and Capital One shall not be obligated to indemnify TSYS and its Affiliates) to the extent an infringement or misappropriation is caused by:
  - (i) TSYS' failure to implement a change in Capital One Materials or specifications used by TSYS to perform the Services if Capital One (A) notifies TSYS that the change is necessary to cure an infringement or misappropriation; and (B) authorizes TSYS to implement the change;
  - (ii) combinations of Capital One Materials or specifications by TSYS with items that were not (A) provided or approved by Capital One or (B) required in order for TSYS to provide the Services;
  - (iii) TSYS' failure to comply with an obligation to a third party relating to the Capital One Materials that was disclosed in writing by Capital One; or
    - (iv) TSYS' use of Capital One Materials or specifications in a manner not permitted by this Agreement.

For the avoidance of doubt, to the extent this Agreement authorizes TSYS to use any items (including Capital One Materials) for its own benefit or to provide products or services to any third party, TSYS does so at its own risk and Capital One shall not be liable to TSYS, its Affiliates, or any third party claiming under or through them for infringement or misappropriation of any Intellectual Property Rights resulting from uses other than providing the Services to Capital One.

- (c) Remedy for TSYS Infringement. If any item used by TSYS to provide the Services or embedded in the TSYS System becomes, or in TSYS' reasonable opinion is likely to become, the subject of an infringement or misappropriation claim or proceeding (other than a claim or proceeding against which TSYS is not required to indemnify Capital One pursuant to Section 16.4(a)), in addition to indemnifying Capital One as provided in Section 16.4(a), TSYS shall:
  - (i) use Commercially Reasonable Efforts to promptly secure the right at TSYS' expense to continue using the item; or
  - (ii) if this cannot be accomplished with Commercially Reasonable Efforts, then at TSYS' expense, replace or modify the item to make it non-infringing or without

misappropriation, provided that any such replacement or modification will not degrade the performance or quality of the affected component of the Services; or

- (iii) if neither of the foregoing can be accomplished by TSYS with Commercially Reasonable Efforts, and only in such event, then remove the item from the Services, in which case:
  - (A) the Charges shall be equitably reduced to reflect such removal;
  - (B) TSYS shall refund any Charges paid by Capital One to develop or license the [\*\*\*] and any other items or materials that are no longer functional without the [\*\*\*]; and
  - (C) if such removal either (x) relates to a [\*\*\*] or (y) is otherwise [\*\*\*] to the remaining Services (taken as a whole), Capital One may terminate such portion of the affected Services or the entire Agreement without liability by giving TSYS notice designating the termination date.
- (d) <u>Remedy for Capital One Infringement</u>. If any item used by TSYS to provide the Services or embedded in the TSYS System becomes, or in Capital One's reasonable opinion is likely to become, the subject of an infringement or misappropriation claim or proceeding against which Capital One is required to indemnify TSYS pursuant to Section 16.4(b), in addition to indemnifying TSYS as provided in Section 16.4(b):
- (i) TSYS shall after consultation with Capital One cease using such item and if so instructed by Capital One (and at Capital One's expense) substitute any suitable replacement item available which does not infringe; and
- (ii) TSYS shall be excused for its failure to perform its obligations under this Agreement to the extent such failure is caused by TSYS' ceasing to use such item and until such a suitable replacement item is substituted.
- (e) <u>Sole and Exclusive Remedy</u>. Except in the event of a breach of Section 14.7, this Section 16.4 provides each party's sole and exclusive remedy with respect to infringement and misappropriation claims that are subject to indemnification under this Section 16.4.

# 16.5 Indemnification Procedures

With respect to third-party claims the following procedures shall apply:

(a) <u>Notice</u>. Within ten (10) Business Days after receipt by any entity entitled to indemnification under this Section 16 of notice of the assertion or the commencement of any action, proceeding or other claim by a third party in respect of which the indemnitee will seek indemnification pursuant to any such Section, the indemnitee shall notify the indemnitor of such claim in writing. No failure to so notify an indemnitor shall relieve it of its obligations under this Agreement except to the extent that it can demonstrate damages attributable to such failure. Within fifteen (15) days after receipt of notice from the indemnitee relating to any claim, but at least ten (10) days before the date on which any response to a complaint or summons is due, the indemnitor shall notify the indemnitee in writing if the indemnitor elects to assume control of the defense and settlement of that claim (a "Notice of Election").

- (b) <u>Procedure Following Notice of Election</u>. If the indemnitor delivers a Notice of Election relating to any claim within the required notice period, the indemnitor shall be entitled to have sole control over the defense and settlement of such claim, which it shall defend actively and with all reasonable diligence; provided that (i) the indemnitee shall be entitled to participate in the defense of such claim and to employ counsel at its own expense to assist in the handling of such claim; and (ii) the indemnitor shall obtain the prior written approval of the indemnitee (not to be unreasonably withheld or delayed) before entering into any settlement of such claim or ceasing to defend against such claim except that such approval shall not be required if the indemnitor acknowledges in writing its responsibility for payments and other liabilities under the settlement and the settlement does not provide for an equitable remedy against the indemnitee, make the indemnitor responsible for any damages, or include an admission of the indemnitee. After the indemnitor has delivered a Notice of Election relating to any claim in accordance with the preceding paragraph (and, in fact, diligently defends the claim), the indemnitor shall not be liable to the indemnitee for any legal expenses incurred by the indemnitee in connection with the defense of that claim. In addition, the indemnitor shall not be required to indemnify the indemnitee for any amount paid or payable by the indemnitee in the settlement of any claim for which the indemnitor has delivered a timely Notice of Election if such amount was agreed to by the indemnitee without the prior written consent of the indemnitor.
- (c) <u>Procedure Where No Notice of Election Is Delivered Or The Claim Is Not Otherwise Defended.</u> If the indemnitor does not deliver a Notice of Election relating to a claim, or otherwise fails to assume the defense of a claim, within the required notice period or fails to diligently defend the claim, the indemnitee may defend the claim in such manner as it may deem appropriate (without any obligation to consult with or obtain any consent from the indemnitor). If it is determined that the indemnitor failed to defend a claim for which it was liable, the indemnitor shall not be entitled to challenge the amount of any settlement or compromise paid by the indemnitee.

# 16.6 Subrogation

If an indemnitor shall be obligated to indemnify an indemnitee pursuant to this Section 16, the indemnitor shall, upon fulfillment of its obligations with respect to indemnification, including payment in full of all amounts due pursuant to its indemnification obligations, be subrogated to the rights of the indemnitee with respect to the claims to which such indemnification relates.

Nothing in this Section 16 shall limit a party's rights or obligations under Section 15 and nothing under Section 15 shall limit a party's rights or obligations under this Section 16.

# 7 LIABILITY AND FORCE MAJEURE

# 17.1 General Intent

Subject to the specific provisions of this Section 17, it is the intent of the parties that each party shall be liable to the other party for any actual damages incurred by the non-breaching party as a result of the breaching party's failure to perform its obligations in the manner required by this Agreement.

#### 17.2 Liability Restrictions

- (a) SUBJECT TO SECTIONS 17.2(c) AND 17.2(f), IN NO EVENT, WHETHER IN CONTRACT, IN TORT (INCLUDING BREACH OF WARRANTY, NEGLIGENCE AND STRICT LIABILITY IN TORT), OR OTHERWISE, SHALL A PARTY BE LIABLE TO THE OTHER PARTY (OR ANYONE CLAIMING UNDER OR THROUGH THE OTHER PARTY) FOR INDIRECT OR CONSEQUENTIAL, EXEMPLARY, PUNITIVE OR SPECIAL DAMAGES ARISING OUT OF, RESULTING FROM, OR IN ANY WAY CONNECTED WITH THE PERFORMANCE OR BREACH OF THIS AGREEMENT, EVEN IF SUCH PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES IN ADVANCE.
- (b) Subject to Sections 17.2(c), 17.2(d), and 17.2(e) each party's total cumulative liability to the other, whether in contract, in tort (including breach of warranty, negligence and strict liability in tort), or otherwise arising out of, resulting from, or in any way connected with, the performance and all breaches of this Agreement shall be limited in the aggregate to an amount equal to [\*\*\*] to TSYS pursuant to this Agreement for proper performance of the Services for the [\*\*\*] months prior to the month in which the most recent event giving rise to liability occurred; provided that if such most recent event giving rise to liability occurs (i) prior to [\*\*\*] days after the Commencement Date, such aggregate amount shall be equal to \$[\*\*\*]; and (ii) after [\*\*\*] days after the Commencement Date but prior to the end of Processing Year [\*\*\*], such aggregate amount shall be equal to \$[\*\*\*]. [\*\*\*], and amounts recoverable under Sections 17.2(c), 17.2(d), and 17.2(e) shall not count against and shall not reduce the amounts available under the foregoing limitations. This Section 17.2(b) does not limit Capital One's obligation to pay the Charges for the Services. Any amounts paid under Section 17.2(f) shall count against and reduce the aggregate amount available under this Section 17.2(b). In order to avoid a double recovery by Capital One, if a cause of action accrued prior to [\*\*\*] after the Commencement Date and Capital One has recovered damages under the [\*\*\*] Agreement, the limitation of liability hereunder shall be reduced by the amount recovered under the [\*\*\*] Agreement (but not, in any event, below the amount of the limitation of liability which would have been provided hereunder without consideration of charges under the [\*\*\*] Agreement).
  - (c) The limitations set forth in Sections 17.2(a) and 17.2(b) shall not apply with respect to:
    - (i) claims that are the subject of indemnification pursuant to either Section [\*\*\*] or [\*\*\*];
  - (ii) a claim that is subject to indemnification pursuant to either Section [\*\*\*] or Section [\*\*\*] if, as to such claim, (x) the indemnifying party has breached its representation in Section [\*\*\*], or (y) the indemnifying party has [\*\*\*] the [\*\*\*] of [\*\*\*] (that is, the indemnifying party had [\*\*\*] ([\*\*\*] of [\*\*\*]) that [\*\*\*] and [\*\*\*] to [\*\*\*] the [\*\*\*] specified in Section [\*\*\*] or [\*\*\*], as applicable);
    - (iii) damages occasioned by a party's breach of its obligations under Section [\*\*\*]; and/or

- (iv) damages occasioned by TSYS' repudiation of this Agreement, including a failure to perform during a dispute, any improper or wrongful termination, or abandonment of the work under this Agreement by TSYS.
- (d) The limitations in Section 17.2(b) shall not apply with respect to [\*\*\*] claims that are the subject of indemnification pursuant to:
  - (i) Section [\*\*\*];
  - (ii) Section [\*\*\*];
  - (iii) Section [\*\*\*];
  - (iv) Section [\*\*\*];
  - (v) Section [\*\*\*];
  - (vi) Section [\*\*\*]; and/or
  - (vii) Section [\*\*\*].
- (e) The limitations in Section 17.2(b) shall not apply with respect to [\*\*\*] claims occasioned by:
  - (i) any [\*\*\*]; and/or
  - (ii) a party's breach of Section [\*\*\*].
- (f) The limitations in Section 17.2(a) shall not apply to damages incurred by Capital One and its Affiliates which in the aggregate exceed [\*\*\*] Dollars (\$[\*\*\*]) (i.e., such amount is a deductible, not a threshold) but which do not in the aggregate exceed [\*\*\*] Dollars (\$[\*\*\*]) with respect to:
  - (i) damages occasioned by any [\*\*\*]; and/or
  - (ii) damages occasioned by a party's breach of Section [\*\*\*].
  - (g) Each party shall have a duty to mitigate damages for which the other party is responsible.
  - (h) If more than one of the exclusions above applies, the exclusion that provides the greatest relief to the aggrieved party shall apply.

# 17.3 Force Majeure

(a) Neither party shall be liable for any default or delay in the performance of its obligations under this Agreement (other than the obligation to pay amounts due under this Agreement): (i) if and to the extent such default or delay is caused, directly or indirectly, by fire, flood, earthquake, elements of nature or acts of God, riots, pestilence, civil disorders, governmental orders under the Defense Production Act, or any other cause beyond the reasonable control of such party, (ii) provided the non-performing party is without fault in causing such default or delay, and such default or delay could not reasonably be circumvented by the non-performing party through the use of alternate sources, workaround plans or other means (including, with respect to TSYS, by TSYS meeting its obligations for performing disaster recovery services as provided in this Agreement) (each such event, a "Force Majeure Event").

- (b) In such event the non-performing party shall be excused from further performance or observance of the obligations so affected for as long as such circumstances prevail and such party continues to use Commercially Reasonable Efforts to recommence performance or observance promptly. In the case of TSYS, with respect to matters not specifically addressed by the [\*\*\*], Commercially Reasonable Efforts shall include TSYS acting to [\*\*\*] the [\*\*\*] of the Services in the same manner that it would act if it were the owner of both its business and the [\*\*\*]. The party so delayed in its performance shall immediately notify the party to whom performance is due by telephone (to be confirmed in writing within [\*\*\*] hours of the inception of such delay) and describe at a reasonable level of detail the circumstances causing such delay.
- (c) If any Force Majeure Event substantially prevents, hinders or delays performance of any Services, Capital One may at its option immediately procure such Services from an alternate source, and Capital One shall be liable for payment for the cost of such Services from the alternate source for so long as the delay in performance shall continue.
  - (d) If any Force Majeure Event substantially prevents, hinders or delays performance of any Services for [\*\*\*] hours, then at Capital One's option:
  - (i) Capital One may terminate any portion of this Agreement so affected without liability as of a date specified by Capital One in a notice of termination to TSYS and the Charges payable under this Agreement shall be equitably adjusted to reflect those terminated Services. Capital One shall provide such notice within [\*\*\*] days after the right to so terminate arises; or
  - (ii) Capital One may terminate this Agreement without liability on either party's part as of a date specified by Capital One in a notice of termination to TSYS. Capital One shall provide such notice within [\*\*\*] days after the right to so terminate arises.
- (e) TSYS shall not have the right to any additional payments from Capital One for costs or expenses incurred by TSYS as a result of any Force Majeure Event.

#### 17.4 Limitation of Actions

No action under this Agreement may be brought by either party more than [\*\*\*] years after the right, claim, or cause of action accrues. If the prior sentence is not enforceable under Applicable Law with respect to a particular action under this Agreement, the two year period shall be extended for such action to the shortest period for which the prior sentence is enforceable under Applicable Law with respect to that action. Actions relating to third party claims shall be deemed to accrue at the time the third party claim is made.

# 18 DISPUTE RESOLUTION

Any dispute between the parties arising out of or relating to this Agreement, including with respect to the interpretation of any provision of this Agreement and with respect to the performance by TSYS or Capital One, shall be resolved as provided in this Section 18.

# 18.1 Informal Dispute Resolution

- (a) Subject to Sections 18.1(b), 18.1(c), and 18.4, the parties initially shall attempt to resolve their dispute informally, in accordance with the following:
  - (i) Upon the notice by a party to the other party of a dispute ("<u>Dispute Date</u>") which the Capital One Contract Manager and the TSYS Services Manager have been unable to resolve, each party shall appoint a designated representative who does not devote substantially all of his or her time to performance under this Agreement, whose task it will be to meet for the purpose of endeavoring to resolve such dispute.

- (ii) The designated representatives shall meet as often as the parties reasonably deem necessary in order to gather and furnish to the other all information with respect to the matter in issue which the parties believe to be appropriate and germane in connection with its resolution. The representatives shall discuss the problem and attempt to resolve the dispute without the necessity of any formal proceeding.
- (iii) During the course of discussion, all reasonable requests made by a party to the other for non-legally privileged information reasonably related to the subject matter of the dispute shall be honored in order that a party may be fully advised of the other's position.
  - (iv) The specific format for the discussions shall be left to the discretion of the designated representatives.
- (b) After [\*\*\*] days following the Dispute Date and prior to commencement of any arbitration as permitted under Section 18.1(c), either party may initiate mediation of the dispute by submitting to the AAA (the "<u>DM</u>") and to the other party a written request for mediation, setting forth the subject of the dispute and the relief requested. The parties shall cooperate with DM and each other in the mediation process. The mediation shall be conducted in accordance with the applicable practices and procedures of DM. Upon commencement of arbitration as permitted under Section 18.1(c), either party, upon notice to DM and to the other party, may terminate the mediation process. Each party shall bear its own expenses in the mediation process and shall share equally the charges of DM.
  - (c) Arbitration of a dispute may be commenced by either party upon the earlier to occur of any of the following:
  - (i) the designated representatives mutually conclude that amicable resolution through continued negotiation of the matter does not appear likely; and
  - (ii) [\*\*\*] days have elapsed from the Dispute Date if neither party has requested mediation under Section 18.1(b), or [\*\*\*] days have elapsed from the Dispute Date if either party has requested mediation within the required time frame (these periods shall be deemed to run notwithstanding any claim that the process described in this Section 18.1 was not followed or completed).

#### 18.2 Arbitration

(a) Except as set forth in Section 18.4, any dispute, claim or controversy arising out of or relating to this Agreement or the breach, termination, enforcement, interpretation, or validity of this Agreement, including the determination of the scope or applicability of this Agreement to arbitrate, shall be determined by arbitration in New York, New York before three (3) arbitrator(s). The arbitration shall be administered by AAA pursuant to its Commercial Arbitration Rules and Procedures then in effect. Judgment on the award may be entered in any court having jurisdiction. This clause shall not preclude parties from seeking provisional remedies in aid of arbitration from a court of appropriate jurisdiction.

- (b) Each party shall bear its own arbitration costs and expenses and all other costs and expenses of the arbitration shall be divided equally between the parties; provided however, that the arbitrator may, in the award, allocate all or part of the costs of the arbitration, including the fees of the arbitrator and the reasonable attorneys' fees of the prevailing party.
- (c) This Section 18.2 shall not limit either party's right to obtain injunctive relief pursuant to Section 18.4. The institution and maintenance of an action for injunctive relief pursuant to Section 18.4 shall not constitute a waiver of the right of either party to submit the dispute, claim, or controversy to arbitration.

#### 18.3 Continued Performance

EXCEPT AS OTHERWISE DIRECTED BY THE OTHER PARTY WITH RESPECT TO THE SUBJECT MATTER OF THE DISPUTE, EACH PARTY SHALL CONTINUE PERFORMING ITS OBLIGATIONS UNDER THIS AGREEMENT WHILE A DISPUTE IS BEING RESOLVED EXCEPT (AND THEN ONLY) TO THE EXTENT THE ISSUE IN DISPUTE PRECLUDES PERFORMANCE (DISPUTE OVER PAYMENT SHALL NOT BE DEEMED TO PRECLUDE PERFORMANCE) AND WITHOUT LIMITING EITHER PARTY'S RIGHT TO TERMINATE THIS AGREEMENT AS PROVIDED IN SECTION 19.

# 18.4 Injunctive Relief; Specific Performance

- (a) Each of the parties acknowledges that the non-breaching party would be irreparably harmed and the total amount of monetary damages for any injury to such party will be extremely difficult to calculate with specificity and therefore an inadequate remedy, and accordingly, the non-breaching party shall be entitled to seek temporary and permanent injunctive relief without posting bond or proving damages, against the breaching party and its employees, officers, directors, agents, representatives or independent contractors and to enforce specifically this Agreement and the terms and provisions thereof, if:
  - (i) any of the provisions of Section 8, 12, or 13 were not performed in accordance with their specific terms or were otherwise breached;
  - (ii) a party did not [\*\*\*]; or
  - (iii) [\*\*\*].
- (b) In such a circumstance, the non-breaching party may without delay institute an action in any court of the United States or any state or jurisdiction having subject matter jurisdiction, in addition to any remedy to which such party may be entitled, at law or in equity. If such court should find that one of the events set forth in paragraphs (i) through (iii) has occurred, the alleged breaching party agrees that, without any additional findings of irreparable injury or other conditions to injunctive relief (including the posting of bond), it shall not oppose the entry of an appropriate order compelling performance by the breaching party and restraining it from any further breaches.

# 18.5 Governing Law

(a) This Agreement and performance under it shall be governed by and construed in accordance with the law of the State of New York without regard to any portion of its choice of law principles that might provide for application of a different jurisdiction's law. The parties expressly intend to avail themselves of the benefit of Section 5-1401 of the New York General Obligations Law.

(b) To the maximum extent permitted under Applicable Law, no portion of this Agreement will be subject to the Uniform Computer Information Transactions Act ("UCITA") or Article 2 of the Uniform Commercial Code (each prepared by the National Conference of Commissioners on Uniform State Laws) as currently enacted or as may be codified or amended from time to time by any jurisdiction. To the extent that any aspect of this Agreement or any license granted under this Agreement is unclear or disputed by the parties and UCITA, if applied, would clarify the license or resolve the dispute independently of UCITA by applying the intent of the parties at the time they entered into this Agreement.

# 18.6 Expedited Dispute Resolution

Where expressly permitted in this Agreement, a party may submit a dispute for expedited dispute resolution in accordance with the following process ("Expedited Dispute Resolution"):

- (a) The party initiating the Expedited Dispute Resolution will: (i) provide the other party with notice of the dispute and its intent to initiate the Expedited Dispute Resolution process ("Expedited Dispute Notice"); and (ii) initiate binding arbitration with AAA.
  - (b) The arbitration shall be administered using the AAA's Streamlined Arbitration Rules and Procedures, modified as follows:
  - (i) Each party will have five (5) Business Days following the issuance of the Expedited Dispute Notice to provide AAA with a written brief explaining its position on the dispute.
  - (ii) The AAA arbitrator shall have ten (10) days to arbitrate the dispute and make an initial determination. Thereafter, each party will have three (3) days to file reply briefs. The AAA arbitrator will then issue a final, binding decision within two (2) days of receiving the parties' reply briefs.
    - (iii) The parties will each pay fifty percent (50%) of the fees payable to AAA for the Expedited Dispute Resolution.

# 18.7 Binary Arbitration

Where expressly permitted in this Agreement, a party may submit a dispute for resolution by binary arbitration. In such case, the parties shall follow the Expedited Dispute Resolution set forth above, except that the AAA arbitrator shall be required to decide which party's proposed resolution of the issue is more appropriate in the context of this Agreement. The arbitrator shall not have the option of deciding upon a resolution that reflects a compromise between (or is outside of) the parties' respective proposed resolutions. For example, if the annual charge for a New Additional Service is to be the subject of binary arbitration, and Capital One has proposed annual charges of one million dollars (\$1,000,000) while TSYS has proposed annual charges of one million five hundred thousand dollars (\$1,500,000), the AAA arbitrator shall determine which of the two proposals is more appropriate and may not select an amount somewhere between (or outside of) the two proposals. The AAA arbitrator's decision shall be binding on the parties.

# 19 TERMINATION

# 19.1 Termination for Cause

(a) If TSYS:

- (i) [\*\*\*] of this Agreement that is curable within [\*\*\*] days and does not cure the breach within [\*\*\*] days after receiving written notice of the breach from Capital One;
- (ii) [\*\*\*] of this Agreement that is not curable within [\*\*\*] days but is curable within [\*\*\*] days and following written notice of the breach from Capital One fails to:
  - (A) proceed promptly and diligently to correct the breach;
  - (B) develop within [\*\*\*] days after such notice a complete plan for curing the breach; and
  - (C) cure the breach within [\*\*\*] days after such notice;
  - (iii) [\*\*\*] of this Agreement that is not curable within [\*\*\*] days;
- (iv) within the [\*\*\*] period prior to giving notice as provided below, commits [\*\*\*] breaches of its duties or obligations (whether or not [\*\*\*]) which [\*\*\*] of this Agreement;
  - (v) ceases to be qualified by any Association to process Capital One transactions;
  - (vi) incurs a [\*\*\*]; or
- (vii) pursuant to Section 2.2(c) of Schedule C Capital One has the right to terminate this Agreement, has notified TSYS of its election to so terminate and any dispute has been resolved pursuant to Section 18,

then Capital One may, by giving notice to TSYS (within [\*\*\*] days of the expiration of the cure period, if any, provided above, or, in the case of items (v) through (vii), within [\*\*\*] days after the right to terminate arises), terminate this Agreement (in whole but not in part) as of a date specified in the notice of termination. Any termination by Capital One for breach shall not constitute an election of remedies and shall be without prejudice as to Capital One's other rights resulting therefrom.

(b) If:

- (i) Capital One fails to pay TSYS when due undisputed Charges that in the aggregate equal at least [\*\*\*] months of total Charges under this Agreement, and fails to make such payment within [\*\*\*] days after receiving notice from TSYS of the failure to make such payment;
- (ii) pursuant to the dispute resolution process described in Section 18, it is finally determined that Capital One has committed a [\*\*\*] and [\*\*\*] breach of its obligations under Section 13 in a manner that poses a serious threat to material TSYS Intellectual Property Rights, and Capital One fails to cure such breach within [\*\*\*] days after such determination is made; or

(iii) pursuant to Sections 2.2(d) and 2.2(e) of Schedule C TSYS has the right to terminate this Agreement, has notified Capital One of its election to so terminate and any dispute has been resolved pursuant to Section 18,

then TSYS may, by giving notice to Capital One (within [\*\*\*] days of the expiration of the cure period, if any, provided above), [\*\*\*] as of a date specified in the notice of [\*\*\*] that is at least [\*\*\*] days thereafter. For the avoidance of doubt, [\*\*\*] that are in violation of [\*\*\*] and [\*\*\*] with Section [\*\*\*] shall not be deemed a breach giving rise to a TSYS right of termination. WITHOUT LIMITING SECTION 10.2(C), THIS SECTION PROVIDES THE ONLY CIRCUMSTANCES UNDER WHICH TSYS MAY TERMINATE THIS AGREEMENT. A MATERIAL FACTOR IN CAPITAL ONE'S DECISION TO CHOOSE TSYS TO PERFORM THE SERVICES IS TSYS' AGREEMENT THAT IT WILL NOT TERMINATE UNDER ANY OTHER CIRCUMSTANCE. Any termination by TSYS for breach shall not constitute an election of remedies and shall be without prejudice to TSYS' other rights resulting therefrom.

#### 19.2 Termination for Convenience

- (a) Until the Commencement Date, Capital One may terminate this Agreement for convenience and without cause at any time by giving TSYS notice, and the termination shall be effective on the receipt of such notice. In such event Capital One shall pay to TSYS on the effective date of termination the applicable Termination Charge specified in Section 6.1 of Schedule C.
- (b) After the Commencement Date but during the Initial Term, Capital One may terminate this Agreement for convenience and without cause by giving [\*\*\*], which may not be [\*\*\*] after the notice is given. In such event, Capital One shall pay to TSYS on the effective date of termination the applicable Termination Charge specified in Section 6.1 of Schedule C, [\*\*\*].
- (c) During any Renewal Term, Capital One may terminate this Agreement for convenience and without cause by giving TSYS notice designating the termination date, which may not be earlier than [\*\*\*] months after the notice is given. In such event Capital One shall not be liable to pay TSYS any termination charge.
  - (d) If a purported [\*\*\*] by Capital One under Section 19.1 is ultimately determined not to have been [\*\*\*].
- (e) Capital One's payment of the applicable Termination Charges (if any) provided in Section 6.1 of Schedule C shall result in full and complete satisfaction of all of TSYS' and its Affiliates' claims against Capital One arising on account of a termination for convenience and is in consideration of TSYS' and its Affiliates' waiver of all existing or thereafter arising, known or unknown, claims against Capital One arising on account of such termination, but shall not limit or constitute satisfaction or waiver of any other claims TSYS and its Affiliates may have.

# 19.3 Termination Due To Election of License Option

This Agreement may also be terminated as provided in Section 8.13.

# 19.4 Termination Due To Force Majeure Event

This Agreement may also be terminated in whole or part as provided in Section 17.3(c).

# 19.5 Termination Due To [\*\*\*] Constraints

If a [\*\*\*] requires in writing that Capital One stop receiving Services from TSYS, and such requirement is based on an act or omission of TSYS, then Capital One may by giving notice to TSYS terminate this Agreement as of a date specified in the notice of termination. To the extent [\*\*\*] the [\*\*\*], Capital One will provide written notice to TSYS at the time the Capital One Contract Manager or other Capital One senior executive personnel become aware of any significant concerns or reservations of the [\*\*\*] as to Capital One's ability to [\*\*\*] with [\*\*\*], the manner in which the Services are being provided, or the risk that an order from a [\*\*\*] may be issued. Capital One will reasonably cooperate with TSYS (at TSYS' cost) with respect to any reasonable actions TSYS elects to take to address the issue raised by the [\*\*\*] following receipt of such notice. If Capital One terminates this Agreement under this Section 19.5, it will [\*\*\*] TSYS for [\*\*\*] pursuant to 6.1(a)(i) of Schedule C but will not be responsible for payment of the [\*\*\*].

# 19.6 Termination Due To TSYS' Financial Inability To Perform

If either Moody's Investors Service or Standard & Poors rates TSYS' credit at below investment grade or TSYS or any entity that at the time Controls TSYS:

- (a) files a petition in bankruptcy;
- (b) has an involuntary petition in bankruptcy filed against it which is not challenged within [\*\*\*] days and dismissed within [\*\*\*] days;
- (c) becomes insolvent;
- (d) makes a general assignment for the benefit of creditors;
- (e) fails to pay its debts as they mature or admits in writing its inability to pay its debts as they mature;
- (f) has a receiver appointed for its assets;
- (g) has any significant portion of its assets attached;
- (h) receives a "going concern" notification or similar qualification in any opinion from its auditor in connection with any audited financial statements filed with the Securities and Exchange Commission or delivered to its investors or to potential investors; or
- (i) is de-listed from any national stock exchange pursuant to an order of the SEC at the request of the exchange made because the exchange deems it necessary for the protection of investors,

then Capital One may by giving notice to TSYS terminate this Agreement for default as of the date specified in such notice of termination.

# 19.7 Extension of Termination/Expiration Effective Date

Regardless of the basis for termination (including a termination for cause by TSYS), Capital One may extend the effective date of termination/expiration [\*\*\*] or more times as it elects in its discretion, provided that the total of all such extensions shall not exceed [\*\*\*] months after the effective date of termination/expiration in place immediately prior to the initial extension under this Section and, if termination is pursuant to Section 19.1(b), if required by TSYS, Capital One (at Capital One's option) either (a) pre-pays the estimated monthly Charges

at least [\*\*\*] days prior to each month or (b) posts an irrevocable letter of credit (or performance bond) in sufficient amount to meet the next [\*\*\*] months of ongoing Charges to be incurred and pays all new Charges for Services when due. For any notice or notices of such extensions provided to TSYS less than [\*\*\*] days before the then-scheduled date of termination/expiration, Capital One shall also reimburse TSYS for additional Out-of-Pocket Expenses caused by such notices.

# 19.8 Termination/Expiration Assistance

Beginning twelve (12) months prior to expiration of this Agreement, on such earlier date as Capital One may request, or commencing upon (1) a notice of termination (including notice of termination based upon default by Capital One) or of non-renewal of this Agreement, and continuing through the effective date of expiration or, if applicable, of termination of this Agreement (as such effective date may be extended pursuant to Section 19.7) or (2) notice from Capital One that one or more Portfolios will be deconverted from the TSYS System, and continuing through the effective date of deconversion of such Portfolios, TSYS shall provide to Capital One, or at Capital One's request to Capital One's designee (including a competitor of TSYS) ("Capital One Designee"), the reasonable termination/expiration assistance requested by Capital One to allow the Services to continue in accordance with this Agreement and to facilitate the orderly transfer of the Capital One Accounts and Services, as applicable, to Capital One or the Capital One Designee ("Termination/Expiration Assistance"). Capital One's request may be in the context of Capital One's [\*\*\*], reduction, or removal of a portion of the Services or Capital One Accounts, as applicable, in accordance with Section 3.2, although this Agreement is not itself being terminated. Termination/Expiration Assistance shall include the Migration and Deconversion assistance described in Schedule A and the following:

- (a) <u>General</u>. As requested by Capital One, TSYS shall provide the Services and Termination/Expiration Assistance until all applicable Capital One Accounts are deconverted or the Licensed Materials are migrated to Capital One. Actions by TSYS under this Section 19.8(a) shall be subject to the other provisions of this Agreement, including all applicable Service Levels. Charges for such activities by TSYS are provided in Section 6.2 of Schedule C.
- (b) <u>TSYS</u>[\*\*\*]. If Capital One exercises the License Option, Capital One or a Capital One Designee shall be permitted to undertake, pursuant to a mutually agreed process that reasonably limits the number of [\*\*\*], to [\*\*\*] any TSYS [\*\*\*] that [\*\*\*] on the [\*\*\*] months during the [\*\*\*] month period prior to the date Capital One exercises the License Option. In such event the following shall apply:
  - (i) TSYS shall waive its rights, if any, under agreements with such TSYS [\*\*\*] the [\*\*\*] of such TSYS [\*\*\*] to [\*\*\*] or [\*\*\*] Capital One or a Capital One Designee;
    - (ii) Capital One or a Capital One Designee shall have [\*\*\*] to such [\*\*\*] for [\*\*\*] and [\*\*\*]; and
    - (iii) taken together, Capital One and all Capital One Designees may not [\*\*\*] more than [\*\*\*] ([\*\*\*]) TSYS [\*\*\*].
- (c) <u>Information</u>. TSYS shall provide Capital One and the Capital One Designee with information concerning the Services reasonably necessary for the timely transition

of the Services from TSYS to Capital One or the Capital One Designee, including the data TSYS possesses regarding Portfolios in such form as Capital One may reasonably request, together with instructions concerning the format and means of accessing such data.

- (d) <u>Bid Assistance</u>. As part of evaluating whether to undertake or allow termination/expiration or renewal of this Agreement, Capital One may consider obtaining, or determine to obtain, offers for performance of services similar to the Services following termination/expiration of this Agreement. As and when reasonably requested by Capital One for use in such a process, TSYS shall reasonably cooperate with Capital One and shall not frustrate the process, including by [\*\*\*] Capital One develops in connection with such process. In connection therewith Capital One shall be permitted, without obligation of confidentiality to TSYS to provide information about Capital One's Accounts, the information covered by Section 19.8(c), and [\*\*\*] to Capital One, although it may not provide the [\*\*\*] reports provided by TSYS.
  - (e) Capital One may seek Expedited Dispute Resolution pursuant to Section 18.6 in the event of a dispute related to this Section 19.8.

# 20 GENERAL

# 20.1 Binding Nature and Assignment

This Agreement shall be binding on the parties hereto and their respective successors and assigns. Neither party may, or shall have the power to, assign this Agreement without the prior written consent of the other, except that Capital One may assign its rights and obligations under this Agreement without the approval of TSYS to:

- (a) an entity which acquires all or substantially all of the assets of Capital One;
- (b) an entity which acquires all or substantially all of the Capital One Accounts receiving Services under this Agreement; or
- (c) to the successor in a merger or acquisition of Capital One;

provided that in no event shall such assignment relieve Capital One of its obligations under this Agreement.

Any other assignment by operation of law, order of any court, or pursuant to any plan of merger, consolidation or liquidation, shall be deemed an assignment for which prior consent is required and any assignment made without any such consent shall be void and of no effect as between the parties.

# 20.2 No Collateral Documents

It is the intent of the parties that this Agreement and the documents referenced herein shall provide all the terms of the relationship between them and that (except as specifically provided to the contrary herein) no representative of Capital One shall be required to enter into any additional contractual obligation (either personally or on behalf of Capital One). Without limiting the generality of the foregoing if either party's or its Affiliates' (or its representatives') personnel are requested or required by the other party or its Affiliates (or, in the case of TSYS, any subcontractor) to execute any releases or waivers to obtain access to such party's or its Affiliates' (or, in the case of TSYS, its subcontractors') premises they shall be void and

unenforceable and shall not be pleaded or introduced in any action. Such individuals are [\*\*\*] to be [\*\*\*] of this [\*\*\*] and [\*\*\*].

# 20.3 Mutually Negotiated

Each party acknowledges that the limitations and exclusions contained in this Agreement have been the subject of active and complete negotiation between the parties and represent the parties' agreement based upon the level of risk to Capital One and TSYS associated with their respective obligations under this Agreement and the payments to be made to TSYS and services to be obtained by Capital One pursuant to this Agreement. The terms and conditions of this Agreement (including any perceived ambiguity herein) shall not be construed in favor of, or against, either party by reason of the extent to which any party or its professional advisors participated in the preparation of the original or any further drafts of this Agreement as each party is a sophisticated business entity and has fully negotiated this Agreement with the full participation of counsel and other advisors and it represents their mutual efforts and their mutual intent.

# 20.4 Notices

All notices, requests, demands and determinations under this Agreement (other than routine operational communications), shall be in writing and shall be deemed duly given:

- (i) when delivered personally (against a signed receipt);
- (ii) on the designated day of delivery (other than a weekend or Federal holiday) after being timely given to an express overnight courier with a reliable system for tracking delivery;
  - (iii) on the next Business Day when sent by electronic mail with a copy sent by another means specified in this Section 20.4; or
- (iv) four (4) Business Days after the day of mailing, when mailed by United States mail, registered or certified mail, return receipt requested and postage prepaid and addressed as follows:

# In the case of Capital One:

Capital One Services, Inc. 15000 Capital One Drive Mail Stop 12073-0360 Richmond, VA 23238

Attention: Group Manager, SCM IT Services

with copies to:

Capital One Services, Inc. 1680 Capital One Drive McLean, VA 22102 Attention: Associate General Counsel,

Transactions

and

Capital One Services Inc. 15000 Capital One Drive Mail Stop 12076-0360 Richmond, VA 23238 Attention: Director, Helix Program

Management

and, in the case of notices of renewal, default, or termination:

Pillsbury Winthrop Shaw Pittman LLP 2300 N Street, NW Washington, DC 20037 Attention: James L. Alberg

In the case of TSYS:

with a copy to:

Total System Services, Inc. Building B-4 1600 First Avenue Columbus, Georgia 31901-1804

Attention: President

Total System Services, Inc Building C-3 1600 First Avenue

Columbus, Georgia 31901-1804 Attention: General Counsel

Either party may from time to time change its address or designee for notification purposes by giving the other prior notice of the new address or designee and the date upon which it will become effective.

# 20.5 Counterparts

This Agreement may be executed in several counterparts, each of which shall be considered an original but all of which taken together shall constitute but one and the same agreement.

#### 20.6 Relationship of Parties

In furnishing the Services, TSYS is acting as an independent contractor, and (subject to the express provisions of this Agreement) TSYS has the sole right and obligation to supervise, manage, contract, direct, procure, perform or cause to be performed, all work to be performed by TSYS under this Agreement. No contract of agency and no joint venture is intended to be created hereby. TSYS is not an agent of Capital One and has no authority to represent Capital One as to any matters, except as expressly authorized in this Agreement. Neither party's employees shall be deemed employees of the other party and the actual employer shall be responsible for reporting and payment of all wages, unemployment, social security and other payroll taxes, including contributions from them when required by law. Capital One does not and shall not have actual, potential, or any other control over TSYS or its employees.

# 20.7 Severability

If any provision of this Agreement conflicts with the law under which this Agreement is to be construed or if any such provision is held invalid by a competent authority, such provision shall be deemed to be restated to reflect as nearly as possible the original intentions of the parties in accordance with Applicable Law. The remainder of this Agreement shall remain in full force and effect.

# 20.8 Consents and Approval

Except where expressly provided that consent may be arbitrarily withheld by or an act is in the discretion of a party, where approval, acceptance, consent, or similar action by either party is required under this Agreement, such action shall not be unreasonably delayed or withheld. An approval or consent given by a party under this Agreement shall not relieve the other party from responsibility for complying with the requirements of this Agreement, nor shall it be construed as a waiver of any rights under this Agreement, except as and to the extent otherwise expressly provided in such approval or consent. Except as otherwise expressly provided herein, a party's failure to insist in any one or more instances upon strict performance of any provision of the Agreement, or failure or delay to take advantage of any of its rights or remedies under this Agreement, or failure to notify the other party of any breach, violation, or default, shall not be construed as a waiver or construction of any such performance, provision, rights, breach, violation, or default either then or in the future or the relinquishment of any of its rights and remedies.

# 20.9 Waiver of Default; Cumulative Remedies

- (a) Except as otherwise expressly provided herein, a delay or omission by either party to exercise any right or power under this Agreement shall not be construed to be a waiver thereof. A waiver by either of the parties of any of the covenants to be performed by the other or any breach thereof shall not be construed to be a waiver of any succeeding breach thereof or of any other covenant herein contained.
- (b) Except as otherwise expressly provided herein, all remedies provided for in this Agreement shall be cumulative and in addition to and not in lieu of any other remedies available to either party at law or in equity.

#### 20.10 Survival

Any provision of this Agreement that contemplates performance or observance subsequent to termination or expiration of this Agreement shall survive termination or expiration of this Agreement and continue in full force and effect.

# 20.11 Public Disclosures

- (a) All media releases, public announcements and public disclosures by either party relating to this Agreement or the subject matter of this Agreement, including promotional or marketing material, but not including announcements intended solely for internal distribution or disclosures to the extent required to comply with Applicable Law beyond the reasonable control of the disclosing party, shall be coordinated with and approved by the other party prior to release.
- (b) If a party determines that disclosure is required to comply with Applicable Law it shall promptly inform the other party and coordinate such disclosure with the other party. The disclosing party shall limit disclosure to that which is necessary and shall give due consideration to comments the other party and its counsel provide regarding the nature of the disclosure.
- (c) Notwithstanding the foregoing, in proposals and other limited distribution marketing materials TSYS may list Capital One as a customer and describe in general terms the services provided by TSYS under this Agreement.

#### 20.12 Service Marks

Without limiting disclosures permitted by Section 20.11, without prior written consent (which may be arbitrarily withheld and if granted revoked), neither party shall use the name, service marks, or trademarks of the other.

# 20.13 Third Party Beneficiaries

Except as provided in Sections [\*\*\*] and [\*\*\*], this Agreement is entered into solely between, and may be enforced only by, Capital One and TSYS, and this Agreement shall not be deemed to create any rights in third parties, including suppliers and customers of a party, or to create any obligations of a party to any such third parties.

# 20.14 Nonsolicitation of Employees

Subject to Section [\*\*\*], from the Effective Date until [\*\*\*] months after termination of this Agreement neither party shall directly or indirectly solicit or seek to procure, without the prior written consent of the other party: (i) in the case of Capital One, the employment of TSYS' employees engaged in the provision of the Services during the period they are so engaged and for [\*\*\*] months thereafter; and (ii) in the case of TSYS, Capital One's employees engaged in the provision of its information systems during the period they are so engaged and for [\*\*\*] months thereafter, except neither party shall be precluded from hiring any such employee who (A) initiates discussions regarding such employment without any direct or indirect solicitation by it; (B) responds to any public advertisement placed by it; or (C) has been terminated by the other party prior to commencement of employment discussions between it and such employee.

# 20.15 Covenant of Good Faith and Fair Dealing

Each party, in its respective dealings with the other party under or in connection with this Agreement, shall act in good faith and with fair dealing.

# 20.16 Entire Agreement; Amendment

This Agreement, including any Schedules and Exhibits referred to herein and attached hereto, each of which is incorporated herein for all purposes, constitutes the entire agreement between the parties with respect to the subject matter contained in this Agreement and supersedes all prior agreements, whether written or oral, with respect to such subject matter, including the Confidentiality Agreement dated May 27, 2004 and Letter of Intent dated October 19, 2004. Neither the course of dealings between the parties nor trade practices shall act to modify, vary, supplement, explain, or amend this Agreement. If either party issues any purchase order, terms or conditions, or other form, it shall be deemed solely for the administrative convenience of that party and not binding on the other party, even if acknowledged or acted upon. No change, waiver, or discharge hereof shall be valid unless in writing and signed by an authorized representative of the party against which such change, waiver or discharge is sought to be enforced. There are no promises, representations, warrantees, or other commitments relied upon by either party that are not expressly set forth herein.

IN WITNESS WHEREOF, the parties have each caused this Agreement to be signed and delivered by its duly authorized officer, each as of the Effective Date.

TOTAI	L SYSTEM SERVICES, INC.	CAPIT	AL ONE SERVICES, INC.
By:	/s/ Philip Tomlinson	By:	/s/ Robert M. Alexander
Name:	Philip Tomlinson	Name:	Robert M. Alexander
Title:	Chief Executive Officer	Title:	Senior Vice President, Marketing and Analysis

# CERTIFICATION FOR QUARTERLY REPORT ON FORM 10-Q OF CAPITAL ONE FINANCIAL CORPORATION AND CONSOLIDATED SUBSIDIARIES

# I, Richard D. Fairbank, certify that:

- 1. I have reviewed this quarterly report of Capital One Financial Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 1, 2005

CAPITAL ONE FINANCIAL CORPORATION

By: /s/ RICHARD D. FAIRBANK

Richard D. Fairbank Chairman of the Board, Chief Executive Officer and President

# CERTIFICATION FOR QUARTERLY REPORT ON FORM 10-Q OF CAPITAL ONE FINANCIAL CORPORATION AND CONSOLIDATED SUBSIDIARIES

### I, Gary L. Perlin, certify that:

- 1. I have reviewed this quarterly report of Capital One Financial Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 1, 2005

CAPITAL ONE FINANCIAL CORPORATION

By: /s/ GARY L. PERLIN

Gary L. Perlin Executive Vice President and Chief Financial Officer

# Certification

# Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code), I, Richard D. Fairbank, Chairman and Chief Executive Officer of Capital One Financial Corporation, a Delaware corporation ("Capital One"), do hereby certify that:

The Quarterly Report on Form 10-Q for the period ended September 30, 2005 (the "Form 10-Q") of Capital One fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of Capital One.

Dated: November 1, 2005

By: /s/ RICHARD D. FAIRBANK

Richard D. Fairbank Chairman of the Board, Chief Executive Officer and President

# Certification

# Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code), I, Gary L. Perlin, Executive Vice President and Chief Financial Officer of Capital One Financial Corporation, a Delaware corporation ("Capital One"), do hereby certify that:

The Quarterly Report on Form 10-Q for the period ended September 30, 2005 (the "Form 10-Q") of Capital One fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of Capital One.

Dated: November 1, 2005

By: /s/ GARY L. PERLIN

Gary L. Perlin Executive Vice President and Chief Financial Officer